

CDRI - Cambodia's Leading Independent Development Policy Research Institute



CAMBODIA LAND TITLING URBAN BASELINE SURVEY REPORT

A CDRI Publication

December 2007

Cambodia Land Titling Urban Baseline Survey Report

A CDRI Publication



CDRI Cambodia's Leading Independent Development Policy Research Institute

in collaboration with the

Ministry of Land Management, Urban Planning, and Construction (MLMUPC)

Phnom Penh, Cambodia

December 2007

© 2007 CDRI

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means—electronic, mechanical, photocopying, recording, or otherwise—without the written permission of the Cambodia Development Resource Institute

ISBN 978-99950-52-19-5

Cambodia Land Titling Urban Baseline Survey Report

December 2007

CDRI

∽ 56 Street 315, Tuol Kork, Phnom Penh
➢ PO Box 622, Phnom Penh, Cambodia
[∞] (855-23) 881-384 / 881-701 / 883-603 / 881-916 ≞ (855-23) 880-734
e-mail: cdrimail@online.com.kh website: http://www.cdri.org.kh

Layout and cover photos: Oum Chantha and Kim Chettra

Printed and Bound in Cambodia by JSRC Printing House, Phnom Penh

Acknowledgements	i
Acronyms	iii
List of Tables and Figures	v
Executive Summary	1
Chapter 1. Introduction	11
Chapter 2. Literature Review	13
2.1 Gender and Land Titling	15
2.2 Land Titling in Urban Cambodia	17
2.3 Methodology	17
Chapter 3 Desearch Hypotheses	10
2 1 Decidential Land and Property	20 20
2.2. Commercial Investments and Property	20 21
3.2. Commercial investments and Property	
3.5. Creat	
2.5 Land Administration	
3.5 Land Administration	25
5.6. Gender	20
Chapter 4. Methodology	
4.1. Survey Site Selection	27
4.2. Household Sampling	
4.3. Survey Instrument and Fieldwork	
4.4. Limitations	29
Chapter 5. Phnom Penh Municipality Baseline Survey	
5A.0. Background and Current Land Situation	
5A.1. Zoning and Planning	
5A.2. Infrastructure Rehabilitation and Development	
5A.3. Real Estate and Housing Markets	
5A.4. Informal Settlements	
5A.5. Peri-Urban Areas	
5B.0. Phnom Penh Baseline Survey	
5B.1. Household Characteristics	
5B.2. Plot and Unit Characteristics	
5B.3. Land Transactions	
5.4. Conflicts	
5B.5. Credit	
5B.6. Small and Medium Enterprises	
5B.7. Expected Benefits of Land Titling	
5B.8. Summary: Expected Land-Titling Impacts in Phnom Penh	
Chapter 6. Siem Reap Baseline Survey	59
6Å.0. Background Överview	
6A.1. Tourism Development	61
6A.2. Business and SME Development	61
6A.3. Land Markets	62
6A.4. Role of the Apsara Authority	62
6B.0. Siem Reap Baseline Survey	63
6B.1. Household Characteristics	63
6B.2. Plot/Unit Characteristics	64
6B.3 Land Transactions	70

Table of Contents

6B.4. Conflicts	
6B.5. Credit and Finance	
6B.6. Small and Medium Enterprises	74
6B.7. Expected Benefits of Land Titling	
6B.8. Summary of Siem Reap Land Titling Impacts	76
Chapter 7. Serei Saophoan Baseline Survey	
7.1. Household Characteristics	
7.2. Plot/Unit Characteristics	
7.3. Land Transactions	
7.4. Conflicts	
7.5. Credit and Finance	
7.6. Small and Medium Enterprises	
7.7. Expected Benefits of Land Titles	
7.8. Summary	91
References	
Annex A	
Annex B	
CDRI Working Papers	

Acknowledgements

The Cambodia Development Resource Institute thanks the Ministry of Land Management, Urban Planning, and Construction, Royal Government of Cambodia, for the unique opportunity it has provided to conduct the Urban Land Titling Baseline Survey Project (BSP). CDRI would also like to acknowledge the contribution of the World Bank for its technical support and consultation, as well as CIDA for its financial support, without which the urban baseline survey project would not have been possible.

We especially want to thank the many people in Phnom Penh, Siem Reap, and Serey Sophoan who took time out from their busy schedules to meet with the enumerators and patiently answer their many questions. Thanks are also due to the many Commune Council chiefs and members who took time to meet with CDRI researchers and help them better understand the dynamics of local land situations in their areas. Special thanks are also due to the many village chiefs who helped the survey groups in the field while also taking time to answer their many questions about land in their respective villages.

The Baseline Survey team members would also like to express their appreciation to the provincial LMAP and Cadastral officials who provided excellent overviews of the provincial LMAP work as well as data and other information as requested. The LMAP teams working in Phnom Penh also took time to participate in the project orientation for field enumerators and made many valuable suggestions concerning the design of the field survey instrument. The LMAP field teams in all three urban areas are to be congratulated for performing a complex task with professionalism under difficult circumstances. The BSP team would also like to express their deep appreciation to staff and officials from ANZ Royal Bank, Bonna Realty, Phnom Penh Municipality, and UN-Habitat for the time they took away from their busy schedules to meet with baseline survey researchers.

Many thanks are due to the Ministry of Land Management, Urban Planning, and Construction for their support and collaboration on this project. Dr. Ou Vuddy played a key role in helping to ensure that the work proceeded according to plan in her capacity as the Ministry's project liaison. Mr. Sar Sovann has also provided much appreciated support and guidance in his capacity as the LMAP Director. H.E. Im Chhun Lim has provided leadership and inspiration for the project in keeping with his vision of providing fair and impartial land administration services for all people of Cambodia. The BSP team also acknowledges the ongoing support and interest of the representatives from CIDA, GTZ and FinnMap.

Finally, thanks are due to the many staff at CDRI who provided essential logistical and administrative support throughout the entire project. The BSP staff would especially like to thank Ms. Pon Dorina, Mr. Sok Chamroeun, and Phim Runsinarith for their contributions to the design of the survey instrument and support in the field. Also, this study would not have been possible if it had not been for the ongoing interest and support from CDRI's Executive Director Mr. Larry Strange, CDRI's Director of Research, Dr. Hossein Jalilian, the Executive Manager Mr. Ray Hossinger and his successor Mr. Ung Sirn Lee, who has continued to provide support for the project and help bring it to a successful conclusion.

This report was prepared under the overall coordination and guidance of Dr. Brett Ballard (Senior Research Advisor) in collaboration with Mr. Phim Runsinarith (Research Associate). The report was written by Brett Ballard with a section by Phim Runsinarith concerning methodology. Phim Runsinarith prepared the data analysis that is reflected in the many tables throughout the text.

Acronyms

Acleda	Association of Cambodian Local Economic Development Agencies
ADB	Asian Development Bank
APSARA	Authorité pour la Protection du Site et L'Aménagement de la Région
	d'Angkor (Authority for the Protection and Management of Angkor and the
	Region of Siem Reap)
BSP	Baseline Survey Project
CDRI	Cambodia Development Resource Institute
CSES	Cambodia Socio-Économic Survey
DK	Democratic Kampuchea
FDI	Foreign Direct Investment
ha	hectare
HCA	High Capacity Area
HPI	High Potential Impact
JICA	Japan International Cooperation Agency
LAMDP	Land Administration, Management, and Distribution Programme
LCAs	Low Capacity Areas
LMAP	Land Management and Administration Project
LPI	Low Potential Impact
MFI	Micro-Finance Institution
MLMUPC	Ministry of Land Management, Urban Planning, and Construction
MSMEs	Micro, Small and Medium Enterprises
MT	Metric Ton
NGO	Non-Governmental Organization
NPRS	National Poverty Reduction Strategy, 2003-05
PAD	Project Appraisal Document
RGC	Royal Government of Cambodia
SES	Socio-Economic Survey
SMEs	Small and Medium Enterprises

Cambodian Words

Chamcar	ចំការ	Land that is used for growing crops other than rice.
Khum	ឃុំ	Commune
Krom Samaki	ក្រុមសាមគ្គី	'Solidarity groups' formed during the 1980s as a form of collective farming.
Moeun	ម៉ឺន	10,000
Riels	រេវ្រល	Unit of Cambodian Currency (4,000 riels/US\$)
Sangkhat	សង្កាត់	an administrative territorial subdivision equivalent to a commune.

List of Figures and Tables

Figures

Tables

Table 3.1: Sihanoukville Landholding Value Summary	.24
Table 4.1: Interview Quality Assessment (percent)	. 29
Table 5.1: Phnom Penh Land Value Estimates (USD/sq. m.)	.34
Table 5.2: Construction Approvals in Phnom Penh (USD m)	.34
Table 5.3: Consumption Quintiles (USD/year/person)	.37
Table 5.4: Annual Per Capita Consumption by District (USD)	. 38
Table 5.5: Number of Households Surveyed, by District and Consumption Quintile	.38
Table 5.6: Mode of Acquisition by District (Plot/Unit No. 1)	. 39
Table 5.7: Plot Purchases, 2000–05, by Quintile (USD per Plot/Unit No. 1)	. 39
Table 5.8: Mode of Acquisition, by Consumption Quintile (All Plots/Units)	.40
Table 5.9: Land Documentation, by Location (Plot/Unit 1)	.41
Table 5.10: Land Documentation, by Location (All plots/units)	.42
Table 5.11: Access to Plot/Unit 1, by District	.42
Table 5.12: Electricity Services (No. of Households Having Electricity)	.43
Table 5.13: Sewage Service (No. of Households Having Service)	.43
Table 5.14: Plot/Unit 1 Use (% of Households)	.44
Table 5.15: Plot Use, by District (No. of Households)	.44
Table 5.16: Land Values (USD/plot)	.45
Table 5.17: Reported Plot/Unit Values (USD), by District and Consumption Quintile	.45
Table 5.18: No. of Buildings per Plot/Unit 1, by Consumption Quintile	.46
Table 5.19: Housing Quality, Consumption Quintile by Roofing Type	.46
Table 5.20: Plot Purchases 2000–05, by Quintile (All Plots)	.47
Table 5.21: Land Purchases (2000–05), by District and Consumption Quintile (All Plots)	.47
Table 5.22: Land Purchases (2000–05), by District (All Plots)	.47
Table 5.23: Average Price of Land Purchased (2000–05)	.48
Table 5.24: Estimated Value of Land Purchased (2000–05)	.48
Table 5.25: Land Sales (2000–05), by District and Quintile	.49
Table 5.26: Land Sales (2000-05), by District and Consumption Quintile (USD/Plot)	.49
Table 5.27: Mode of Documentation of Land Sales, by Quintile (percent)	.50
Table 5.28: Credit Sources, by Consumption Quintile	.51
Table 5.29: Credit Activity (2003–05), by Quintile	. 52
Table 5.30: Credit Use, by Consumption Quintile (percent of loans)	.52
Table 5.31: SME Start-ups, by Year and Quintile	.53
Table 5.32: Small Business Financing, by Sources and Consumption Quintile	.54
Table 5.33: SME Employment, by Type and Consumption Quintile	.55
Table 5.34: Expected Benefits, by Consumption Quintile	.56
Table 6.1: Projected Population Density in the Survey Areas	.60
Table 6.2: Composition of Land Use (hectares)	. 60
Table 6.3: Growth in Tourism Services, 1998–06*	. 62
Table 6.4: Consumption Quintiles (USD/year/person)	.63
Table 6.5: Household Surveys, by Gender and Consumption	.64
Table 6.6: Mode of Acquisition, by Commune (Plot/Unit 1)	. 65
Table 6.7: Mode of Acquisition, by Commune (All Plots)	. 65
Table 6.8: Mode of Acquisition, by Consumption Quintile (All Plots)	. 65

Table 6.9: Land Documentation, by Location (Plot/Unit 1)	66
Table 6.10: Land Documentation, by Location (All plots)	67
Table 6.11: Access, by Commune (Plot/Unit 1)	67
Table 6.12: Electricity Services, by Type of Access (No. of households)	67
Table 6.13: Plot/Unit 1 Use, by Consumption Quintile (No. of Households)	68
Table 6.14: Land Values, by Access and Commune (USD/plot)	69
Table 6.15: Reported Plot/Unit 1 Values, by Commune and Consumption Quintile (USD)	69
Table 6.16: No. of Buildings per Plot/Unit 1, by Consumption Quintile	69
Table 6.17: Roofing Type, by Consumption Quintile	70
Table 6.18: Average Purchase Price, 2000–05 (All Plots) (USD/plot)	70
Table 6.19: Current (Dec. 2005) Estimated Value of All Plots (USD/Plot)	71
Table 6.20: Average Land Sale Values (USD), by Location (2000–05)	71
Table 6.21: Land Sales, by Mode of Documentation	72
Table 6.22: Credit Use, by Consumption Category	73
Table 6.23: SME Start-ups, by Year and Quintile	74
Table 6.24: Sources of Small Business Financing, by Consumption Quintile	74
Table 6.25: SME Employment, by Consumption Quintile	75
Table 6.26: Expected Benefits of Land Titling, by Quintile	76
Table 7.1: Households Surveyed, by Consumption Quintile and Sex of Household Head	79
Table 7.2: Average Consumption, by Quintiles and Sex of Head (USD/year/person)	80
Table 7.3: Mode of Acquisition, by Commune (Plot/Unit No. 1)	80
Table 7.4: Mode of Acquisition, by Gender of HH Head (Plot/Unit 1)	81
Table 7.5: Mode of Acquisition, by Commune (All Plots/Units)	81
Table 7.6: Type of Land Documentation, by Location (Plot/Unit 1)	82
Table 7.7: Type of Land Documentation, by Location (All Plots)	82
Table 7.8: Type of Access, by Commune (Plot/Unit 1)	82
Table 7.9: Households with Electricity Service, by Type of Access (Plot/Unit 1)	83
Table 7.10: Households with Sewage Service, by Type of Access, (Plot/Unit 1)	83
Table 7.11: Average Land Values, by Type of Access (USD/plot)	84
Table 7.12: Reported Plot/Unit 1 Values (USD), by Consumption Quintile	84
Table 7.13: No. of Buildings Per Plot/Unit 1, by Consumption Quintile	84
Table 7.14: Roofing Type, by Consumption Quintile	85
Table 7.15: Average Plot/Unit Purchase Prices (USD), 2000–05, by Quintile (All Plots)	85
Table 7.16: Average Price of Plot Purchases (2000–05)	86
Table 7.17: Average Price of Plot Purchases (2000–05), by Commune and Quintile	86
Table 7.18: Average Price of Land Sales, by Commune and Year (2000–05)	86
Table 7.19: Average Price of Land Sales, by Commune and Quintile (2000-05)	87
Table 7.20: Credit Use, by Consumption Category	88
Table 7.21: SME Start-ups, by Year and Quintile (2000–05)	89
Table 7.22: Small Business Financing, Sources by Consumption Quintile	89
Table 7.23: Type of SME Employment, by Consumption Quintile	90
Table 7.24: Expected Primary Benefits of Land-Titling, by Quintile	90

Executive Summary

1. Introduction

The Ministry of Land Management, Urban Construction and Planning (MLMUPC) is implementing a land management and administration project (LMAP) with support from international donors in order to strengthen land tenure security and land administration systems in Cambodia. Among other activities, LMAP is undertaking a systematic land-titling programme in which 1 million titles will be issued during the first phase (2003–07). During this phase, LMAP is issuing land titles in the urban and peri-urban areas of Phnom Penh municipality and Siem Reap district, as well as other urban areas in the country.

The Cambodia Development Resource Institute (CDRI), in collaboration with the MLMUPC, collected baseline survey data in and around Phnom Penh, Siem Reap and Serei Saophoan (i.e., Banteay Meanchey) from October to December 2005. CDRI conducted 2706 household interviews in areas representing a mix of property characteristics and land use patterns, as well as dynamics (e.g., transactions, documentation, conflicts). The primary objective of the urban baseline survey, as with an earlier rural survey (CDRI 2004), is to generate data that will provide a basis for a systematic comparative evaluation of the economic and social impact of the land-titling programme after three years.

The expected benefits of land titles in urban areas include increased investment in residential and commercial property, improved access to formal credit, more efficient markets that allocate land to more economically productive uses, fewer conflicts and better land administration services, including the use of the official registry to facilitate land transactions. Other expected benefits include increased government revenue from taxes on land transactions. Government planners and others also expect more secure land tenure through land titles to play an important role in reducing poverty in both rural and urban areas. Land titles are therefore expected to strengthen the institutional framework of urban development and thus contribute to sustainable macro-economic growth.

The economic impact¹ of land titles in the three urban areas must also be considered in the wider context of the rapid growth and longer term development plans of each, as outlined in the master plans for Phnom Penh and Siem Reap, because the areas vary in economic, social and geographic characteristics. However, the rapid growth of the real estate and housing markets, the expanding business sector, including small and medium enterprises (SMEs) and foreign direct investment (FDI), along with the expanding credit markets for consumer loans, home mortgages and commercial investments, suggest high consumer and business confidence in all three urban areas in the short-medium term. As a result, LMAP's systematic land-titling may be very timely in strengthening the institutional framework that supports economic growth in these urban areas.

The potential social impact of land titles must also be considered in the context of the wide disparities in wealth between upper and lower income groups, primarily with respect to equity, gender and poverty reduction. For example, upper income households could benefit more than others from the titles because the higher value of their property may make them eligible for larger loans from institutional lenders, while lower income households may not be

¹ The use of the term "impact" in this instance refers to the degree to which land titles may or may not affect measurable change in selected indicators. An "impact" can be either positive or negative in nature, although the sense of the word as used throughout this report is generally positive.

able to obtain credit because they may not meet other loan requirements. As for the distribution of benefits according to the sex of household head, female-headed households should be able to obtain more loans that can be used for housing improvements and business start-ups and expansion. Women's security of tenure may also be strengthened in cases of divorce or the death of a husband, although tenure rights represented by land titles must be upheld by the local authorities and courts in cases of conflict. In terms of poverty reduction, the urban poor may benefit from secure tenure in cases where they are eligible for land titles. There are, however, concerns that the poor could end up losing land by selling it for short-term gain in response to rapidly increasing land values. In cases where the poor reside in areas for which they may not be eligible for land titles, alternative approaches need to be developed to regularise or otherwise provide secure land tenure for them.

The assumptions and predictions of government planners and donor advisers about the expected benefits from land titling are primarily based on property rights theories that link investment incentives to secure land tenure, as well as scattered empirical evidence from research in countries where such programmes have been implemented. The basic argument is that people are more likely to invest resources in productive activities when they are confident that they, or their heirs, will enjoy the benefits of such investments in the future. (Brandao and Feder 1996; Barzel 1997). Another set of arguments supporting improved tenure security in the form of land titles links property ownership to the creation of potential wealth and affluence. De Soto (2000), for example, has argued that the poor in developing countries actually possess substantial assets, albeit in forms of "dead capital" that cannot easily be used for investment. This argument suggests that governments should provide property ownership in the form of legal titles that the poor can use as collateral for credit for investments, leading to significant reductions in poverty and improved well-being.

Formal land titles, however, are not the only governance mechanism that provides security of land tenure. Researchers have identified a number of other land tenure mechanisms in urban areas around the world, including customary tenure and a wide range of non-formal tenure categories with different interlocking forms of legitimacy (Payne 2000; Mulamir and Payne 2001). As a result, there may be no absolute standard by which security of land tenure can be assessed. The use of formal titles to govern land tenure must be understood as part of a market approach to urban development that takes place within a broad arena of formal institutions, including credit institutions, regulated real estate markets and official land management and registry systems.

In addition to land titles, land use classification, long-term planning (e.g., master plans) and enforceable zoning regulations are also essential for promoting better urban land management and efficient development. Land classification that clarifies private and public lands ² will help reduce conflicts and promote good governance. For example, private encroachments on public space, such as informal settlements, pose a variety of difficult governance issues, including opportunities for informal tax collection by public officials. Long-term planning by city officials and private investors is also facilitated by enforceable zoning regulations. For example, residential, commercial and industrial activities all require a variety of services involving different technologies, and certain private activities impose costs (i.e., externalities) on the public that can be minimised with better planning and zoning. Private investors must also be able to make accurate predictions about land use and values over time. The potential impact of land titles on investments may be reduced in the absence of publicly available master plans and enforceable zoning regulations.

The methodology used in the urban baseline survey employs a simple model to make predictions about likely impacts when location and household factors are combined. For

² State land in Cambodia is classified as either Private State Land or Public State Land. Private state land can be converted to private use, such as an economic concession. Public state land is land that serves a public purpose, and generally cannot be converted to private use without a specific decree.

example, high potential impact (HPI) households—based on annual per capita consumption quintiles in each urban area survey—will benefit most in high capacity areas (HCAs) situated along main roads and streets characterised by better access to utilities and services. Conversely, low potential impact (LPI) households will benefit least in low capacity areas (LCAs) situated off the main streets with less access to utilities and services. The degree to which HPI households in LCAs and LPI households in HCAs benefit from land titles will depend on where and at what rate urban infrastructure and access to utilities and services improve. This again suggests that overall land-titling impacts may be undermined by the absence of strategic development master plans and enforceable zoning regulations. It also suggests that land-titling effects can be enhanced by strong enforcement of property rights in the face of encroachment.

This model structures the research according to a series of testable hypotheses using quasiexperimental methods that compare household survey data from the current baseline survey (T0) with data that will be collected at a later point in time (T0 + x). The research focuses on several key indicators of land titling impacts, including housing improvements and business investments (e.g., start-ups, expansions, employment) access to and the use of formal credit, land use, land values and transactions, the use of the official registry to facilitate land transactions and conflict reduction and resolution. While potentially robust, this approach faces several constraints involving "tracing problems" (i.e., residents who leave a particular area after titles have been issued) and "time-lapse problems" (i.e., short-term impacts may vary from medium and longer terms impacts). "Measurement problems" may also arise when the gap between the time of the baseline survey and the issuance of titles becomes so large that intervening variables distort or cloud the impact of land titles on key indicators.

2. Findings and Observations

The direction and rate of growth in the three urban areas varies according to geographic, demographic, economic and institutional circumstances. Phnom Penh's economy is structured by manufacturing (garments) and services (schools, financial institutions, tourism), as well as transportation as it positions itself strategically with regard to regional and sub-regional highway and rail transport corridors. The high rate of natural population growth and migration are driving the city to expand in all directions except eastward across the Mekong River.³ Five planned satellite cities and new transport infrastructure will likely steer the direction and set the pace of land market development in impacted areas for the foreseeable future. Elsewhere, land use patterns and infrastructure development may unfold in a haphazard way according to private sector interests in the absence of a master plan with clear zoning regulations.

Siem Reap's economy is largely driven by rapidly expanding tourism and, to a much lesser degree, transport. Siem Reap's high rate of population growth from immigration will increase demand for residential housing, while increasing numbers of domestic and international tourists will place more demand on accommodation, services and infrastructure. Siem Reap is likely to expand primarily to the east and south-east, because growth possibilities west and north of the city are constrained by Apsara Authority polices governing land use in the Angkor Heritage Zones 1 and 2. For this reason, Siem Reap may currently have a stronger land use planning and enforceable zoning mechanisms than Phnom Penh.

Serei Saophoan is located in the north-west of the country at the convergence of National Route 5 connecting to Battambang and Phnom Penh and National Route 6 connecting to Siem Reap to the east and Poipet to the west on the Thailand-Cambodia border. This strategic location suggests that Serei Saophoan will likely develop into an important transportation and commercial hub in the medium- to long-term. The surrounding area is fertile farmland in

³ Phnom Penh is also expanding upward as more multi-floor buildings (residential and commercial) are being constructed in response to increasing land values and rental prices.

which agriculture is becoming increasingly mechanised, and as a result, Serei Saophoan could eventually emerge as an important agro-processing zone. Its commercial and residential areas are structured according to a more or less rectangular grid with wide, albeit largely unpaved, roads and streets that should facilitate the development of urban infrastructure and services in the future. It is likely that the city's commercial sector will initially expand in three directions along the major national routes, while residential areas will develop off the major highways.

Given such variations in the characteristics and dynamics of each urban area, it is likely that land-titling impacts will vary across the peri-urban and urban sectors of each as well as across different households. Generally speaking, the recent trends in the key indicators of such impacts suggest that land titles are likely to have multi-dimensional impacts, many of which may be indirect, and will therefore help strengthen the institutional framework of Cambodia's current growth. The degree to which such benefits are equitably distributed, however, will depend on how LMAP prioritises potentially conflicting or competing objectives that may sidetrack poverty reduction objectives. For example, the distribution of benefits will depend on where the various provinces and municipalities decide to target and sequence land-titling efforts, and the degree to which the focus is on optimising either the economic or social impact of land titles. The economically and socially optimal impacts of land titles therefore may not be the same across all households.

2.1. Housing and Home Improvements

More secure tenure in the form of land titles is expected to stimulate investment in housing and home improvements and to facilitate more efficient markets that allocate land assets to their most economically productive use. Lower consumption quintile households may make modest housing improvements according to available resources, capacity to borrow and location relative to transport and services. Such activities can be measured according to the quality of building materials used and the extent of improvements. Upper consumption quintile households, on the other hand, may have already made home improvements, although they may add or improve other buildings on their plots. Rather than make such improvements, these households might purchase additional land for productive or speculative purposes. They may also invest in housing complexes that would increase the overall supply of housing. The actual distribution of such housing would depend on the price of individual units and rental markets. This is a matter of particular importance for the poor because access to good affordable housing represents a key component of poverty reduction efforts in urban areas. For this reason, it will also be important to monitor housing rental prices, particularly in Phnom Penh and Siem Reap.

2.2. Land Use

Land titles may have an indirect impact on land use patterns to the extent they facilitate more efficient transactions that provide greater returns on investments. As well, the nature and direction of land use changes in and around the three urban areas will be highly dependent on the long-term development strategies of each, which will in turn require transparent land use planning and enforceable zoning mechanisms. In Phnom Penh, it is possible that in the three surveyed urban districts, a shift in land use will occur as residential property is converted to commercial and/or rental property, while in the peri-urban areas changes in land use may involve a shift from residential and agricultural uses to commercial (including industrial) uses. In the urbanising areas of Siem Reap, land use impacts will be directed eastward along National Route 6 and, to a lesser extent, along the road connecting Siem Reap to the Tonle Sap Lake, as agricultural land is converted to commercial and residential uses, while land use change in other peri-urban areas will be constrained by Apsara Authority zoning regulations in cultural heritage sites. In Serei Saophoan, land use is likely to change rapidly in all directions, especially to the west, east and south of the city as transport infrastructure is improved and regional trade increases.

2.3. Land Transactions

Land transactions will be the primary mode of land acquisition, and hence land redistribution, in all three urban areas. As a result, the share of land acquired through state allocation is likely to decline, perhaps sharply, unless social land concession mechanisms are established in one or all of the areas. Such shifts will not be the direct result of land titles *per se*, but rather the result of expanding land markets for productive and speculative investment. Land titling, however, will surely facilitate the process by reducing the transaction costs associated with transfers, thus making contractual exchange between buyer and seller more efficient. As a result, land titles will serve as a key institutional mechanism through which land assets are mobilised for medium- and large-scale development projects according to the each city's long-term master plan. This is particularly the case in and around Phnom Penh, where several large commercial and transport projects are already under way or being planned, as well as in Siem Reap, where a hotel zone (Cite d'Angkor) is planned on the north side of National Route 6, across from Psar Leu.

The degree to which land transactions will be recorded in the official registry is another matter. The survey data show that people in all three urban areas continue to rely primarily on documents that may be officially endorsed but are not legal in terms of validating ownership. There is some evidence that people continue to use such methods to validate land transactions at the village or commune level even after the issuance of the LMAP titles. This is a matter of fundamental importance because one of the key reasons for undertaking systematic land-titling has been to facilitate land transfers through the official registry. It is also important to observe that transactions that take place outside the official registry will contribute to continued conflicts over land and represent a major source of lost revenue for the government.

The limited number of cases of conflict reported in the urban survey will not be useful for measuring the extent to which land titles help reduce conflicts. In the follow-up survey, interviewers will need to ask households specifically about their perceptions concerning the role land titles may or may not have played in reducing conflicts. It will be especially important to inquire about the role that land titles have played in resolving conflicts that subsequently arise after they have been issued. Another important line of inquiry will concern the role of local authorities and the courts in enforcing land titles in cases of conflict.

2.4. Access to Commercial Credit

With respect to increased access to commercial credit, the research hypotheses predict that the volume of commercial loans will increase as households use land titles as collateral for bank credit. Two of the most visible areas of potential impact are likely to be the housing and real estate sector, as discussed above, and the business sector. Although it appears that people continue to use own resources for business start-ups, there is some anecdotal evidence of a recent increase in the number of loans for business operations and expansion. The indirect employment impact of land titles may be significant if they help stimulate increased investment in business start-ups and expansions.

There is also evidence of increased borrowing for real estate, including speculation, and housing investments. The issuance of land titles is expected to sustain, if not accelerate, these trends in the near to medium term. As a result, the number of loans is expected to increase along with the average amount of each loan. The use of the loans is likely to shift as well, as more people increase borrowing from commercial banks. Although these trends may be fairly consistent across all quintile groups, upper consumption quintile households may benefit more than lower quintile households from land titles because the higher value of their land will make them eligible for larger loans.

2.5. Small and Medium Enterprises

The survey data suggest that SMEs are a potentially important source of employment in all three urban areas. The employment impact of land titles may be significant if titles help stimulate increased SME investment and business expansions by promoting a sense of more secure tenure and providing access to more affordable credit.

According to recent discussions with banking officials, the number of loan applications and approvals has increased significantly over the past year, especially for business expansions, including SMEs. Investors are still reluctant to borrow from commercial banks for business start-ups because they are not willing to take loans when they are unsure of the returns on the investment. As a result, business start-ups continue to rely primarily on own sources and savings plus family and friends. Once the business is up and running and owners see a steady and predictable flow of returns, they appear more willing to obtain a loan from commercial banks in order to expand.

Such shifts in attitudes concerning commercial banks appear to be a relatively recent phenomenon. As the business climate in Cambodia improves and investors develop more trust in the banking sector, it is likely that the trend toward increased demand for credit for a variety of purposes will continue. Because commercial banks require "hard" land titles to secure such loans, it is quite likely that land titles will contribute to increased consumer and business borrowing. Such an increase, however, could not be solely attributable to land titles; rather, it would represent the merging of several key variables to form a virtuous cycle of development and growth. These variables would include increased tenure security in the form of land titles, business experience and skill (i.e., entrepreneurship), political stability and security, banks that want to lend and know how to do so and people's institutional trust in banks. The development of land and credit markets, as well as the business sector, all seem to rely on many of the same factors regarding trust in institutions. Once these factors are well in place, land and credit markets may function more efficiently, thus enabling entrepreneurs to predict returns on investments with greater accuracy and reliability.

2.6. Gender

The research hypotheses predict that land titles will have a wide range of impacts on genderrelated concerns pertaining to land. Many indicators identified in the rural baseline survey, such as landholdings and access to affordable credit, are also applicable to assessing the impact of land titling on gender in urban and peri-urban areas. Other indicators include investments in housing and business start-ups and expansion. It is expected that femaleheaded households will obtain an increased number of loans that are invested in housing and business improvements. The rate at which female-headed households obtain loans may, however, depend on a variety of circumstances, including household assets such as land.

The research hypotheses also predict that women's tenure security will be strengthened. One way to assess this will be to examine the outcome of cases involving the death of a husband or divorce and evaluate the extent to which the spouse's or widow's land rights have been upheld by the courts or other conflict resolution mechanisms. Another way to assess this is to examine cases involving female-headed households. While a household survey may enable researchers to count the number of such cases, qualitative approaches are also required to understand better the social and legal dynamics of such cases.

2.7. Observations on Methodology

The data collected in Phnom Penh municipality, Siem Reap and Serei Saophoan conform largely to expected trends and patterns and therefore appear reliable. Only in a few instances, such as the surprisingly low number of reported land-related conflicts and apparent inconsistencies concerning housing construction (e.g., roofing materials), did the data show questionable or puzzling results. More rigorous efforts concerning land and building valuation

techniques and procedures should also be employed in the follow-up survey. MLMUPC personnel familiar with land valuation techniques should be more involved in helping to train field enumerators and to oversee quality control regarding valuation in the field. Most importantly, the follow-up survey should incorporate qualitative research tools into the overall methodology in order to provide more texture to the household survey data, because many of the important nuances concerning the social and economic impacts of land titles cannot be fully captured using quantitative research methods that rely solely on household surveys.

3. Policy Implications and Research

During the course of the baseline survey project, several issues emerged as important areas requiring more focused follow-up. These include issues associated with targeting land tenure interventions in support of the government's poverty reduction objectives, land use planning and zoning and facilitating land transactions through the official registry. This discussion concludes with recommendations for further research in matters pertaining to the urban poor and gender equity in land tenure security.

3.1. Poverty Reduction

More secure tenure in the form of land titles is expected to stimulate more investments in housing and home improvements as well as to facilitate more efficient markets that allocate land assets to their most economically productive use. Because housing is one of the central issues associated with the urban poor, important questions arise concerning the degree to which land titles will stimulate improved housing for the poor. In cases where the poor reside on land that they rightfully occupy and for which they are eligible to receive title, the research hypothesis predicts that these households will over time begin either to invest own resources or to use titles to borrow to improve their housing. However, there are serious concerns that the poor could sell their land in response to rapidly increasing land values. If this is the case, then alternative modes of providing secure land tenure may need to be developed in accordance with the poverty reduction objectives of the land-titling programme. For example, informal communities could initially be granted communal land titles along with and in support of community upgrading efforts. Individual titles could be issued later, at which time households could sell land according to their ownership rights.

In cases where the poor do not have a rightful claim to the land they occupy, they are vulnerable to eviction as the state develops infrastructure and land values increase. Alternative modes of land occupancy—such as social land concessions—are, however, a more viable and sustainable solution than forced evictions. For example, in cases where the poor occupy state land, they could be granted temporary occupancy rights over a specific period while alternative arrangements are planned and developed. Although land titles may indeed play an important role for the poor who own land, for other poor more innovative interventions may be required in order to provide secure tenure leading to housing and home improvements as well as other benefits.

3.2. Land Use Planning and Zoning

The sequential relationship between land use planning, zoning and titling is important in terms of optimising the benefits from land titles in urban areas. The private and social benefits of land titling are optimised in urban areas where land use master plans and enforceable zoning regulations are in place prior to the titling exercises. The land-titling efforts in all three urban areas have begun before there has been clear boundary demarcation between state public, state private and private land. While the land-titling programme represents an important and integral step in this process, special efforts must be made in certain circumstances to avoid issuing titles in specific areas. For example, in some areas where land

has been transacted in an unplanned and haphazard way, there may be insufficient space available for infrastructure or services.

In Phnom Penh, and to a lesser extent Siem Reap and Serei Saophoan, however, the private sector has been largely determining land use patterns in the absence of master plans and zoning regulations. This represents potential inefficiencies associated with distortions in the distribution of public utilities and services that serve private interests rather than the public interest. As a result, it is possible that the issuance of land titles may exacerbate problems associated with the current mix of land uses in any given area. Without government zoning regulations and enforcement, the uncertainties associated with uncontrolled mixed land uses could (1) drive up transaction costs associated with competition over public resources among private individuals, (2) impede incentives for productive investments, particularly for small investors, and (3) undermine the development of efficient land markets.

A master plan that outlines Phnom Penh municipality's development objectives through 2020 as well as zoning and land use planning was drafted in 2003 but has yet to be approved. In the meantime, many public lands and buildings have been sold illegally, and there are few structures and little land remaining for developing infrastructure and utilities. Moreover, a great deal of land occupied for commercial purposes has not been productively used but is held for speculation. The environment and aesthetics of the city are being undermined by an excessive concentration of commercial activities in small areas, as well as illegal construction. It is important that the master plan for Phnom Penh be approved and made public as soon as possible.

Such haphazard growth and development contribute to poorly functioning land markets and undermine good governance because they encourage speculation by government staff or those with access to insider information. For example, if speculators acquire all the land in a given area, the result will be the need eventually to adjust land ownership and boundaries to accommodate necessary infrastructure. Such adjustments will be costly and time consuming, and will very likely generate conflict. Therefore, prior to titling, land use in underdeveloped areas should be demarcated in accordance with zoning regulations and urban development plans.

The major exception to this set of observations concerns the Apsara Authority in Siem Reap, which is mandated to preserve and maintain the Angkor Heritage Park. Land use within the Apsara jurisdiction, especially Zones 1 and 2, is strictly enforced. While Apsara is not always as transparent as it should be, its approach to strictly enforcing zoning regulations in the Siem Reap area should serve as a model of land use planning and zoning enforcement in other urban and peri-urban areas, especially in Phnom Penh. Given the rapid population growth and urban expansion, this is one of the most urgent priorities facing the government.

3.3. Land Transfers and the Official Registry

As in rural areas, in all three urban survey areas, there is a strong preference for using "official but not legal" means of facilitating land transactions. This suggests that if—as expected—land sales continue to increase in urban areas after the issuance of titles, people may still continue to document transactions locally in a manner that involves government officials but is not legal. There is some anecdotal evidence of this occurring already in areas where titles have been issued as well as documented evidence in areas such as Prey Nob in Sihanoukville (ADI, 2007). This is a matter of fundamental importance because one of the key reasons for undertaking systematic land titling has been to facilitate land transfers through the official registry. It is also important to observe that transactions that take place outside the official registry will contribute to continued conflicts over land and represent a major source of lost revenue for the government.

In terms of land transactions and land conflict resolution, the data show a clear preference for managing land matters at the village and commune levels. With particular respect to land

transactions, this suggests that a long-term objective for land administration and local governance would be to locate cadastral authority and related functions (e.g., land sales tax collection) at the commune level. There would also need to be objective incentives to encourage people to handle land transactions through the official registry. For example, failure to pay tax should invalidate the legality of a land sale. The land sales tax could also be lowered to encourage better compliance. In cases involving land conflicts, clear jurisdiction and lines of appeal are required to empower local authorities and the courts to resolve cases objectively according to the law.

3.4. Research Priorities

Two areas of particular concern require specifically targeted research. One area is land tenure security in informal urban settlements throughout the country. This research would consider issues pertaining to tenure security and property ownership in so-called illegal settlements as well as slum communities where the poor rightfully occupy the land on which they reside. The first component of such research could include a comprehensive social assessment, including conflicts and their resolution, of slum communities that could be eligible for regularising tenure and upgrading prior to the eventual issuance of individual land titles. This research could help inform efforts to provide people with alternative modes of secure tenure in support of poverty reduction objectives. A second component of the research could include reference to issues pertaining to evictions and resettlement, including the potential impact of land titling in relocated settlements.

A second area of research is land tenure security for women. This research would look specifically at how land titles have improved or strengthened women's land tenure security and the well-being of female-headed households. In both cases, qualitative research methods would need to be employed because many related issues may be sensitive and complex, and would therefore not be easily captured with a quantitative survey. Indeed, many of the issues pertaining to gender equity in land tenure rights and security are primarily social and cultural in nature and require innovative research methodologies.

Chapter 1. Introduction

The Ministry of Land Management, Urban Construction, and Planning (MLMUPC) is implementing a land management and administration project (LMAP) with support from international donors in order to strengthen land tenure security and land administration systems. Among other activities, LMAP is implementing a systematic land-titling programme that will issue 1 million titles during the first phase (2003–07). During this time, LMAP will issue land titles in the Phnom Penh municipality and Siem Reap town, including both periurban and urban areas, as well as other urban areas.

In 2004, the Cambodia Development Resource Institute (CDRI), in collaboration with the MLMUPC, completed a baseline survey for the rural land-titling project covering 1232 rural households in five provinces, as well as 99 households in Sihanoukville city centre (Ballard and So 2004). Given that a significant share of land-titling resources are being devoted to Phnom Penh and other urban areas in Cambodia, the MLMUPC then extended the Baseline Survey Project (BSP) to include urban areas in and around Phnom Penh, Siem Reap and Serei Saophoan (Banteay Meanchey). During the two months from October to December 2005, CDRI conducted 2706 household interviews in and around these cities in areas representing a mix of land use patterns and dynamics, as well as property characteristics.

The expected benefits of land titles in urban areas include increased investments in residential and commercial property, improved access to formal credit, more efficient land markets, fewer conflicts and better land administration services, including the use of the official registry to facilitate land transactions. Other expected benefits include increased government revenue from taxes on land sales and, eventually, property taxes on formally registered land. Government planners and others also expect secure land tenure in the form of land titles to play an important role in reducing poverty in both rural and urban areas. As a result, land titles may play a significant role in strengthening the institutional framework governing the urban development of these and other urbanising areas in Cambodia and thus contribute to sustainable macro-economic growth.

The expected economic impacts of land titles in the urban and peri-urban areas of Phnom Penh, Siem Reap and Serei Saophoan must be considered in the wider context of the rapid growth and longer term development plans outlined in the master plans for each area. Although each area varies according to economic, social and geographic characteristics, the rapid growth of the real estate and housing markets, the expanding business sector, including small and medium enterprises (SMEs) and foreign direct investment (FDI), along with the expanding credit markets for consumer loans, home mortgages and commercial investments, suggest that consumer and business confidence is high in all three urban areas in the short to medium term. As a result, the introduction of the LMAP systematic land-titling appears to be timely in strengthening the institutional framework that supports urban economic growth.

The potential social impacts of land titles must also be considered in the context of the wide disparities in wealth between upper and lower income groups, primarily with respect to equity, gender and poverty reduction. For example, upper income households tend to own property that is of higher value than the property owned by lower income households. This suggests that upper income groups may benefit more from land titles because the higher value of their property may make them eligible for larger loans from commercial lenders, while lower income households may not be able to obtain loans because they may not meet other

formal requirements. It is expected that female-headed households will be able to obtain more loans that can be used for housing improvements and business start-ups and expansion. Women's tenure security may also be strengthened, for example in cases involving divorce or the death of a husband, although tenure rights represented by LMAP land titles must be upheld by the local authorities and courts in cases of conflict. In terms of poverty reduction, the urban poor may benefit from secure tenure in cases where they are eligible for land titles. In other cases, more innovative approaches to providing secure tenure for the urban poor may need to be developed.

The primary objective of the urban baseline survey project, as with the earlier rural baseline survey project, is to generate data that will provide a basis for a systematic comparative evaluation of the economic and social impacts of the land-titling programme after three years. The purpose of this report is to provide an overview of the international literature and research hypotheses guiding the design of the urban survey methodology and framework, as well as to present some of the key findings from the baseline survey. The degree to which trends and patterns appear to conform to theoretical predictions or empirical evidence from other studies will also be discussed. The discussions concerning Phnom Penh and Siem Reap will be framed in the context of the master plans guiding urban development in the two cities through 2020. The discussion will also include an assessment of the reliability of the data collected as well as the research methodology, when relevant. Finally, key issues or problems pertaining to the design and implementation of the land-titling programme are identified and recommendations made where appropriate.

The report is structured as follows. Section 2 reviews the international literature concerning urban land issues, with a particular emphasis on urban land titling. Section 3 discusses the research hypotheses that predict the expected land-titling impacts and structure the survey instrument for residential households and small commercial businesses. Section 4 outlines the field research methodology, including sampling, survey instruments and fieldwork organisation. Sections five to seven then discuss the main findings in Phnom Penh, Siem Reap and Serei Saophoan, respectively. Section 8 concludes by summarising the key observations and makes several recommendations concerning policy and implementation, as well as for additional research.

Chapter 2. Literature Review

The assumptions and predictions that government planners and donor advisers make about the expected benefits from any land-titling programme are primarily based on property rights theory that links investment incentives to secure tenure, as well as some empirical evidence from research in countries where titling programmes have been implemented. The basic argument is that people are more likely to invest resources in productive enterprises when they are confident that they, or their heirs, will enjoy the benefits of such investments in the future. For example, Brandao and Feder's (1996) assertion that "secure individual (or corporate) property rights are critical in establishing a structure of economic incentives for investment in land-based activities" applies to both urban and rural areas.

Another set of arguments supporting improved tenure security in the form of land titles links property ownership to the creation of potential wealth and affluence. De Soto (2000) has argued that the poor in developing countries actually possess substantial assets, albeit in forms of "dead capital" that cannot easily be used for investment. This argument suggests that governments should provide property ownership in the form of titles that are enforceable by law so the poor can use them as collateral with which to secure credit for investments in housing and productive enterprises, thus leading to significant reductions in poverty and improved well-being. Such views have attracted widespread support from some governments and international development agencies, including the World Bank. For example, Deninger (2003) asserts that "provision of secure tenure to land improves the welfare of the poor, particularly by enhancing the asset base of those whose land rights are often neglected, and, creates incentives needed for investment, paramount to sustainable economic growth".

Payne *et al.* (2007: 5) observed, however, that "reviews of de Soto's analysis and proposals in the professional media have generally been critical". Some researchers have warned against applying single policy options globally, irrespective of prevailing local circumstances that vary and are complex. Other critiques include specific cautions about over estimating the potential impacts on access to formal credit as well as a general lack of convincing empirical evidence concerning the actual impact of land titling. Another important critique of strictly economic approaches to and rationales for land titling is that such arguments disregard or ignore the important fact that property rights and land tenure also entail political and social relationships between individuals and groups. Such relationships make issues associated with the exercise of political power important areas for consideration and analysis.

Another set of critiques concerns the fact that formal land titles are not the only governance mechanism that provides security of land tenure. Researchers have identified a number of other land tenure mechanisms in cities around the world, including customary tenure and a wide range of non-formal tenure categories with different interlocking forms of legitimacy (Payne 2000; Mulamir and Payne 2001). As a result, there may be no absolute standards by which security of land tenure can be assessed. The use of formal land titles to govern land tenure must be understood as part of a market approach to development that takes place within a wider arena of formal institutions, including credit institutions, regulated real estate markets and official land registry systems (Payne *et al.*, 2007).

This observation suggests, then, that with the right set of enabling institutions in place, secure tenure in the form of land titles may indeed contribute to the development of economically efficient land markets as commercialised land rights are, over time, purchased by people who

are willing and able to pay the highest price. Kim (2004: 278), for example, has observed that when bids are based on estimates of how much profit can be derived from owning land, "the highest bidder is the one who can most efficiently capitalize on the asset". Other types of property investments that people and/or firms may make in urban areas include new home construction and housing improvements, new start-ups or expansions of SMEs and larger scale commercial or industrial production.

Such investments can generate increased demand for financial resources, which in turn may stimulate the development of a more efficient financial services sector. In this way, land titles are expected to increase people's access to formal credit institutions. This could be especially important in terms of poverty reduction because land titles may enable poorer community members to access lower cost loans for productive purposes. Payne (2000), however, has argued that the real issue regarding credit for the poor is not so much collateral, but rather the banking sector's reluctance to provide small loans to poor borrowers. In Thailand, the potential impact of land titles on access to credit also did not appear to be as clear in urban as in rural areas. In terms of credit supply, Onchan and Aungsumalin (2002) found that the establishment of commercial banks was mostly influenced by government policies rather than the land-titling project. These observations suggest that increases in demand for credit may not necessarily result in corresponding supply increases solely due to the issuance of land titles, at least in the short term. Payne *et al.* (2007), for example, have since concluded that titling has not, at least in the short-term, generated any significant improvement in access to formal credit.

Land-titling programmes are also a key component of efforts to govern urban land markets more efficiently so that scarce land resources are eventually allocated to their most economically productive use. Land markets that are characterised by poor information, unpredictable registration and transfer procedures and various types of conflicts, as in Cambodia, are economically inefficient because of the high transaction costs associated with the negotiation and enforcement of property rights and contractual exchange. Secure and predictable property rights reduce procedural uncertainties and provide more accurate information about actual land values relative to use. If this is indeed the case, land titles should help investors make more accurate assessments of the costs and eventual returns on investment in sectors such as housing and real estate.

Land markets that are characterised by various types of conflicts and ineffective governance are also socially inefficient. This includes cases where people lose their land without fair and adequate compensation and are forced to migrate in search of income opportunities. Another set of potential social costs concerns the distribution of land ownership and access to housing. Over time, more efficient land markets may result in a greater concentration of land in the hands of fewer households. The social costs associated with such a distribution may be somewhat offset by fair and adequate compensation along with viable employment opportunities for those who are selling or otherwise losing their tenure rights.

At the same time, there is some evidence that land titles can have a significant impact on the market value of land. For example, residential plots in Jakarta with clear title sold for 45 percent more than comparable plots without clear title (Dowall and Leaf 1989). In Davao, land values increased by 58 percent after titling (Simon).⁴ The type of tenure documentation also has influenced the sale value of land in Ho Chi Minh City's emerging real estate market (Kim 2004). Such impacts on the value of land may have important implications for the poor as lower income households could be squeezed out of the land markets as land values increase. In terms of housing, increasing costs associated with land values could be passed on to tenants, again squeezing out lower income households and perhaps eventually forcing at least some into unauthorised settlements.

⁴ The reference to Dowall and Leaf (1989) and Simon is in Payne *et al.* 2007: 37–38. There is no year provided for Simon.

As in rural areas, the relationship between land titling and land values will have an important bearing on land use patterns in urban and peri-urban areas. In rural areas, for example, the general progression of increasing land values is from less productive (extensive) to more productive intensive farming and cropping systems, while in urban areas the progression may range from residential to commercial and industrial uses. In peri-urban areas, the progression may run from various agricultural uses to residential, commercial or industrial uses, or some mix. In all three areas, increasing land values associated with more economically productive land use will spark more competitive land markets. Just as land values in rural areas may vary according to location and productivity, land values in urban areas will vary according to location in land values in peri-urban and urban areas, however, may be even more exaggerated, in part due to higher population densities and a more diverse range of land uses, thus stimulating a demand for more efficient and effective land management and administration systems.

In addition to land titles, land use classification, long-term planning and enforceable zoning regulations are also essential for promoting better urban land management and efficient development. Land classification that clarifies private and public lands may also reduce conflicts and promote good governance. For example, private encroachments on public space, such as roadside markets, often pose a variety of safety issues, while providing opportunities for informal tax collection by public officials. Long-term planning by city officials and private investors is also facilitated by enforceable zoning regulations. For example, residential, commercial and industrial activities all produce different amounts and kinds of wastes that require different technologies and management in order to preserve and promote public sanitation and health.⁵ Private investors also need to be able to make accurate predictions about land values over time. In this sense, the potential impact of land titles on investments may be undermined by the absence of publicly available master plans and enforceable zoning regulations.

As for urban land administration, an improved and transparent land registry system may promote the development of more efficient land markets by reducing informal transaction costs associated with the contractual exchange of land and directing land transactions toward more economically productive uses. A more efficient and transparent official registry could also, in the long run, contribute to the general state revenues through efficient land tax and user fee systems. Burns (2006),⁶ for example, has provided evidence that the land-titling programme in Thailand has generated significant tax revenue for the government, revenue flows that continued even during and after the 1997 property market crash. In some cases, though, land titling in urban areas has failed to have this effect, largely because of people's reluctance to pay taxes when it was not clear what services they would receive in exchange. Another concern is the impact that such taxes might have on the newly titled poor whose incomes remain "low and irregular" (Payne *et al.*, 2007: 49).

2.1 Gender and Land Titling

Payne *et al.* (2007: 19–20) observes that gender is increasingly recognised as "an important variable in assessing the social impacts of tenure policy and therefore titling programs". However, there are "relatively few studies that analyse the impact of land titling processes on gender equity, although this is one of the main social objectives put forward to justify titling programmes". As with other impact indicators, the evidence concerning the role of land titling in promoting gender equity is decidedly mixed. While there are several encouraging reports from around the world, there are also many indications that there is still a long way to go before there is gender equity in terms of land tenure rights. In addition to legal and procedural

⁵ Other examples would include variations in water and electricity requirements.

⁶ The reference to Burns (2006) is in Payne *et al.* 2007: 49.

factors, cultural factors are "an even more entrenched barrier which will take years, and possibly even generations to change". Payne concludes that "land rights cannot be isolated from the legal framework and cultural environment".

The literature that links gender and land issues in Cambodia is uneven. Although many land studies refer to gender issues in general, there has not yet been any study that specifically addresses gender and land. For example, in the recent World Bank report on equity, the section concerning gender inequality observes that "other aspects of rapid economic and social change have resulted in the emergence of new forms of gender inequality" and that "inequalities exist in access to income generating opportunities and control of household assets ..." (WB, 2007:142). The section on land, including land titles, does not discuss gender, however, while in the section on gender, land-related matters are not discussed. In a sense, the linking of land and gender seems to have fallen through the cracks, and the references to land titling impacts are gender neutral.

Research elsewhere refers to gender and land in the context of poverty. Of particular importance are issues pertaining to land tenure security when there are conflicts within households or families concerning inheritance and divorce. A recent CDRI study (FitzGerald, 2007: 120) observed that "women's control over land and other financial assets is constrained by gender relations that grant men authority as heads of households ... Women in the study were also vulnerable to losing their land rights due to lack of formal titles". For example, it has long been a matter of concern that land certificates and other paper used to document land tenure were routinely listed in the name of a male head of household in the cases of married couples. As a result, women's land tenure security has been weak and subject to conflict, making them highly vulnerable to losing their land rights, as in the case of death or divorce where the husband's family repossesses land and other assets, or when they default on loans or experience shocks".

In cases of female heads of household, many have been without certificates or other documents because of the high costs of obtaining them, particularly for certificates. It should be added that such problems concern male-headed households as well, although they are especially important for female-headed households. The CDRI study referred to above also observed that female-headed households "are more likely to lose assets, including land, less able to access credit and employment".

CDRI's rural land titling baseline survey report (Ballard and So 2004: 20–21) also identified a wide range of indicators with which to assess the gender impact of land titles. The study observed that "land titles may have a significant and positive effect on women, especially those who are single heads of households, to the extent they help protect vulnerable people from losing their land due to grabbing or other forms of expropriation". To this should be added issues pertaining to inheritance and the degree to which women contribute to decisions concerning the household economy, including assets such as land. The report went on to say:

Land titles would also have an important impact on women to the extent that they enable people to secure loans with which to invest in agricultural production or other income generating activities (e.g., small businesses) ... Some women, however, may wish to invest in other income generating activities in addition to farming, or switch out of agriculture altogether. In this sense, then, land titles may enable more single women to gain access to larger amounts of credit at more favourable terms, though this would not necessarily imply an increase in agricultural investments. At the same time, land titles may enable other women to sell their land more easily for a better price, while switching to non-farm [employment] opportunities. Though this might show up as an increase in landlessness among single women, it may not necessarily represent a negative outcome if viable employment alternatives are available. In many respects, similar observations may also hold for Phnom Penh and other urban areas, including Siem Reap and Serei Saophoan.

2.2 Land Titling in Urban Cambodia

A limited but growing body of literature concerning land-titling impacts in Cambodia is now emerging, largely spurred by various reports for LMAP and its donors. Ballard and So (2004) discuss the expected benefits of land titles in the rural sector while presenting key baseline data that will be used in an eventual impact assessment. Deutsch (2006) assessed beneficiary attitudes toward land titles in general and found that people value this initiative a great deal for the stronger sense of tenure security. Markussen's (2007) brief spot check on land-titling impacts reaffirmed key issues raised in earlier reports and highlighted several emerging issues (e.g., non-use of official registry in land transactions) that will be addressed in the follow-up study to CDRI's rural and urban baseline surveys. The potential benefits of land-titling are also discussed extensively in the World Bank's *Poverty Assessment* (2006) and *Equity and Development Report* (2007). However, except for several references in the 2006 beneficiary assessment, these reports largely concern rural land titling and do not specifically address the potential impacts of land-titling in urban Cambodia.⁷

The literature that addresses urban land tenure and land-titling matters largely concerns informal communities in Phnom Penh municipality. For example, URC (2002) examines the record on evictions and relocation in Phnom Penh since 1990, and observes that the "greatest benefit settlers have noted from relocation is the attainment of land security and tenure", although individual titles may be some years away. Payne *et al.* (2004) looked specifically at different land tenure arrangements and concluded that the potential negative impacts of individual land titles in informal communities probably outweigh the potential benefits. The proposed alternatives to individual titles include short-term communal land rights and communal leasing, as well as planned relocation under certain circumstances. The issues pertaining to land tenure in informal communities are discussed in more detail in Section 5A.4.

Aside from the work concerning informal communities, there has not been much literature concerning local experience to guide the development of an urban baseline survey. This urban baseline survey report represents an initial effort systematically to identify and discuss the expected benefits of urban land-titling and discuss ways to approach research design, implementation and analysis.

2.3 Methodology

A review of the literature also reveals a variety of methodological challenges concerning social and economic impact assessments of land-titling projects. Payne *et al.* (2007) identify two major areas of concern in this regard. First, the identification and measurement of various factors require "an ability to measure individual variables, which are themselves difficult to quantify". A second set of problems concerns "time-lapse issues" because short-term impacts may vary from medium- and longer-term impacts, and "tracing problems" concerning residents who leave a particular area after titling programmes have been implemented. Moreover, many anticipated impacts, such as changes in land use patterns, involve complex interactions of many variables that must somehow be sorted out (Ballard and So 2004). Many of these same challenges are evident in the research hypotheses that guided this baseline survey, as discussed in the following section.

⁷ Both reports use econometric tools to analyse data from the 2004 Cambodia Socio-Economic Survey (CSES) to make strong assertions concerning the significant impact of land titles in the rural sector in terms of raising crop yields, land values and household consumption. They argue that the main reason is that "well-defined property rights improve the appropriability of returns" (WB 2006: 86).

Chapter 3. Research Hypotheses

This section discusses the research hypotheses that predict the expected land-titling impacts that will be used to guide the subsequent follow-up impact assessment. This discussion draws on the literature discussed above as well as data and observations from the earlier survey in downtown Mittakpheap (Sangkat 2) in Sihanoukville, where 99 residences and small businesses were surveyed in early 2004. The hypotheses discussed below identify the main indicators for assessment and helped structure the survey used for residential households and SMEs used in the urban baseline survey.

The research theory predicts that land-titling impacts will be significant in urban areas characterised by increasing residential, commercial and industrial investments and active or emerging land markets. Key indicators concerning investments include the number of new housing and home improvements, new business start-ups and expansions, changes in land use patterns and demand for credit. Key indicators concerning land markets include the frequency of land transactions, use of the official registry, frequency and type of land disputes and land values relative to plot and building characteristics. The most active urban land markets are likely initially to be located along main roads with good access to public utilities and services.

At the household level, the research theory predicts that land-titling impacts are also dependent on the diversity and level of income sources and capital assets, as well as the size and location of land or other property. These factors influence the household's access to investment credit and capacity to withstand or cope with shocks and emergencies. Holding other factors constant, high potential impact households with more income, capital assets and favourably situated landholdings will tend to benefit more than low potential impact households with less income, fewer assets and less favourably situated landholdings. Unlike rural areas, this hypothesis predicts that female-headed households may also benefit comparably with male-headed households, although this may depend on location and access to credit.

A simple model can, therefore, be constructed with which to make predictions about likely impacts when location and household factors are combined. For example, HPI households⁸ will benefit most in high capacity areas situated along main streets characterised by better access to utilities and services. Conversely, LPI households will benefit least in low capacity areas situated off the main roads, with less access to utilities and services. In between are HPI households in LCAs and LPI households in HCAs. The degree of impact in these two categories may depend on where and at what rate urban infrastructure and access to utilities and services improve. This in turn suggests that overall land-titling impacts may be undermined by the absence of strategic development master plans and enforceable zoning regulations. It also suggests that land-titling effects can be enhanced by strong enforcement of property rights in the face of encroachment.

These observations generate a series of testable hypotheses that structure the research methodology. As in the rural baseline survey, these hypotheses can be tested using quasi-experimental methods that compare household data from the current baseline survey project

⁸ Households in all three urban survey areas were divided into quintiles according to annual per capita consumption. Section 4.4 explains the reasons for this approach.

(T0) with data that will be collected at a later point $(T_0 + x)$ in both project and non-project areas.⁹ The analytical framework that emerges from a consideration of HPI and LPI households within urban HCAs and LCAs provides a robust tool with which to assess the potential economic and social impacts of land titles as predicted by theory and empirical research elsewhere.

Certain economic impacts of the land-titling programme are expected to be observed in both urban and peri-urban areas in and around Phnom Penh, Siem Reap and Serei Saophoan. In the peri-urban areas, land-titling may affect the scope and scale of agricultural investments and land improvements. In both the peri-urban and urban areas, land titles may stimulate housing improvements and business investments. Such effects may vary according to factors such as land size and use, physical infrastructure and social services. Land-titling is also expected to affect both the credit and land markets, including the supply and demand for conflict resolution, and the use of the official registry to facilitate and record land transactions.

3.1. Residential Land and Property

The research theory predicts an overall increase in new home construction and housing improvements following the issuance of land titles. For example, a World Bank housing policy paper of 1993 suggests that tenure security and property rights are important factors influencing housing demand and that insecure tenure leads to under-investment in housing and reduced housing quality.¹⁰ Key indicators include land prices in residential areas, rental prices, improvements in housing conditions and an increase in the number of available housing units.

The research theory predicts, at first glance, that there will be a greater share of new housing and housing improvements among middle and upper income groups with plots located on paved main streets, while there will be fewer such improvements among the lowest quintiles. One might also expect that the expenditures for housing improvements among the upper quintiles would be, on average, higher than such expenditures among lower quintiles.

Household survey data from Sangkat 2 in Sihanoukville, however, do not fully support such predictions. First of all, there were a total of 29 housing improvements reported by the survey group for the three years prior to the survey. Of these, only 20 percent occurred on paved roads, while an equal number occurred on roads with limited access. Most improvements occurred on one-way access roads. Second, the number of housing improvements was fairly evenly distributed across each income quintile, with the middle quintile on one-way access roads being the most active group. Third, the average value of housing improvement expenditures was less among the upper quintile households (USD352) than the lowest quintile households (USD752), while the middle three quintiles reported average expenditures between the two.

One possible explanation for this pattern is that there may not be much scope for further improvements in the high-value houses, other than fencing, furnishings and appliances. There may be much more scope for improvements and expansion in lower value homes, particularly those owned by middle or upper income households. For example, those houses that also serve as shops on the ground floor and are located on main streets probably cannot be improved in terms of building materials because they are constrained by being adjacent to other buildings. Some people, though, may be able to build vertically by adding additional floors. Smaller houses located on side streets and made of thatch or wood could be upgraded more easily with better materials and better fencing.

⁹ The experimental nature of the urban baseline survey follows the same basic framework used in the rural phase of the baseline survey. See Ballard and So (2004).

¹⁰ Referenced in Payne (2000).

It is also important to bear in mind that certain improvements, such as furnishings and appliances, may be more the result of increased income than of land titles. The key indicators concerning home improvements and housing investments should be the use of better building materials and expansions of floors and rooms, as well as any construction of new buildings and better fencing.

3.2. Commercial Investments and Property

The research theory predicts an increase in the scope and scale of new business start-ups and expansions. The key indicators for investment would be some change in the number and types of businesses, as well as changes in the amount of reported business investments. People may also be motivated to make structural or cosmetic improvements in the location where the business is located. As a result, the purpose of certain business-related borrowing and expenditures will be relevant. New business start-ups could involve people entering business for the first time, or those switching from one business to another. Business expansions could include hiring more labour, adding more floor space or adding new service units, depending on the nature of the business.

The research theory predicts that increased investments in a variety of activities will be observed across all income groups. Among lower income groups we may observe such increases in small trade, transportation and light manufacturing. Among the upper income groups, we may see increases in guesthouses, restaurants and other services in terms of new business start-ups, expansions or improvements, as well as trade and manufacturing. As noted above, such increases may not correlate with increases in formal credit.

There is some indication of a linkage between land ownership and business operation. For example, Kang (2005) found that 75 percent of 648 micro, small and medium enterprises MSMEs)¹¹ owned the land upon which their business was located. Of the landowners, about 91 percent (444 enterprises) had land titles. However, of the landowning businesses with land titles, 73 percent were located outside the provincial towns, thus indicating that only 27 percent of those titled landowning businesses were located in town. Landownership among MSMEs may be more of a rural than an urban phenomenon. What is not clear, though, is whether people obtained land titles after starting up their businesses in order better to secure their investment once it was made, or made the investment after having a title because they felt their land tenure was more secure.

A similar set of questions also arises concerning the data from Sangkat 2 in Sihanoukville. Of the 132 residential parcels, 37 (about 28 percent) were titled. Since many parcels featured dual uses (i.e., residence and business), we can assume that about 25 percent of the MSMEs involved titled land, which is fairly close to Kang's MSME findings. Again, it is not clear which way the causal arrow runs between land titles and business investments.

For business start-ups, Kang also found that investors used own sources and savings in 84.6 percent of the cases, and that such sources were the main source of financing (50 percent or more) for at least 75.3 percent of business start-ups. After that, investors used credit from external sources, including family and friends, as well as various forms of supplier and customer credit, to finance business start-ups. Investors rarely used credit from banks or NGOs. This pattern closely resembles the pattern found in the smaller survey in Sangkat 2 in Sihanoukville concerning 90 "sources of capital" used to finance business start-ups. There, over 93 percent of the sources (84 in total) involved own sources, including the household's own savings and income, income from land sales and informal loans from family networks. Some respondents may not have included a specific amount for a business investment because the use of one's own resources may not be perceived as a discrete business-related activity,

¹¹ Enterprises were divided into three size categories: micro (one to two workers), small (three to five workers) and medium (10 to 100 workers).

but rather as part of the family's ongoing cash flow cycles (income and expenditures over time). This aspect of household resources accounts for at least some of the discrepancies between the number of reported SMEs, reported investments and investment sources.

In the MSME survey, Kang concluded that a well-developed banking system is not necessarily required for starting up new businesses, despite the conventional wisdom of traditional development economics. The security of contracts associated with borrowing from formal institutions is offset by high levels of personal trust associated with informal borrowing. This being said, there is general agreement that better access to formal credit could and should play a larger role in financing capital investments. For example, Kang observed that 52 percent of 137 enterprises that sought a bank loan were refused. Of this number, 60 percent were refused because of a lack of collateral. This may help explain, at least in part, why so many people rely on other sources of capital for financing business startups, including those who do not apply for loans because they assume they would be denied.

Seventy percent of the 648 enterprises surveyed ranked a lack of capital as the biggest constraint on business expansion, regardless of location, size or type of enterprise. The lack of capital seems to be associated with an "insufficient accumulation of profit". Kang views the development of a sound banking system and legal security of contracts as a key component for solving investment problems associated with a lack of capital. Therefore, an increase in the demand for capital that is stimulated by land titles that can used for collateral may also have the positive effect of stimulating an increase in the supply of capital.

3.3. Credit

Any discussion concerning investments for housing or business development, as well as for agriculture, raises questions about how such investments may be financed. The research theory predicts that people in both peri-urban and urban project areas will use land titles as collateral to obtain credit from formal lending institutions. If so, we should then observe changes in borrowing behaviour as people either (a) shift away from informal institutions toward more formal institutions or (b) diversify their credit portfolio by borrowing from formal institutions in addition to informal sources (Ballard and So 2004).

We should also observe a shift in the number, size and intended use of loans as people obtain larger loans for productive investments more frequently. Because the amount of a loan may depend on collateral assets and resources for managing repayment, the research theory predicts that households with more capital, income and land may be able to obtain larger loans than those with fewer such assets. This set of hypotheses assumes that formal credit markets perform reasonably efficiently in a particular area (i.e., transaction costs are low).¹²

There are in fact clear indications that the formal credit sector is growing in both urban and rural areas. For example, Acleda has increased from 14 branches with offices in 34 districts in 2000 to 15 branches in 104 districts in 2004, including 25 district offices in the Phnom Penh-Kandal branch. Also, in 2004 Canadia Bank was granted more than USD10 million worth of assistance to expand mortgage operations and increase lending to small businesses, primarily in urban areas. In 2005, ANZ Royal opened operations in both Phnom Penh and Siem Reap that include active lending for businesses engaged in real estate and housing development. Micro-finance institutions (MFIs) are also expanding their operations throughout the country, including urban areas (e.g., Siem Reap, Serei Saophoan).

In Sangkat 2 the survey group as a whole obtained a total of 41 loans in either cash or gold during the six months prior to the survey. Female-headed households obtained about 25 percent of the loans, which was roughly in keeping with the distribution of female-headed households in the general survey population. Over half (21) of the loans were obtained by the two lower income quintiles. Thirteen of these loans were for food shortages or health. Five,

¹² See Ballard (2004) for more discussion of this set of hypotheses.

however, involved business activities in the lowest income quintile. About 25 percent of the loans (10) were obtained by the two upper income quintiles, including five for business activities and five for home repairs or construction. Overall, about 30 percent of the loans were for business-related activities. Another six loans, about 15 percent, were for housing and home construction, while 15 loans, about 38 percent, were for food or medical reasons.

As in the rural areas, there appeared to be more borrowing in the informal sector in Sangkat 2, where 29 (about 70 percent) of the loans were obtained. Eighteen of these loans, about 62 percent, were obtained by the two lowest quintile groups. The two upper quintile groups obtained 11 loans in the formal sector, including eight from Acleda and three from MFIs. The remaining eight loans were taken by the middle quintile (including five from the informal and three from the formal sector). The upper two quintiles also took six loans from the informal sector. This suggests that upper income groups tend to have a wider range of options in access to credit.

The seven Acleda loans all required some form of collateral: two involved receipts for land certificate applications, three involved certificates, one involved an "other asset", and two required some form of group asset. One MFI loan involved a land certificate for collateral. None of the remaining loans, including those in the informal sector, involved collateral. This helps explain the preference for, or at least the level of, borrowing in the informal sector, especially among those in the lower quintiles. On the other hand, it also suggests that some form of land documentation is essential in order to obtain a loan from the formal sector.

Although the research theory predicts that overall formal sector borrowing across all income groups will increase as a result of land titles, the distribution of benefits from such borrowing is difficult to predict. Upper income groups may not borrow much more often than now. One reason is that many of them already have certificates for at least some of their plots, so the LMAP titles will not change their documentation status all that much. Another reason is that they have other assets and more income with which to invest. However, they may borrow larger amounts depending on the use of the loan. This may depend more on their confidence in the medium- to long-term business and investment climate.

On the other hand, at least some households in lower income groups may use their titles as collateral to begin borrowing from the formal sector. Lower quintile households may borrow smaller amounts than the upper quintiles for such activities as small-scale business investments or housing improvements. One reason they may take smaller loans is that they have less income with which to repay. At the same time, lower income households may also continue to borrow from the informal sector for managing problems such as health care and food shortages.

3.4. Land Markets

The research theory predicts that over time titled property will be transferred through land markets to more economically productive uses. We should therefore expect to see shifts in peri-urban and urban land use patterns, including investments in residential and commercial property and other improvements (e.g., better fencing). In both the peri-urban and urban sectors, we should also see the expansion and growth of better functioning real estate markets.

3.4.1 Land Prices

As land use shifts in the direction of more economically productive uses, land prices (sales and rentals) of residential and commercial property will increase. We expect that land prices will increase at a faster rate where land is situated along main roads with public utilities and services. The rate of price increases, however, should be higher in project areas than in control areas, because prices for documented land tend to be higher than for undocumented land. This being said, in countries such as Cambodia, it is also important to distinguish among the different types of paper used to document land ownership. In Sangkat 2 in Sihanoukville, the distribution of residential land shows a similar pattern to that of the rural households: the three upper quintile groups tend to own more plots than the lower two quintiles. Moreover, their plots tend to be larger and generally have a higher value per square metre, except for the second quintile, which has the highest value per square metre. As a result, the average value of the plots owned by each household tends to increase according to income quintile. The difference between the lowest and highest income groups is quite striking: USD5948 per plot vs. USD35,286 per plot (see Table 3.1). Such differences can be explained in part by the location and value of the respondents' houses. There is a clear contrast between the value of the lowest and highest quintile houses: USD4087 vs. USD15,905. One-third of the upper quintile houses are located on a paved road, while 40 percent of the lowest quintile houses are located on roads with only bicycle or motorcycle access.

Annual Income	HH	Plots	Tot.Area	Tot.Value	Plots/HH	Area/Plot	\$/SqM	\$/Plot
(x 10,000 riels)			(SqM)	(\$)				
165-380	19	21	7,609	124,900	1.10	362.33	16.41	5,948
391–584	20	25	8,109	392,000	1.25	324.36	48.34	15,680
600-919.5	20	32	14,581	572,179	1.60	455.65	39.24	17,881
920-1620	20	25	12,064	460,600	1.25	482.56	38.17	18,424
1770-7200	18	29	24,096	1,023,300	1.61	830.89	42.46	35,286
Total	97	132	66,459	2,572,979	1.36	503.47	38.71	19,492

Table 3.1. Smalloukyme Lanunolung value Summary	Table 3.1:	Sihanoukville	Landholding	Value	Summary
---	-------------------	---------------	-------------	-------	---------

Source: CDRI Baseline Survey, 2004

3.4.2 Transactions

Land transactions include both sales and purchases by the households in the survey population. One would expect that land markets located near commercial and administrative centres and/or along main roads would have a greater frequency and volume of land transactions of greater value than markets located further away from commercial and administrative centres and/or paved roads. This prediction assumes that macro-economic and political conditions remain stable.¹³ However, this expectation may not hold in cases where speculators are buying and selling land in peri-urban or other areas slated for development.

In Sangkat 2, the survey group had purchased 89 residential land parcels and 12 other parcels since 1989. Most of these transactions took place prior to 1997. There had been 19 reported land sales among the survey group since 1989, about one sale for every five households. Thirteen of these transactions took place prior to 1997, and the remaining six took place in 1997 or since. The most recent reported sales were two in 2000, one in 2001 and three in 2000. Of the 19 sales, 12 involved residential land, one involved rice land, four involved *chamkar* land, and two involved other types. The upper two income quintile groups accounted for 10 of the land sales, while the lower two quintiles accounted for only two. The upper three quintiles have more land and therefore more economic flexibility than the two lower income quintiles.

As noted in the above discussion concerning rural land transactions, people buy and sell land with or without titles. The number and frequency of transactions within the survey group probably will depend more on land use and values, which are more a function of the business and investment climate and the availability of land for sale. Many households own only their one residential plot, many of which double as a place of employment or trade. As long as

¹³ Chan and Acharya (2002) have observed that the volume of land transactions seems to mirror macro-economic growth patterns.
people can sustain their livelihoods, they are not likely to sell their only land. People are also motivated to retain at least some land in order to pass it on to their children. On the other hand, households with multiple plots, particularly those among the upper income quintiles, may be more likely to sell land for investment purposes, as already observed.

According to the data from Sangkat 2, there may be more land purchases than sales among the survey group, particularly among those households with higher incomes looking to invest savings in urban real estate and/or rural farmland. The pattern of land transactions may involve fewer sales to people from outside the survey areas and more purchases of land both in and outside of the areas by the survey groups.

3.5. Land Administration

The research theory predicts an increase in the percentage of transactions that are facilitated through the official registry system, particularly in more active markets where land values are increasing. The key indicator would be an increase in the percentage of land sales and inheritances that are recorded and facilitated through the official registry. However, this may depend on several factors. First, the fees and other costs associated with official transfers must be reasonable. Of particular concern is the fact that some buyers may still wish to avoid paying the four percent land transaction tax. Second, the degree of confidence that buyers have in the security of the LMAP titles depends in part on the ability and willingness of local authorities, including the courts, to enforce objectively the tenure rights represented by the titles. Third, the degree to which people intend to use the titles as collateral for loans may depend on their confidence in the overall business environment and political stability. Finally, people must understand the rules and procedures governing the use of the official registry, which in turn depends on the quality of information available.

The degree to which people use the official system may also vary according to the capacity of the household to pay related fees and taxes. Households with more income and wealth may be more inclined to use the official registry than less well-off households. These predictions assume that (1) transaction costs associated with official registration will be lower than they are now; (2) people will have more confidence in the security of tenure than they do now; and (3) people will have sufficient knowledge of the procedures and the capacity to access the system. Again, the degree to which all households use the official registry for land transactions will ultimately depend on the government's ability and willingness to enforce these and other kinds of procedural requirements.

3.5.1 Land Conflicts/Dispute Resolution

The research theory predicts that secure land titles will, over time, reduce the volume and frequency of land disputes by clarifying ownership, parcel boundaries and transaction procedures. The experience in Cambodia (So *et al.*, 2002), however, suggests that clarifying boundaries and ownership may initially stimulate new conflicts or arouse dormant disputes. This may very well be the case in urban areas where the growth of the real estate market is being spurred by rapidly increasing land values. In such a situation, one would expect the scope and scale of conflicts to escalate in areas where land values are rising. In the short to medium term, we expect the volume of disputes, particularly regarding boundaries and interfamily transfers, to increase. However, the number of disputes should decline with the passage of time once the LMAP titles are issued. It will be very important to monitor how local authorities and the courts mange dispute resolution involving LMAP titles. If tenure security in the form of land titles is objectively upheld and enforced by the authorities, this will represent a great step forward for the rule of law.

The rural land-titling baseline survey conducted in 2004 may provide some guidance on this question. The survey found 60 land conflict cases during the two years prior to the survey. Some 38.3 percent of the cases involved boundary conflicts with neighbours, followed by

conflicts with relatives (32.0 percent) and with other villagers (21.6 percent). About 10 percent involved encroachment on the part of authorities or powerful people. The report observed that most land disputes are local in nature and that the scope and scale of conflicts are highly situational. It also noted that the low number of conflicts found in the survey may also be due to the fact that LMAP titling has tended to take place in stable areas where customary rights seem to be working well, and that many such conflicts have long been resolved.

3.6. Gender

The research hypotheses predict that land titles will have a wide range of impacts on genderrelated matters pertaining to land. Many of the same indicators identified in the rural baseline survey, such as landholdings and capital assets, are also applicable to assessing the impact of land-titling on gender equity. Other indicators include access to affordable credit and investments in housing and business start-ups and expansion. It is expected that femaleheaded households will obtain an increased number of loans that are invested in housing and business improvements. However, the rate at which female-headed households obtain loans may depend on a variety of circumstances, including household well-being.

It is also important to note that the degree to which female-headed households benefit from land titles is likely to vary according to the amount of assets they control. For example, the rural baseline survey of 907 households in 2004 found that female-headed households on average had smaller landholdings than male-headed households (1.17 ha vs. 1.75 ha) and fewer plots per household (3.78 vs. 4.44) that were smaller in size (0.30 ha vs. 0.39). Moreover, female-headed households consistently have fewer other assets, including livestock, durable assets, non-farm fixed assets and non-machine farm assets. Female-headed households also had fewer adult workers than male-headed households (3.4 vs. 4.4).

These disparities are reflected in the mode of land acquisition. The survey showed that female-headed households had a much higher percentage of plot acquisitions from the state (70.9 percent) than did male-headed households (51.3 percent), while the percentage of plot acquisitions through inheritance was much lower for female-headed households (11.2 percent) than for male-headed households (24.6 percent). The percentage of plot acquisitions by purchase and clearing was also lower for female-headed households. These patterns suggest that female-headed households are less able to acquire additional land. The fact that they have less labour and fewer assets than male-headed households indicates a constraint on the amount of land that can be farmed, while less income implies a constraint on buying land.

The research hypotheses also predict that women's tenure security will be strengthened. One way to assess this will be to examine the outcome of cases involving the death of a husband or divorce and evaluate the extent to which the woman's land rights have been upheld. Another way to assess this would be to examine cases of conflict involving households headed by single women. While household surveys may enable surveyors to categorise and count the number of such cases, a qualitative approach will also be required to understand better the social and legal dynamics of such cases.

Chapter 4. Methodology

This section provides a detailed account of the methodology for site selection and household sampling, as well as a brief account of the implementation strategy. Generally speaking, the selection criteria primarily concerned the diversity and dynamics of land use patterns and the expected degree of change in a particular area. In all three survey locations, one set of areas included households that presented a mix of residential and commercial uses. A second set included households in peri-urban transitional areas spanning a mix of agricultural, residential, commercial and potential industrial uses. A third set of areas where land titles were expected to be issued at a later date was intended to serve as a control. In principle, the character and composition of the control areas should resemble the two experimental urban areas (i.e., a mix of residential and commercial properties) as closely as possible.

4.1. Survey Site Selection

The provincial selection was based on a variety of factors, including strategic location and infrastructure, as well as potential development trends that will likely impact land markets and land use patterns. Phnom Penh was selected first because of its important economic role in the country and strategic location, as well as the fact that economic and infrastructure development is having profound impacts on housing and real estate markets throughout the municipality. Siem Reap town was selected because the rapid economic growth in the area due to expanding tourism is having a profound impact on land markets and land use patterns. Serei Saophoan, in Banteay Meanchey, was chosen largely because of its strategic location along National Roads 5 and 6, linking much of north-west Cambodia to the rest of the country, as well as transporting many Cambodian products to markets in Thailand. Both tourism and trade along Roads 5 and 6 are likely to increase when the highways are improved in the near future.

4.1.1 Commune Selection

The commune site selection was largely determined by LMAP work schedules. In Phnom Penh, LMAP was working in 16 communes in the six districts at the time of the survey, with plans to issue land titles soon after. Six of those 16 communes—Boeng Reang, Boeng Keng Kang 1, Tuek L'ak 1, Olympic, Chaom Chao and Khmuonh—were selected. Five others communes—Tonle Basak, Boeng Trabaek, Prek Pra, Cheung Aek and Prek Lieb—were chosen as control areas based on the assumption they would be titled last and thus provide a useful time frame for measurement. In Siem Reap, Sala Kamraeuk and Siem Reap commune were the only areas where LMAP was planning to work in the foreseeable future. Sala Kamraeuk was therefore chosen as an urban area and Siem Reap was selected as a peri-urban area. In Serei Saophoan, the three communes of Kompong Svay, Preah Ponlea and Ou Ambel were selected.¹⁴

¹⁴ At the time of the survey, LMAP had not yet begun titling in Banteay Meanchey. LMAP will begin work in the province in 2008 with support from CIDA. The question of the Serei Saophoan baseline survey is discussed in Section 7.

4.1.2 Village Selection

The objective was to achieve as much diversity within the village sample as possible, including both dynamic HCAs and less dynamic LCAs. At least two villages per commune were chosen for study. The initial selection was based on consultations with various people including LMAP and commune council staff, as well as pre-survey site visits. Final decisions were made following discussions with village chiefs and/or other community members about the feasibility of conducting surveys in the area and the likely cooperation of local officials and people.

The criteria for village selection included land market development and activity, land use, infrastructure, population density, proximity to markets and administrative locations and other economic activity. Efforts were also made to avoid surveying in adjoining villages in order to achieve geographic diversity within the commune. Finally, the site selection was also influenced by LMAP work schedules in the survey areas.

4.2. Household Sampling

Within each survey area, households were chosen by stratified random selection. First, the different types of land use were estimated for the survey area through consultation with local officials and drive-by observations. Second, the survey locations were selected according to street type (e.g., main roads, secondary streets with two-way access, tertiary streets with one-way access). Third, as in the rural survey, special efforts were made to ensure a fairly even distribution of well-being groups and to include female-headed households.

Two different approaches were then used to select households according to the information available in the village. In villages where household lists existed, including eight villages where LMAP had already worked, the baseline survey project teams counted off every nth name from the list. In villages where there was no list of households, the teams selected every nth house.

4.3. Survey Instrument and Fieldwork

CDRI employed a revised version of the survey instrument that was used in the urban survey work in Sangkat 2 in Sihanoukville during the first phase of the land-titling baseline survey, taking into account the lessons from the work there, as well as special circumstances in other urban areas (e.g., people's time constraints). The survey instrument was pre-tested twice in Phnom Penh, and the lessons incorporated into further revisions of the instrument. The same survey instrument was then used in Siem Reap and Serei Saophoan. (See Annex A.)

CDRI engaged 12 field enumerators who already had some urban survey experience in landtitling in Sihanoukville. They also had gained extensive prior experience with land-titling surveys during the rural phase of the baseline survey project. The enumerators also received three days of training and orientation, as well as two days of pre-testing the survey instrument, which involved collaboration with LMAP survey teams working in Phnom Penh. The enumerators were divided into four teams of three. Each team had an experienced leader who was responsible for reviewing the data each day to ensure quality control. The team leaders also helped to coordinate work with local officials and to schedule appointments as needed.

The data coding for SPSS¹⁵ had already been developed in conjunction with the baseline survey work in Sihanoukville. However, the data codes were modified to accommodate adjustments in the interview questionnaire. The most diligent field enumerators with prior data entry experience were selected for this phase of the data work.

¹⁵ Statistical Package for the Social Sciences.

Each enumerator was asked to give their perception of the accuracy and reliability of the information obtained in the interview. Table 4.1 shows that the enumerators' assessments of the interviews were quite consistent across all three areas. The table also shows that the quality of the urban interviews was rated somewhat higher than the rural interviews.

Location	Very Good	Fairly Good	Moderate	Low	Total Number
Phnom Penh	6.7	73.7	19.1	0.5	1663
Siem Reap	4.1	72.0	22.4	1.5	536
Serei Saophoan	2.2	74.6	23.1	0.2	507
Total	5.3	73.5	20.5	.6	2706
	Very Good	Good	Medium	Weak	
Rural/Sangkat 2	4.4	56.3	35.4	4.7	1232

Table 4.1: Interview Quality Assessment (percent)

4.4. Limitations

One of the most significant constraints was the limited amount of time that urban residents were willing and able to provide for an interview. While it was possible to interview rural residents for as long as one and a half hours, the pre-testing in Phnom Penh suggested that 45 minutes might be the maximum time that people would allow for the interview. With this in mind, the survey team designed the instrument so that the interviews could be completed in 40–45 minutes. This objective, although imminently practical, posed a serious challenge in designing an efficient instrument in which the optimal amount of reliable information could be obtained in the allotted time.¹⁶

A second important constraint concerned the reliability of information provided. For example, during the first round of pre-testing, the enumerators tried to obtain information about household income. Based on this experience, they felt that many respondents were reluctant to provide information about income and when they did so were likely to understate their income by varying degrees. The survey instrument was then redesigned to collect information about household consumption expenditures over the year prior to the survey. During the second round pre-test, the enumerators found this was easier to do in the limited amount of available time for each interview, and in their opinion also provided more reliable information.

Despite the excellent cooperation that the survey teams received from municipal and local officials, as well as most households, some households refused to grant interviews. As these tended to be wealthier households or those of ranking government officials, it must be kept in mind that there may be some selection bias, resulting in the wealthiest households being under-represented in the sample.

Another limitation was the fact that only the actual owners of the property in question were interviewed. The survey therefore more specifically dealt with the impact that titles have on owner decisions and actions, such as making home improvements or obtaining credit, and may not provide a broader perspective on the impact of titles on land use in general, especially in terms of rentals. Also, and perhaps more importantly, the survey does not capture or reflect a number of important issues that are of concern to households that either legitimately occupy land or occupy state or private land and are, as a result, subject to eviction and/or relocation. This limitation reflects a broader limitation of the LMAP in general, as it is designed to provide titles to owners and is not specifically concerned with issues of housing and access on the part of the poor in informal communities, although the titling programme may indeed impact on such issues in a profound way. There is a real need

¹⁶ One benefit of the shortened survey instrument was an increase in the number of interviews that were completed, which resulted in a larger sample than originally planned.

for a comprehensive study specifically designed to address this difficult yet important set of issues. Special efforts were made to identify and include survey areas with many low-income households in order to understand how land titles affect low-income landowners.

Finally, the timing of the LMAP fieldwork in relation to when land titles were to be issued proved difficult to arrange, because the process includes many steps, several of which are subject to delay (e.g., surveying, adjudication). The basic strategy in identifying baseline survey areas was to select places where titles were to be issued soon after the survey was completed. The narrower the gap between the survey and the issuance of the title, the less chance there is for intervening variables to distort or otherwise affect the impact analysis. For example, if titles are issued a year following the survey, the baseline measurements would be distorted. This is, in fact, the case in Serei Saophoan, where the survey was conducted prior to the LMAP programme even starting. When land titles have been issued relative to the survey dates could have important methodological implications for the follow-up survey. This will be discussed in more detail in the section on recommendations.

Chapter 5. Phnom Penh Municipality Baseline Survey

This section provides a background overview of the current land situation in Phnom Penh. The first part discusses some of the key issues pertaining to land-titling and management, including planning and zoning, infrastructure rehabilitation and development, real estate and housing markets, informal communities and peri-urban areas. The second part presents and discusses some of the key Phnom Penh baseline survey data and concludes with a summary of the main points.

5A.0. Background and Current Land Situation

The area of Phnom Penh is approximately 375 square kilometres. The municipality is administratively divided into seven districts and includes 76 communes. Four districts make up the urban centre, covering only 28 square kilometres, while three districts make up a periurban area of 347 square kilometres. The population of the city in 2004 was estimated to be about 1,042,000, with a total of 205,042 households, about 5.08 persons per household. In 2004, the population density of the urban areas was estimated at about 16,688 per square kilometre, while that of the peri-urban areas was approximately 1289 persons per square kilometre. After natural increase, estimated at 3.2 percent per annum, the most significant factor for population growth is immigration from rural areas, estimated to increase the population by 2.0 percent per annum. There is also rapid growth in new construction and increasing commercial concentration. These factors are placing new demands on land, housing, infrastructure, utilities and services, with adverse effects on traffic, drainage, security, the environment and aesthetics.

Recent analysis of 2004 Cambodia Socio-Economic Survey (CSES) data shows that per capita consumption, a useful measure of well-being, varies greatly between urban and rural areas within regions, including Phnom Penh, where overall per capita consumption exceeds the national average.¹⁷ In the peri-urban areas, per capita daily consumption was approximately 6130 riels, while it was about 9633 riels in the urban areas. According to the 2004 CSES, inequality in Phnom Penh is high, as suggested by a Gini co-efficient value for real household consumption of 0.37. The World Bank's recent poverty assessment (2006) observes that inequality may actually be higher than this in urban areas, particularly Phnom Penh, because the poorest segments of the population are likely to have been underrepresented in the survey sample. The assessment also observes that some of the most pressing issues facing the urban poor are employment and the high costs of services, such as electricity and water, while security of housing tenure is one of the most "overriding" threats to livelihoods and well-being. The poor, who are often crowded on marginal state public land, tend to lack access to basic services, while those who may have nominal tenure rights are vulnerable to efforts to displace them in the face of rapidly increasing land values.

¹⁷ The average monthly household income in Phnom Penh in 1999 was 1,139,553 riels, or USD298.70 at the current exchange rate (NIS 1999). This was considerably higher than the national average for Cambodia, which was 403,334 riels, or USD105.72. There was also considerable variation across households. For example, the lowest decile group averaged 357,778 riels per month, about 3.1 percent of total income, while the highest decile averaged 3,708,141 riels per month, about 32.4 percent of the total.

5A.1. Zoning and Planning

The sequential relationship between zoning and titling is an important concern, because landtitling impacts are optimised in urban areas where land use master plans and enforceable zoning regulations are in place prior to the titling. In Phnom Penh, as in Siem Reap and other urban areas, however, the private sector has largely determined land use patterns in the absence of master plans and zoning regulations. This represents potential inefficiencies associated with distortions in the distribution of public utilities and services that serve larger private interests rather than the public interest. As a result, it is possible that the issuance of land titles may exacerbate problems associated with the current mix of land uses in any given area. Without government zoning interventions and enforcement, the uncertainties of poorly planned land use mixes could drive up transaction costs associated with competition over public resources between private individuals, impede incentives for productive investments, particularly for small investors, and undermine the development of efficient land markets.

A master plan that outlines the municipality's development objectives through 2020 as well as zoning and land use planning was drafted in 2003 but has yet to be approved. Participants at a seminar concerning land use in Phnom Penh (So *et al.* 2001) observed that "the absence of a proper zoning and urban development plan has encouraged civic mismanagement in the city". For example, some public land and buildings had been sold illegally, and there were few structures and little land remaining for developing infrastructure and utilities. Moreover, land occupied for commercial purposes had not been productively used but was being held for speculation. High unofficial fees for land certificates and transfers discouraged people from obtaining formal ownership of their land or houses and encouraged illegal occupations and the production of fake certificates. The environment and aesthetics of the city were also being undermined by an excessive concentration of commercial activities in small areas and by illegal construction.

Generally speaking, the planned development and growth of the city are oriented toward the west because that area is not flooded during the rainy season, when the river rises. Planners have observed that Phnom Penh still has considerable areas for growth, including much of the 347 square kilometres in the three peri-urban districts. Although the general direction of development is known, the fact that zoning and master plans for land use are either undeveloped or otherwise not available to the public suggests that the development of the city has been haphazard and driven by individual private interests at the expense of the public interest.

Such haphazard growth and development contribute to poorly functioning land markets and undermine good governance because they encourage speculation by government staff or those with access to insider information. For example, if speculators acquire much or all of the land in a given area, there may not be sufficient space remaining for public services and/or infrastructure and utilities. The result will be the eventual need to adjust private and public land ownership and boundaries to accommodate necessary infrastructure. Such adjustments will be costly and time consuming, and will very likely generate conflict. Land use in underdeveloped areas should be demarcated in accordance with zoning regulations and urban development plans prior to titling.

5A.2. Infrastructure Rehabilitation and Development

During the Khmer Rouge period, Phnom Penh was virtually abandoned and a substantial portion of the urban infrastructure, including roads, sewage and drainage systems, canals, buildings and utilities, fell into severe disrepair. While many gains were made during the 1980s in getting certain utilities and services up and running, many parts of the city could not be served due to a lack of financial and technical inputs. It was not until the early 1990s and the availability of donor assistance that efforts accelerated to rehabilitate roads, sewerage and drainage and basic utilities throughout the city. Much progress has taken place since then. For

example, the city plans to complete paving all roads in the four urban districts by 2008, at which time it will begin paving roads in the three peri-urban districts.

The draft master plan for municipal development through 2020 envisions significant infrastructure development to integrate Phnom Penh into the mainstream of regional economic development by linking it with other major cities (e.g., Ho Chi Minh City, Bangkok) as well as other lines of trade (e.g., Pan-Asian Highway). For example, in Samraong (Dangkao district), the city plans to develop a dry port for commercial transport where the Battambang and Sihanoukville rail lines meet. A by-pass ring road will connect the dry port to National Routes 1, 4 and 5, while depots along the road will serve feeder roads to shuttle goods, thus reducing large truck traffic into the city.

The city also proposes the development of five satellite cities around the main urban centre in order to relieve the pressures of rapidly increasing population and economic growth. The locations include an area along Route 1 near Chbar Ampov, another area east of the city, an area north of the city near Pong Peasy, Samraong (north-west of the city) and an area near Tuek Khmao south of the city. The areas in Pong Peasy (i.e., Camko City) and near Chbar Ampov have already secured investor interest.

These and other development plans will significantly increase the demand for land in and around the municipality. Private investors will require secure ownership of large tracts of land in order to develop the proposed new satellite cities, while the municipal government will require more land to develop infrastructure, such as the dry port in Samraong and other transportation, including roads and bridges. Land titles will play a significant role in facilitating the transfer of land into the hands of private investors and the government through commercial acquisition. Such developments help explain the growth of an increasingly active real estate market throughout the municipality.

5A.3. Real Estate and Housing Markets

Land values have been increasing despite the general absence of land titles. According to one real estate agency, during the period 2000–07, land prices increased every year except 2003-04 as a result of a high volume of transactions in response to the rapidly growing demand for residential and commercial property throughout the municipality. Property is increasingly a favoured form of investment in both Phnom Penh and Siem Reap, from which investors expect a good return. During the period 2000–03, land prices increased by 10–15 percent, but declined by five per cent during 2003–04 as a result of the political stalemate after the 2003 election. Since then, however, land prices have risen sharply, by 25–35 percent during 2005–07; in some of the more competitive and lucrative areas, prices have increased by 30–80 percent.

Land values vary with a variety of factors, including shape and size of the plot, level (i.e., drainage), location, neighbourhood (i.e., comparable property nearby), utilities and services and amenities. For example, the most desirable shape and size of a plot for villa or townhouse construction is 20 x 30 metres, which allows enough room for some garden. For townhouse construction, a plot must be have at least 16 metres of road frontage and no less that 11 metres of depth. Other things being equal, land situated along a two-way paved road is valued more than land situated along an unpaved two-way or one-way street, or land that can be accessed only by bicycle or on a footpath. The type and quality of utilities and services available to the property are also important. Electricity supply, sewerage and drainage and clean drinking water are all important variables affecting the value of land.

Zone	July-Sept	March-April	%	Comment
	2006	2007	increase	
Α	400–950	850-1500	74.1	1 st Commercial Land
В	400-600	650–900	55.0	2 nd Commercial Land
В	250-400	500-750	92.3	3 rd Commercial Land
D	290-400	550-750	88.4	Prime Residential and residential
				(PR/R)
				BKK 1, Daun Penh, Market Zone
E	150-350	230–530	52.0	PR/R: 7 Makara. ORS 1 – 3*
F	120-280	220-500	80.0	PR/R: BKK 2-3, TSP 1-2, TBS 1-2,
				Olympic, DPO 1-3, PDK*
G	230-280	350-550	76.5	WP, SC*
Н	80-250	180-380	39.4	TTP1-2, TNT, BTB, PDT, BSL,
				TL1-3*
Ι	80-250	180-280	58.9	Tuol Kork

Table 5.1: Phnom Penh Land Value Estimates (USD/sq. m.)

* BKK = Boeung Keng Kang; BSL = Boeung Salang; BTB – Boeung Trabek; DPO = Phsar Depot; ORS = Orussei; PDK = Phsar Doeumkor; PDT = Phsar Doeum Thkov; TBS – Tonle Bassac;

TNT = Tumnup Tuk; TTP = Toul Tum Poung; WP = Wat Phnom

Source: Cambodia Estate Guide

Housing values vary with the area of floor space, the type of construction and quality of building materials. The age of the house, condition and furnishings also influence the value. Amenities such as protected parking also affect the value.

The booming housing market in Phnom Penh is reflected in the data on construction approvals. Table 5.2 shows that between 2000 and 2006 the overall value of construction approvals increased from USD205.4 m to USD323.3 m, an increase of 57.4 percent. During the same period, the value of villas and houses more than doubled, from USD16.4 m to USD33.1 m, and flats grew by 22.1 percent, from USD174.8 m to USD213.4 m.

	2000	2001	2002	2003	2004	2005	2006
Villas/houses	16.4	15.9	23.4	20.0	30.3	45.5	33.1
Flats	174.8	167.8	179.9	91.6	167.6	204.2	213.4
Other	14.2	12.6	16.6	87.3	65.6	109.1	76.8
Total	205.4	196.2	219.8	198.9	263.4	358.8	323.3

 Table 5.2: Construction Approvals in Phnom Penh (USD m)

Another indicator of increasing land values is the increasingly vertical built environment in urban districts. A a number of multi-storey housing units can be observed in Boeng Keng Kang and Tuol Kork, while multi-storey buildings in or near commercial areas are also increasingly common. In the peri-urban area, the built environment associated with urbanisation is also steadily expanding, albeit horizontally.

Prior to the LMAP land titling programme, about 25 percent of the owners of houses and land had been issued land certificates, significantly more than the national figure of 14 percent at the time.

5A.3.1. Condominium Property

There are a large number of co-owned immovable properties in Phnom Penh that entail a complex set of issues regarding registration and titling. In multi-storey buildings, some parts may be privately owned and used (e.g., ground floor shops or upper level flats), while other

parts may be commonly owned and used (e.g., stairwells, roofs). A draft circular on the registration of co-owned buildings and parcels defines a lot as a part of co-owned immovable property consisting of private parts that are attached to a certain percentage of common sections.

The parcel upon which a condominium is located will be registered as a parcel co-owned by each unit owner, while each lot in a condominium will be registered under the name of the lot owner. After registration, land titles will be issued to owners of each lot, along with information about the size, location and type of private sections and the share of common sections connected to it. According to the draft circular, the owner of a private section has the right to sell or otherwise dispose of it, while the rights governing transactions involving common sections are not entirely clear. It is, however, specified that certain rights concern common sections (e.g., new construction, joint ownership of common parts). Such provisions would also apply to parcels of land or one-story buildings where there are several owners of divided lots or units.

5A.4. Informal Settlements

In 2002, there were an estimated 472 informal settlements, consisting of 35,165 households and 175,825 people, excluding homeless households and seasonal labour migrants (URC 2002). A more recent study (Payne *et al.* 2004) found that there were more than 500 informal settlements in and around Phnom Penh, distributed on both public and private land. This study observed that "informal settlements in Phnom Penh occur when land is occupied without official approval or formal tenure status on one of three main categories of land". These categories are (1) settlements on state private land, defined as the property of ministries; (2) settlements on state public land, defined as land belonging to public services, such as railway stations, parks, lakes, rivers, streams and forests, representing about 30 percent of the developed area of Phnom Penh; and (3) settlements on private land, which account for about 65 percent of informal settlements.

Informal settlements in Phnom Penh and other urban areas have sparked considerable controversy. The city has tried to resettle people residing in informal communities to areas outside the city, beginning with forced evictions starting in Phnom Penh in 1990–91. Such efforts often fail to solve the problems associated with informal settlements, as many people eventually return because the resettlement sites have often lacked essential services, basic infrastructure and employment opportunities (URC 2002). More recently, some government planners in concert with some civil society organisations have proposed alternative means of addressing the problems associated with informal communities, including upgrading infrastructure and housing along with different arrangements for providing secure land tenure, as well as planned relocation under certain circumstances.

In May 2003, the Prime Minister announced the government's intention to upgrade 100 informal settlements in Phnom Penh annually for five years. Such plans assume that secure tenure in the form of land titles would serve as a foundation for household investments compatible with upgrading community infrastructure. The issuing of individual titles in formal communities, however, poses potential risks, including increased conflicts during surveying and adjudication, encouraging possible land speculation, especially in areas where formal land prices "offer the possibility of large windfall profits", and encouraging more inmigration into "well-located" communities. Other difficulties include the fact that surveying individual plots in informal communities requires considerable time and raises costs, exacerbating human and financial resource capacity constraints (Payne *et al.* 2004).

Given these and other concerns about the constraints and possible negative impacts of individual titles, some planners and researchers have proposed alternative measures for providing secure tenure to individual households in informal settlements. One set of proposals suggests incremental improvements to tenure security, in three main stages. The first stage

would include an announcement of a general moratorium on evictions and relocations for nine months, during which time all settlements could be surveyed to determine which were suitable for upgrading and formal tenure (i.e. individual land titles) as well as identify those for which relocation was required. The second stage would involve medium-term tenure options for different settlement categories, including certificates of communal land rights and community land leases. The third stage would eventually involve long-term tenure in which residents would have the rights to sell, inherit or otherwise transfer their property on the open market (Payne *et al.* 2004).

5A.5. Peri-Urban Areas

Peri-urban areas feature land use elements of both urban and rural areas, including agricultural production mixed with residential, commercial and industrial uses. The 1999 CSES reported a total extent of agricultural land of 12,067 hectares in Phnom Penh, including 10,000 hectares of paddy, of which about half was irrigated. As in rural areas, more secure land tenure in the form of titles could prompt landowners to increase investments in agriculture. Such investments could include variable inputs such as fertilisers and pesticides, as well as labour, machinery or irrigation. Increased investment should in turn stimulate observable increases in yields and in land and labour productivity. This assumes that market conditions, soil quality and climate are more or less constant. In addition to the availability of credit, the degree to which extension services are available to farmers in a particular area is an important factor that may influence investments.

There is, however, evidence suggesting that the amount of land used for agricultural purposes is rapidly decreasing as land is converted to residential and commercial uses. For example, Chan and Acharya (2002: 18), observed that "a sizeable part of Kandal's agricultural land has been converted to factory sites and residential land as urbanisation spills over beyond the Phnom Penh area. In addition, a number of large parcels have been bought up and neatly fenced in anticipation of more factories and commercial ventures to come up in the future". Although this observation applies to areas in Kandal near Phnom Penh, there is also considerable anecdotal evidence to suggest that a similar process has been under way in the peri-urban areas of Phnom Penh for some time. According to official land transaction data, the overwhelming majority of those buying residential and agricultural land in these areas are from Phnom Penh (Sophal and Acharya 2002). Such transactions are likely to increase as more people from the city and foreign-owned firms purchase land for housing, commercial and industrial investment or speculation. Land values in the three peri-urban districts have been steadily increasing over the past several years, and are likely to continue increasing for the foreseeable future, especially in areas where infrastructure, services and utilities are improving.

5B.0. Phnom Penh Baseline Survey

Six of the seven districts in Phnom Penh were included in the baseline survey. Of the four urban districts, Daun Penh, Chamkar Mon and Tuol Kork were included. The district of 7 Makara was originally included in the survey design, but because of difficulties locating property owners in the area it was later decided to do more surveys in other locations. Moreover, it was not clear how the LMAP work schedule would fit in with the baseline survey objective of conducting surveys close to the time when titles would be issued.¹⁸ All three peri-urban districts of Russey Keo, Dangkao and Meanchey were included in the survey. Russey Keo was not included in the survey design, but was added when it was decided not to survey in 7 Makara. Russey Keo was added also because fieldwork in the other two peri-

¹⁸ The experimental areas were chosen on the basis of where LMAP teams were expected to complete their work soon after the surveys were completed, while the control areas were chosen based on where it was expected LMAP would complete its work last. The choice of control areas assumes that land titles will not have been issued prior to the follow-up survey.

urban districts suggested land markets in peri-urban areas were becoming increasingly dynamic and it was considered important to capture this emerging dynamism in the survey sample.

The data can also be analysed at both commune and village level. While there is considerable variation in the number of key indicators across districts as discussed below, there is also considerable variation across communes and villages within districts. For example, land values and uses vary according to commune and village location. Such variations will need to be explored carefully in the course of sorting out the many intervening factors that may cloud or distort the actual impact of land titles. In the interest of space and the reader's time, the following discussion concerning location is limited to district comparisons. A complete list of villages and communes covered in the survey sample is provided in Annex B.

5B.1. Household Characteristics

A total of 1663 households were interviewed in Phnom Penh during the period 21 October–3 December 2005. Of these, 22.7 percent were headed by women, which is practically the same as the overall national average of 22.4 percent.¹⁹ In the three urban survey districts, 857 households were interviewed, including 175 with female heads, about 20.5 percent. In the three peri-urban districts, 806 households were interviewed, including 202 headed by females, about 25.1 percent.

In addition to location and sex of household head, the sample population can also be analysed according to per capita consumption expenditure. Respondents were asked to report household expenditures between Khmer New Year (mid-April) 2005 and the date of the interview on a number of items, primarily those that would entail larger costs (e.g. medical care) and sometimes one-off expenses (e.g. education, social ceremonies). For items such as utilities, respondents were asked to report average monthly expenses. Finally, they were asked to report weekly expenses for items such as food, transportation and entertainment. This information was aggregated to come up with an estimated per capita annual consumption figure. (The survey questionnaire is in Annex A.)

Table 5.3 shows the average household consumption expenditures for the year prior to the survey. Overall, the average per capita consumption expenditure for male-headed households is about 18 percent higher than for female-headed households. There is, however, a remarkable degree of similarity between male- and female-headed households in quintiles 2 and 3, which show almost exactly the same levels of per capita household consumption, while there are only six and two percent differences in quintiles 1 and 4, respectively. In quintile 5, female-headed households' per capita consumption expenditures exceed those of male-headed households by two percent.

Quintile	Male	e HH	Female HH		No. HH	Mean	Range
	No	Mean	No	Mean			
1	242	290	90	274	332	285	80-402
2	247	502	86	504	333	502	402-604
3	257	721	76	721	333	721	605-860
4	262	1067	71	1048	333	1063	861-1310
5	277	2505	55	2561	332	2514	1312-23,576
Total	1285	1053	378	894	1663	1017	80-23,576

 Table 5.3: Consumption Quintiles (USD/year/person)

¹⁹ According to the CSES 2004, National Institute of Statistics.

Table 5.4 shows the average annual per capita consumption in each survey district. As might be expected, the three urban districts have the highest expenditure, Daun Penh having the highest average at USD1796. The three peri-urban districts are all lower, Russey Keo having the lowest average at USD696 per capita per annum.

District	No. HH	Mean Expenditure	Minimum	Maximum
Daun Penh	84	1796	339	10,790
Chamkar Mon	406	1136	170	8335
Tuol Kork	367	1301	177	23,576
Russey Keo	373	696	80	4205
Dangkao	313	822	109	7531
Meanchey	120	704	165	6598
Total	1663	1017	80	23,576

Table 5.4: Annual Per Capita Consumption by District (USD)

Table 5.5 summarises the distribution of the survey households in each district according to household consumption and sex of household head. It is interesting to observe that the percentage of households headed by women steadily declines according to consumption quintile. About 27.1 percent of households in the lowest consumption quintile are headed by women, whereas 16.6 percent of households in the highest quintile are headed by women.

District	Quin	tile 1	Quin	tile 2	Quin	tile 3	Quin	tile 4	Quin	tile 5		Total	
	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Tot
Daun Penh	0	2	5	3	8	5	17	4	33	7	63	21	84
Chamkar Mon	27	11	62	14	62	13	90	17	93	17	334	72	406
Tuol Kork	22	22	51	18	63	18	67	12	81	13	284	83	367
Russey Keo	93	31	76	17	47	15	43	18	26	7	285	88	373
Dangkao	70	15	38	29	44	18	36	18	35	10	223	90	313
Meanchey	30	9	15	5	33	7	9	2	9	1	96	24	120
Total M/F	242	90	247	86	257	76	262	71	277	55	1285	378	
Total HH	33	32	33	33	33	33	33	33	33	32			1663
% of Total	72.9	27.1	74.2	25.8	77.2	22.8	78.7	21.3	83.4	16.6	77.3	22.7	

Table 5.5: Number of Households Surveyed, by District and Consumption Quintile

5B.2. Plot and Unit Characteristics

This section provides some of the key data concerning plot and unit characteristics covered in the survey sample. The data concern the mode of acquisition, the type of documentation to claim or validate ownership, the location of the plot, the services available to it and plot values. Two different data sets are used. The most relevant data set concerns the plot or unit where the owner resides or works and where the interview took place. This is referred to as Plot/Unit 1 throughout the following discussion and is the primary unit of analysis for land titling impacts. The total of "Plot/Unit 1" is 1663, equal to the number of household interviews.

The second data set that is occasionally discussed concerns all the plots that are owned by the respondents. These data are compared to the Plot/Unit 1 data as appropriate to highlight certain points (e.g., mode of acquisition, mode of documentation) and to serve as a check on the Plot/Unit 1 data. The total number of "all plots" represented in this data set is 2834, or 1.70 plots per household. This is higher that the average of 1.36 plots per household found in the 2004 Sihanoukville survey in Sangkat 2 in Mittapheap. Both urban area averages,

however, are well below the average of 4.23 agricultural plots per household found in the rural LMAP titling areas, as one would expect given rural land use patterns and factors such as population density. About 92.5 percent of "all plots" are located in Phnom Penh municipality and 2.2 percent "near Phnom Penh", while the remaining 5.3 percent are located elsewhere.

5B.2.1. Mode of Acquisition

Table 5.6 shows the mode of acquisition for Plot/Unit 1 according to location. It is interesting that the predominant mode of acquisition for all of Phnom Penh municipality is by purchase (50.5 percent), followed by state allocation (35.4 percent) and inheritance or family donation (12.9 percent). This pattern is significantly different from that found in the rural baseline survey, where the predominant modes of acquisition were state allocation (55.6 percent) and inheritance (21.7 percent), followed by purchase (15.2 percent) and clearing (7.4 percent). This pattern underscores the increasingly prominent role that transactions are playing in redistributing land throughout the municipality.

Table 5.6 also shows clearly that the predominant mode of acquisition for Plot/Unit 1 in the three urban districts is by purchase (71.5 percent of all plots), while the predominant mode of acquisition in the three peri-urban districts is through allocation by the state (49.6 percent) followed by purchase (28 percent) and inheritance (20.2 percent). Although one would expect the peri-urban areas to approximate the rural areas more closely in terms of a variety of indicators, the mode of acquisition suggests that the urban tendency toward land acquisition by purchase is making inroads on the outskirts of the city. The data discussed below in Section 5.3 concerning land transactions suggest that this trend has been accelerating over the past several years.

District	Given b	by State	Inhe	rited	Purcl	hased	Ot	her	Total
	Ν	%	Ν	%	Ν	%	Ν	%	
Daun Penh	18	21.4	0		66	78.6	0		84
Chamkar Mon	70	17.2	24	5.9	309	76.1	3	.6	406
Tuol Kork	100	27.2	27	7.4	238	64.9	2	.6	367
Russey Keo	192	51.5	76	20.4	96	25.7	9	2.4	373
Dangkao	171	54.6	64	20.4	71	22.7	7	2.3	313
Meanchey	37	30.8	23	19.2	59	49.2	1	.8	120
Total	588	35.4	214	12.9	839	50.5	22	1.4	1,663

 Table 5.6: Mode of Acquisition by District (Plot/Unit No. 1)

Table 5.7 shows that of the 839 plots/units acquired through purchase, 229 (27.3 percent) were purchased between 2000 and 2005. About 62 percent (142) of these purchased properties are in the three urban districts, where the average plot/unit values are considerably higher than in the three peri-urban districts. Over half (57.2 percent) of the purchases between 2000 and 2005 took place during the second half of that period.

 Table 5.7: Plot Purchases, 2000–05, by Quintile (USD per Plot/Unit No. 1)

Location	2000	2001	2002	2003	2004	2005	Average
Daun Penh	37,333	18,000	60,000	117,625	141,143	112,500	100,786
Chamkar Mon	41,200	18,475	20,614	25,318	25,479	25,667	25,231
Tuol Kork	21,273	40,050	31,750	26,612	27,324	25,400	27,820
Russey Keo	5317	1290	1660	3463	5500	8125	3635
Dangkao	2267	875	3225	2,594	16,550	48,000	6247
Meanchey	12,167	4,000	20,000	7890	3517		8292
Average	16,933	16,379	17,017	24,778	35,255	41,332	25,471
No Unit/Plots	40	32	26	52	60	19	229

Table 5.8 shows that the predominant modes of land acquisition for all 2834 plots (henceforth "all plots") covered in the Phnom Penh survey are purchase (52.3 percent), state allocation (34.6 percent), and inheritance (12 percent). Perhaps not surprisingly, the percentage of plots acquired through purchase steadily increases according to consumption levels, especially for male-headed households, while the percentage acquired from state allocation and through inheritance varies across consumption levels, especially for female-headed households. Overall, 54.4 percent of male-headed households and 44.7 percent of female-headed households purchased the plots they own. Almost an identical percentage, 31.8 percent, of male-headed households received the land they currently own from the state. This may reflect of the fact that male-headed households have generally higher levels of consumption, which suggests they have more resources with which to purchase land. A similar percentage of male- and female-headed households, 12.5 and 10.2 percent, respectively, acquired land through inheritance.

The distribution of acquisition modes of all plots according to location also closely approximates the pattern for Plot/Unit 1. For example, in the three urban districts, 73.1 percent of all plots have been acquired through purchase, while 20.1 percent were acquired from the state. In the peri-urban areas 46.1 percent of the plots were acquired from the state, while 35.7 percent were acquired through purchase. The distribution of plot/unit ownership according to the sex of household head closely approximates the survey sample, as 22.7 percent of the survey households were headed by women, while 22.8 percent of all plots covered in the survey are owned by female-headed households.

Quintile	St	ate	Inh	erit	Purc	hase	Cl	ear	Ot	her	То	tal
	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F		
1	146	59	73	13	125	51	3	1	4	0	47	75
2	147	69	55	15	204	51	4	0	3	0	54	18
3	137	56	62	10	235	55	2	1	2	0	56	50
4	142	54	52	17	273	58	1	2	4	0	60)3
5	134	36	35	8	368	61	5	0	1	0	64	18
											Μ	F
Total M/F	706	274	277	63	1,205	276	15	4	14	0	2217	617
% of M/F	31.8	34.3	12.5	10.2	54.4	44.7	.7	.6			78.2	21.8
Total	98	30	34	40	14	81	1	9	1	4	28	34
% of Total	34	.6	12	2.0	52	3	0	.6	0	.5		

 Table 5.8: Mode of Acquisition, by Consumption Quintile (All Plots/Units)

Given the acceleration of land transactions throughout the municipality (Section 5.3 below), the mode of acquisition will certainly shift in the direction of purchase, followed by inheritance. As a result, the share of land acquired through state allocation is likely to decline, perhaps sharply, unless some social land concession mechanism is established within the municipality (e.g., resettling residents evicted from informal settlements). Such shifts will occur in all three sectors—urban, peri-urban, and rural—but will not be the direct result of land titles *per se*. Particularly in the peri-urban and urban areas, they will be the result of expanding land markets for both speculative and productive investment. Land-titling, however, will surely facilitate the process by reducing the transaction costs of purchase and sale, thus making such exchanges more efficient. The degree to which transactions will be recorded with the official registry is another matter, and is discussed below (Section 5B.3.3).

5B.2.2. Documentation

Table 5.9 shows that urban and peri-urban households rely on a variety of documentation modes. As in the rural areas, it appears that urban households have relied on unofficial documentation. For Plot/Unit 1, this includes survey papers (50.2 percent) and receipts for certificate applications (22.8 percent), making a total of 73 percent of plots officially

recognised to a certain extent, but not legally documented. Only 5.4 percent of households reported having an official certificate, which is far below the 25 percent estimate cited in Section 5.0 above.

District	App.	Survey	Certificate	Sales	ComVil	No	Total
	Receipt	Paper		Letter	Transfer	Paper	
Daun Penh	2	57	21	2	2	0	84
Chamkar Mon	47	122	36	29	108	61	403
Tuol Kork	11	312	16	8	16	4	367
Russey Keo	138	188	4	1	33	8	372
Dangkao	106	153	7	2	23	21	312
Meanchey	74	1	5	4	29	7	120
Total N	378	833	89	46	211	101	1658 *
% of Total	22.8	50.2	5.4	2.8	12.7	6.1	

 Table 5.9: Land Documentation, by Location (Plot/Unit 1)

* 5 missing.

App. Receipt = *Application Receipt; ComVil* = *Commune or Village level*

Although the number of plots that have a certificate is surprisingly low compared to other studies, the average value of plots documented with certificates is significantly higher than plots documented with other types of paper. The average value of plots documented with certificates was USD100,704, while the average of land documented with survey papers, the most frequently used mode of documentation in the sample, was USD41,157. The average value of land documented with receipts for certificate applications, the second most frequent form of documentation, was USD30,451. Village and commune letters of ownership transfer averaged USD22,336 and USD37,398, respectively. Surprisingly, the average value of plots with "no documents available in the house" was USD7708 and the average of plots with lost documentation was USD40,396. Under certain circumstances, undocumented plots of this value would appear to expose owners to conflicts with others.

Table 5.10 shows a somewhat similar pattern for all plots covered in the survey. About 41.6 percent of all plots are documented with survey papers, while another 26.4 percent are documented with receipts for land certificate applications, making a total of 67.5 percent of plots documented officially but not legally. Only 6.1 percent of the plots were documented with a land certificate. Documentation by commune or village transfer agreements, however, is higher for all plots than in the Plot/Unit 1 sample. This may be due in part to the fact that more plots are located outside one's village or commune in the wider sample, and people may feel more secure about having their more distant plots secured with seemingly stronger documentation.

Once LMAP completes its systematic titling, the structure of land tenure documentation will shift toward a much higher share of documentation using land titles, in principle 100 percent. This is one of the main objectives of the LMAP land-titling. In this regard, there are two important points to consider. The first is the degree to which land transfers, most importantly transactions and inheritances, will be facilitated through the official registry. There is already anecdotal evidence that many such transfers are being done the usual way via "soft" documents (e.g., certificate application receipts). Another concern is the enforceability of the LMAP titles. The degree to which such titles are upheld by state authorities and courts in cases of conflict over ownership will be an important test of tenure security. As discussed in Section 3.6 above, the question of enforcement is of special interest to the tenure rights of women, particularly in cases of divorce or the death of a husband.

District	App.	Survey	Certificate	Sales	ComVil	No	Total
	Receipt	Paper		Letter	Transfer	Paper	
Daun Penh	18	66	28	4	18	6	140
Chamkar Mon	87	138	64	38	176	74	577
Tuol Kork	48	348	38	13	84	8	539
Russey Keo	287	356	23	10	68	19	763
Dangkao	212	267	12	4	114	42	651
Meanchey	93	2	6	5	41	9	156
Total	745	1177	171	74	501	158	2826*
% of Total	26.4	41.6	6.1	2.6	17.7	5.9	

 Table 5.10: Land Documentation, by Location (All plots/units)

* 8 missing

App. Receipt = Application Receipt; ComVil = Commune or Village

5B.2.3. Location and Services

Table 5.11 shows the access to Plot/Unit 1 in each district. Some 61.2 percent of the survey plots are located along main paved roads or 2-way access streets, paved or otherwise. Special efforts were made to survey households with more difficult access, and this is reflected in the number of surveyed plots with one-way access (22.4 percent) or bicycle and foot access (16 percent). Although there is not a general survey of plot locations for Phnom Penh in terms of access, the sample is probably biased in favour of property located along main paved roads and 2-way access streets.

District	Main Paved	2-way	1-way	Bike	Foot	Total
	Road	Access *	Access*	Access	Access	
Daun Penh	69	4	4	7	0	84
Chamkar Mon	166	128	46	66	0	406
Tuol Kork	66	213	81	6	1	367
Russey Keo	38	172	90	69	4	373
Dangkao	15	100	131	66	1	313
Meanchey	28	25	21	46	0	120
Total	382	642	373	260	6	1663
% of Total	25.4	38.6	22.4	15.6	0.03	

Table 5.11: Access to Plot/Unit 1, by District

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

Households were asked about the utilities and services they receive, including electricity, phone (land line), sewerage and water for cooking. Table 5.12 shows that 61.7 percent of the households in the sample have electricity from the state service. Perhaps not surprisingly, the households in the three urban districts tend to have a higher rate of electricity service than do the peri-urban areas, except for Meanchey, where 84 percent of survey households have state electricity, despite the fact that fewer than half (44.2 percent) of the households are located along main paved roads and 2-way access streets.

District	Main Paved	2-way	1-way	Bike	Foot	Total	% of HH
	Road	Access*	Access*	Access	Access		
Daun Penh	69	4	4	7		84	100
Chamkar Mon	166	120	43	49		378	93.1
Tuol Kork	66	213	80	6	1	366	99.7
Russey Keo	2	7	17	12		38	10.2
Dangkao	10	12	18	19		59	18.8
Meanchey	28	11	18	44		101	84.2
Total	341	367	180	137	1	1026	61.7
% of Total	33.2	35.8	17.5	13.4			

 Table 5.12: Electricity Services (No. of Households Having Electricity)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

Fewer households (53.2 percent) have municipal sewage service. As with electricity, one would expect the three urban districts to have a higher percentage of service than the periurban areas, and Table 5.13 shows that this is indeed the case. The survey plots in Daun Penh and Chamkar Mon have almost identical rates of electrical and sewage services, while the survey plots in Russey Keo and Dangkao both have low rates for both services. Even though the survey plots in Meanchey have a high coverage rate for electricity, they have a very low rate of coverage for sewage (4.2 percent).

District	Main Paved	2-way	1-way	Bike Access	Foot	То	tal
	Road	Access*	Access*		Access		
							%
Daun Penh	69	4	4	7		84	100
Chamkar Mon	166	124	42	48		380	93.6
Tuol Kork	52	198	69	1	1	321	87.5
Russey Keo	1	11	6	1		19	5.1
Dangkao	14	20	15	27		76	24.3
Meanchey	2	1	0	2		5	4.2
Total	304	358	136	86	1	885	53.2
% of HH	34.4	40.4	15.4	9.7	.1		

 Table 5.13: Sewage Service (No. of Households Having Service)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

5B.2.4. Land Use and Value

In this section, some of the key data on plot/unit type and land uses and land values are presented. Because of the expected impact of land titles on housing and property improvements, some information on buildings and housing quality is also presented.

5B.2.4.1. Plot/Unit Types and Uses

Table 5.14 summarises Plots/Units 1 according to consumption quintile and use. At least 88.6 percent of the plots were used only as residences, or as residences along with other uses. The percentage of residential only use declines as consumption increases, while residential/commercial use generally increases along with consumption. The same is also true for mixed uses. One assumes that households with higher consumption levels may have more diversified income sources, as evidenced by a larger share of combined residential and commercial uses as well as mixed uses. In contrast, a greater share of the lower consumption quintile households may have to earn income away from the home.

Quintile	Residential	Commercial	Res/Com	Res/Rent	Res/Ag	Mixed	Total
					-		Number
1	71.7	.6	14.5	7.2	1.5	4.5	332
2	57.1	1.2	19.2	5.4	5.1	12.0	333
3	58.9	.9	23.4	2.7	5.7	8.4	333
4	54.1	.3	25.2	2.4	5.1	12.9	333
5	53.6	.6	23.8	1.5	5.4	15.1	332
% of Total	59.0	.7	21.2	3.8	4.6	10.6	1663

 Table 5.14: Plot/Unit 1 Use (% of Households)

Land titles are likely to have an indirect impact on land use patterns to the extent that they facilitate more efficient transactions and increase returns on investments. It is possible that in the three urban districts a shift will occur as residential property is converted to commercial and/or rental property, while in the peri-urban areas changes in land use may involve a shift from residential and agricultural uses to commercial or mixed uses. The nature and direction of land use changes in and around Phnom Penh will be highly dependent on the municipality's long-term development strategy. A key component of such a strategy will be clear and enforceable zoning, as discussed in Section 5.1. The data presented in Table 5.15 provide a good basis for assessing these hypotheses in a follow-up study.

District Residential Commercial Res/Com Res/Rent Res/Ag Mixed Total Daun Penh Chamkar Mon Tuol Kork Russey Keo Dangkao Meanchey Total 21.2 3.8 10.6 % of Total 59.0 .7 4.6

Table 5.15: Plot Use, by District (No. of Households)

5B.2.4.2. Land Values²⁰

Respondents were asked how much they would receive if they were to sell their plot (Plot/Unit 1) at the time of the interview. In some cases, they may have overestimated the value of their plot, because bank officials and real estate agents have observed that loan applicants or potential sellers tend to overstate the value of their land when it is in their interest to do so. Other respondents may have underestimated the value of their land, perhaps not completely trusting the purpose and intentions of the interviewers. In any event, the data concerning reported plot values closely conform to expected trends. First, Table 5.16 shows that average plot values in the three urban districts are significantly higher than the averages for the three peri-urban districts. In the case of Daun Penh and Russey Keo, the urban plots are valued at five times the peri-urban plots. In the other districts, the urban plot values are slightly more than double the peri-urban plot values.

Second, as expected, plot values increase along with better access, in part because access to services such as electricity and sewerage is generally easier. Although main paved roads and

²⁰ For the purposes of this report, land values are presented in terms of total USD per plot as reported by respondents, as we are interested in the total value of the plots and the total amounts involved in transactions (i.e., sales, purchases).

2-way access streets both represent two-way access, the reported average value of plots situated along main paved roads is almost double the average reported value of plots situated along either paved or unpaved streets. This is consistent across four of the six districts, and in the other two districts the plots along main paved roads are valued substantially higher than those situated along two-way streets.

District	Main Paved	2-way	1-way	Bicycle	Foot	,	Total
	Road	Access *	Access *		Access		
						No	Ave \$
Daun Penh	125,917	39,500	45,000	18,200		73	109,370
Chamkar Mon	66,163	59,664	32,023	18,131		390	52,715
Tuol Kork	92,226	53,803	19,590	11,250	5000	348	52,106
Russey Keo	54,348	18,837	19,225	14,942	4800	372	21,684
Dangkao	70,867	30,553	19,709	16,261	7000	312	24,878
Meanchey	37,500	27,154	18,024	14,196		120	23,003
Average Value	77,238	40,518	21,226	15,901	5200	4	-0,412
Total HH	363	627	366	253	6	10	615 **

Table 5.16: Land Values (USD/plot)

* For a vehicle (Please see Survey Questionnaire, Question 2.1. ** 48 no response

The data also conform closely to expectations when average plot values are analysed according to consumption quintiles. As Table 5.17 shows, the value of the plot steadily increases with the consumption quintile. The average values between the first and fifth quintile range from USD20,108 to USD70,589, while the range between the middle three quintiles is of course somewhat less. This general pattern holds for all districts, except for Chamkar Mon and Dangkao, where the average value dips between quintiles 2 and 3, and quintiles 3 and 4, respectively.

District	Quint 1	Quint 2	Quint 3	Quint 4	Quint 5		Total
						HH	Ave \$
Daun Penh	32,500	82,000	117,500	65,176	139,559	73	109,370
Chamkar Mon	24,378	45,719	41,268	57,820	70,132	390	52,715
Tuol Kork	32,317	39,919	48,404	49,506	74,950	348	52,106
Russey Keo	18,388	18,450	21,260	25,233	37,322	372	21,684
Dangkao	14,163	24,427	29,016	21,952	43,692	312	24,878
Meanchey	21,009	18,000	25,650	20,909	32,500	120	23,003
Average Value	20,108	31,561	37,746	42,818	70,589	4	0,412
Total HH	328	322	324	323	318	1	615 *

 Table 5.17: Reported Plot/Unit Values (USD), by District and Consumption Quintile

*N = 48 no response

5B.2.5. Buildings and Housing

Data on the number, use and quality of buildings on the survey plot were also collected. This information will be important for assessing potential land-titling impacts according to the hypothesis that people will respond to more secure tenure by investing in housing and building improvements (which may include more buildings), and may use titles as collateral for credit for making improvements. Table 5.18 shows that many plots, about 21.5 percent, have more than one building. In this section, only a small sample of the available data for the primary building is presented.

Quintile	1	2	3	4	5	6	Total
1	269	48	12	1	0	1	331
2	255	67	8	2	1	0	333
3	252	55	19	3	2	2	333
4	264	49	15	5	0	0	333
5	263	44	11	7	3	3	330
Total	1303	263	65	18	6	5	1660*

Table 5.18: No. of Buildings per Plot/Unit 1, by Consumption Quintile

* 3 missing

With regard to building quality, information was collected on the materials used for roofing, walls and floors. Since roofing is generally considered a good indicator of household consumption level and well-being, one would expect to find a higher proportion of low-cost roofing among the lower quintiles. Over time, the overall quality of roofing would be expected to change, houses at each consumption level upgrading their roofing material, although upper quintile households may remain static because they are already using better materials. The data on roofing material, presented in Table 5.19, suggest that this assumption may be misleading. Although the quality of roofing generally does improve with the quintile level, there is a higher than expected level of zinc roofing in the upper two quintiles. Thus roofing may not be a useful indicator of household consumption levels after all, at least in urban areas, and as a result may not be a useful indicator of land-titling impacts.

Quintile	Thatch/Tent	Zinc sheets	Tile	Wood/plywood	Concrete	Total
1	1	233	73	3	21	331
2	1	196	94	5	37	333
3	1	187	83	5	57	333
4	0	188	88	8	48	332
5	0	144	106	14	66	330
Total	3	948	444	35	229	1659*

Table 5.19: Housing Quality, Consumption Quintile by Roofing Type

* 4 missing

5B.3. Land Transactions

This section presents some of the key data on land purchases and sales and the documentation used to facilitate and validate such transactions. Given the rapidly expanding land markets throughout the municipality and the role that land titles are expected to play in facilitating more efficient land transfers through the official registry, this section is especially germane to assessing the impact of land-titling in urban and peri-urban areas.

5B.3.1. Purchases

As observed above, 1470 of all 2834 plots (52.3 percent) covered in the survey were acquired through purchase. As shown in Table 5.20, of these plots, 631 (42.9 percent) were purchased during 2000–05. During this period the number of plot purchases increased steadily with consumption quintile. For example, households in quintiles 1 and 2 purchased 11.7 and 17.1 percent of the plots, respectively, while households in quintiles 4 and 5 purchased 22.3 and 29.8 percent. Comparing the first three years of this period to the second three years, it appears that the rate of land purchases is accelerating across all consumption quintiles. Land accumulation through purchase appears to be accelerating faster in the upper two quintiles than in the lower two quintiles.

Quintile	20	000	20	001	20	002	20	003	20	04	20	05	To	otal
	No	%	No	%	No	%	No	%	No	%	No	%	No	%
1	7	9.5	11	14.9	10	1.35	16	21.6	14	18.9	16	21.6	74	11.7
2	17	15.7	11	10.2	14	13.0	22	20.4	28	25.9	16	14.8	108	17.1
3	16	13.3	10	8.3	20	16.6	23	19.2	29	24.2	22	18.3	120	19.0
4	15	10.6	12	8.5	17	12.1	24	17.0	47	33.3	26	18.4	141	22.3
5	29	15.4	11	5.9	19	10.1	32	17.0	49	26.1	48	25.5	188	29.8
Total	84	13.3	55	8.7	80	12.7	117	18.5	167	26.5	128	20.3	631	

Table 5.20: Plot Purchases 2000–05, by Quintile (All Plots)

As for the location of all plot purchases during 2000–05, Table 5.21 shows that 52.9 percent took place in the peri-urban districts. Daun Penh and Meanchey have the lowest percentage of purchased plots. For Daun Penh, this is probably largely due to the high value of plots, while in Meanchey, where the average value of land is much lower, it could reflect a low level of services and poor access.

District	Quin	tile 1	Quin	tile 2	Quin	tile 3	Quin	tile 4	Quin	tile 5	Тс	otal
	No.	%	No.	%								
Daun Penh			1	.2	5	.8	9	1.4	24	3.8	39	6.2
Chamkar Mon	3	.3	21	3.3	10	1.6	28	4.4	51	8.1	113	17.9
Tuol Kork	9	1.4	21	3.3	25	4.0	38	6.0	52	8.2	145	23.0
Russey Keo	33	5.2	41	6.5	26	4.1	38	6.0	14	2.2	152	24.1
Dangkao	21	3.3	20	3.2	37	5.9	24	3.8	44	7.0	146	23.1
Meanchey	8	1.3	4	.6	17	2.7	4	.6	3	.5	36	5.7
Total	74	11.7	108	17.1	120	19.0	141	22.3	188	29.8	631	100

 Table 5.21: Land Purchases (2000–05), by District and Consumption Quintile (All Plots)

Table 5.22 shows that of the 631 all plot purchases in 2000–05, a total of 412 (65.3 percent) occurred in the last three years. Of this number, 221 (53.6 percent) were purchased by households in the three peri-urban districts, mostly in Russey Keo and Dangkao. The remaining 46.4 percent of purchases during the three-year period 2003-05 were purchased by households in the two urban districts of Chamkar Mon and Tuol Kork. The plot purchases in Daun Penh all occurred during 2000-02. Somewhat surprisingly, there were fewer property purchases reported in 2005 than in 2003 and 2004, except in Dangkao, where there was a 36.8 percent increase.

	2000	2001	2002	2003	2004	2005	Total
Daun Penh	8	3	2	10	9	7	39
Chamkar	15	14	12	25	33	14	113
Mon							
Tuol Kork	19	14	19	28	41	24	145
Russey Keo	16	18	27	29	37	25	152
Dangkao	19	3	17	17	38	52	146
Meanchey	7	3	3	8	9	6	36
Total	84	55	80	117	167	128	631
% of Total	13.3	8.7	12.7	18.5	26.5	20.3	100

Table 5.22: Land Purchases (2000–05), by District (All Plots)

Of the 1490 reported land purchases, a total of 278 (18.7 percent) took place during 1980–89, including as many as 45 in both 1985 and 1989. Although the data for this period almost certainly contain some reporting errors, the surprisingly large number suggests there was a nascent land market in the Phnom Penh area even before 1989, when the government reintroduced private property rights through Instruction Number 3, along with Sub-decree 25. Among other provisions, the sub-decree established ownership rights for residential land of up to 2000 sq. metres and possession rights for cultivated land of up to five hectares. The 1992 Land Law then provided for land tenure certificates that confirmed occupancy and use rights, although allowing only possession rights rather than ownership in rural areas.

Table 5.23 shows the reported average plot purchase price according to consumption quintile and district. Overall, the purchase price per plot increases with the consumption quintile. This conforms to the expected pattern that those better off would tend to pay more per plot. This is also consistent across all locations, except for quintile 2 in Meanchey and Quintile 4 in Daun Penh. The pattern according to location also conforms to expectations, as the purchase price per plot in the three urban districts is decidedly higher than the price in the peri-urban districts. This is consistent across all quintiles.

Table 5.23: Average Price of Land Purchased (2000–05), by District and Consumption Quintile (USD/Plot)

District	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Total
Daun Penh		40,000	52,800	25,388	98,308	75,434
Chamkar		19,638	11,130	23,611	22,997	21,198
Mon	8250					
Tuol Kork	11,378	18,424	18,908	18,652	33,216	23,157
Russey Keo	1907	2814	2871	7657	7350	4284
Dangkao	4743	4423	6476	5152	9446	6643
Meanchey	2503	10250	6300	5988	7333	5946
Total	4103	10,106	10,709	14,368	30,916	16,669

It is interesting to compare Table 5.23 with Table 5.24, which shows the estimated current value of each plot at the time of the interview. The difference between the reported purchase price and the current (November 2005) estimated value is quite remarkable. Across each consumption quintile and district, the current estimated plot value is considerably higher than the reported purchase price. While this data could reflect understated purchase prices and/or inflated estimates of value, the data provide a clear example of how people in Phnom Penh perceive the exploding land values in the city. Of particular interest is the current estimated value for land in peri-urban areas, which are well over double the reported purchase price.

Table 5.24: Estimated Value of Land Purchased (2000–05), by District and Quintile (USD/Plot)

District	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Total
Daun Penh		42,000	160,000	32,750	129,056	105,387
Chamkar	16,700	34,000	19,889	35,138	33,336	32,310
Mon						
Tuol Kork	15,125	33,912	29,667	29,039	74,213	44,784
Russey Keo	10,424	9,457	10,320	22,229	24,314	14,403
Dangkao	10,255	12,219	20,024	13,929	38,981	22,383
Meanchey	7375	21,125	14,397	10,250	15,667	13,229
Total	10,830	19,636	23,919	25,356	54,522	30,875

5B.3.2. Land Sales

Almost exactly half (831) of the survey households reported selling a total of 1017 plots of land since 1980, an average of 1.22 per selling household. The percentage selling land since 1980 was remarkably consistent across all consumption quintiles. Just over half of all sales took place since 1997, while one-third (33.6 percent) occurred since 2001. About 21.8 percent occurred in 2004–05. As with the purchase data, the land sales data suggest that the pace of land transactions is accelerating in and around Phnom Penh.

About 53.3 percent (443) of the plot sales took place during 2000–05. The distribution of sales according to district and consumption quintile is shown in Table 5.25. Seventy-two percent of all reported land sales during this period took place in the peri-urban districts, mostly in Russey Keo and Dangkao, while the remaining 28 percent occurred in the three urban districts, mostly in Chamkar Mon and Tuol Kork. Meanwhile, 49.4 percent of all plots were sold by households in quintiles 4 and 5, while 32.7 percent of plots were sold by quintiles 1 and 2.

District	Quin	tile 1	Quint	tile 2	Quin	tile 3	Quin	tile 4	Quin	tile 5	To	otal
	No.	%	No.	%	No.	%	No.	%	No.	%	No.	%
Daun Penh	0		1	.2	1	.2	2	.5	9	2.0	13	2.9
Chamkar	2	.5	5	1.1	8	1.8	13	2.9	11	2.5	39	8.8
Mon												
Tuol Kork	7	1.6	11	2.5	10	2.3	18	4.1	26	5.9	72	16.3
Russey Keo	32	7.2	32	7.2	19	4.3	39	8.8	23	5.2	145	32.7
Dangkao	24	5.4	23	5.2	31	7.0	31	7.0	39	8.8	148	33.4
Meanchey	6	1.4	2	.5	10	2.3	6	1.4	2	.5	26	5.9
Total	71	16.0	74	16.7	79	17.8	109	24.6	110	24.8	443	100

Table 5.25: Land Sales (2000–05), by District and Quintile

Table 5.26 shows that the reported value of plots sold also varied according to location, the three urban districts showing a significantly higher value per plot than the three peri-urban districts. The value of plots sold also generally increases across consumption quintiles, particularly from quintiles 3 through 5. The reported average value of plots purchased and plots sold varies with the year of transaction and the location.

Table 5.26: Land Sales (2000–05), by District and Consumption Quintile (USD/Plot)

District	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Average
Daun Penh		8400	50,000	70,000	78,250	68,700
Chamkar Mon	12,500	20,000	7438	16,885	40,214	21,701
Tuol Kork	8071	22,227	19,000	40,761	45,288	33,364
Russey Keo	4961	4740	7212	9490	9585	7159
Dangkao	4326	6904	8956	9522	16,692	10,043
Meanchey	5100	4500	12,070	13,167	21,500	10,858
Average	5277	9086	10,568	16,858	34,094	16,862
Total No.	71	74	79	109	110	443

5B.3.3. Transaction Documentation

Table 5.27 shows that a significant proportion of land sales (92.6 percent) were facilitated by changing names on the application receipts. A majority (59.4 percent) of these sales were

documented at the commune level, while 19.7 and 13.3 percent were documented at the village and district level, respectively. The high percentage of commune documentation across all quintiles may reflect the fact that the commune is viewed as an important locus of local governmental authority, at least for land transactions. It is also interesting, however, that the share of sales documented at the village level steadily decreases according to quintile, while steadily increasing at the district level. This suggests that transaction costs associated with documentation may be an important factor in determining how people document land transactions, with the lower quintile groups documenting more frequently locally, while a greater percentage of upper quintile households document with the district.

Quintile	Village	Commune	District	Province	Title/MLUPC	Contract	Total No.
1	33.1	48.9	8.4	1.1	0	8.4	178
2	24.1	64.6	6.2	1.0	0	4.1	195
3	24.6	53.8	15.9	2.1	0	3.6	195
4	12.0	67.0	13.4	1.9	.1	5.3	209
5	8.5	60.9	20.9	5.1	.2	3.8	235
Total No.	199	601	135	24	3	50	1012*
% of total	19.7	59.4	13.3	2.4	0.3	4.9	

 Table 5.27: Mode of Documentation of Land Sales, by Quintile (percent)

* 5 missing

This observation suggests that if, as expected, land sales increase among the survey population after titles are issued, people may still continue to document transactions locally in ways that are not really legal. There is anecdotal evidence of this occurring already in areas where titles have been issued (e.g., Markussen, 2007) as well as documented evidence in areas such as Prey Nob in Sihanoukville (ADI, 2007). This is a matter of fundamental importance because one of the key reasons for undertaking systematic land titling was to facilitate land transfers through the official registry. Transactions that take place outside the official registry will contribute to continued conflicts over land and represent a major source of lost revenue for the government.

5.4. Conflicts

Households were asked, "Have you ever had conflicts over land or other property since the national election of 2003"? Somewhat surprisingly, only 36 instances of conflict were reported, about 2.2 percent of the Phnom Penh survey households, during the period in question July 2003–December 2005). This represents 1.5 percent of all the plots covered in the survey sample, which is only slightly higher than the 1.3 percent of conflicted plots in the rural survey sample. In an increasingly active land market in which values are rapidly climbing, one would expect more conflicts in and around Phnom Penh. There are, however, several reasons why this figure may not be too far off within the survey population. One is the fact that the LMAP has tended initially to target titling efforts where there is less conflict over land. Moreover, in many of the areas of the municipality, particularly where boundaries have been secured by fencing, many land conflicts may well have been resolved prior to 2003. That being said, it is also entirely possible that land-related conflicts have been under-reported within the survey population.

Within the 36 cases, there is no discernible pattern over time. Almost two-thirds of the cases involved residential land, while about one-third involved agricultural land. More than 60 percent of the households reported seeking resolution from either the village chief or the commune leader, while 10 percent of the cases were resolved with the help of neighbours or through agreement. This suggests, albeit not with any degree of certainty given the small sample, a preference for managing conflicts locally. However, fewer than half of the cases

(17) were reported to have been resolved, and only one-third of the households reported satisfaction with the result.

This limited number of cases of conflict will probably not be useful for measuring the extent, if any, to which land titles will have helped reduce land conflicts in a follow-up impact study. In addition to asking people about land conflicts experienced since receiving titles, interviewers will also need to ask households more specifically about their perceptions concerning the role that titles may or may not have played in reducing conflicts. It will also be important to inquire about the role that land titles have played in resolving conflicts. For example, in the event that conflicts arise over land that has been titled, how have the conflicts been resolved? This suggests that semi-structured questions must be added to the survey in order to allow for more probing follow-up. Most important are questions concerning the role the courts and local authorities play in enforcing land titles in cases of conflict.

5B.5. Credit

Of the 1663 households covered in the survey, 359 reported obtaining a total of 501 loans during the period since the national election of 2003. Although the number of loans per borrowing household, 1.40, is similar to the ratio in the rural survey, the percentage of surveyed households reporting credit activity, about 21.6 percent, is much lower than the 54.7 found in the rural survey. It is also interesting that the urban survey covered a period of about two and a half years, while the rural survey covered only the six months prior to the survey (in early 2004). By comparison, the small sample in Sihanoukville found that 31 households out of 99, or 31.3 percent, had obtained 41 loans in cash or gold, a ratio of about 1.32 loans per borrowing household. The Sihanoukville survey also covered a six-month period prior to the interviews in early 2004.

There are several possible reasons for this pattern. One is that the number of loans may be under-reported, just as people may under-report their income or consumption. However, one might expect that as households are generally better off in urban areas than in rural areas, they may have less need to borrow. In this regard, the pattern in loan activity across the different consumption levels is interesting. Table 5.28 shows that the percentage of households in each quintile that report any credit activity steadily decreases as consumption increases. For example, quintile 1 households accounted for 26.9 percent of the borrowing, while quintile 5 households accounted for only 14.0 percent.

According to the research hypothesis, land titles are expected to stimulate increased borrowing from formal sources. It is likely that the trends discussed above will accelerate the actual numbers of loans, with lower quintile households continuing to borrow at an even more rapid rate than upper quintile households. The upper quintile groups, however, are likely to borrow larger amounts.

Source	Quint	tile 1	Quin	tile 2	Quin	tile 3	Quin	tile 4	Quin	tile 5	Т	`otal
	No.	%	No.	%	No.	%	No.	%	No.	%	No.	%
Relative	28	20.7	31	24.8	28	30.8	36	45.0	25	35.7	148	29.5
Friend	3	2.2	13	10.4	3	3.3	8	10.0	5	7.1	32	6.4
Moneylender	28	20.7	20	16.0	15	16.5	16	20.0	17	24.3	96	19.2
NGO	6	4.4	8	6.4	9	9.9	0		0		23	4.6
MFI	8	5.9	9	7.2	2	2.2	3	3.8	5	7.1	27	5.4
Acleda	62	45.9	44	35.2	34	37.4	17	21.3	13	18.6	170	33.9
Other	0		0		0		0		5	1.9	5	1.0
Total	135	26.9	125	25.0	91	18.1	80	16.0	70	14.0	501	100

Table 5.28: Credit Sources, by Consumption Quintile

Table 5.29 also shows that credit activity, as measured in the number of loans, increased each year in each consumption category during 2003–05. Of particular interest is the fact that just over half (50.8 percent) of the loans reported during this period occurred in 2005.

Quintile	200)3	20	04	20	05	Тс	otal
	No.	%	No.	%	No.	%	Loans	HH
1	30	22.2	29	21.5	76	56.3	135	332
2	19	15.2	40	32.0	66	52.8	125	333
3	21	23.3	21	23.3	47	52.2	89	333
4	15	18.8	31	38.8	33	41.3	79	333
5	18	25.7	21	30.0	31	44.3	70	332
Total	103	20.7	142	28.5	253	50.8	498*	1663

Table 5.29: Credit Activity (2003–05), by Quintile

* 3 missing

Table 5.28 also shows the distribution of loans according to source. In terms of potential landtitling impacts, it is important to note that although family and friends combined provide 35.9 percent of all loans, these sources are closely followed by Acleda Bank with 33.9 percent. Moneylenders account for 19.2 percent. Only a few years ago the formal banking sector would have accounted for a much lower percentage of loans, while the share for relatives, friends and moneylenders would have been much greater. NGOs and MFIs account for another 10 percent of loans for the formal sector. The research hypothesis predicts that land titles will increase household access to formal commercial loans. If this is the case, we should observe an increase in the share of loans from Acleda and other commercial banks Based on discussions with bank loan officers, this shift may already be under way.

Table 5.30 shows the distribution of loan uses across consumption quintiles. There are several aspects of the data that bear closer scrutiny, especially regarding key land-titling impact indicators such as business activity, real estate and housing and farming. A total of 35.8 percent of the loans were used for business-related purposes. Real estate, home improvements and housing construction together accounted for another 20 percent of all reported loans, while farming accounted for 7 percent. Food, health, and education accounted for another 22.8 percent of loans, while loan repayment and transport accounted for 12.4 percent.

Credit Use	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	То	tal
						No.	%
Farming	25.7	40.0	22.9	8.6	2.9	35	7.0
Business Start- up	16.7	38.3	20.0	11.7	13.3	60	12.0
Business Expansion	15.2	22.8	13.9	24.1	24.1	79	15.8
Business Input	32.5	17.5	20.0	22.5	7.5	40	8.0
Food	45.7	28.7	8.6	14.3	2.9	35	7.0
Health	31.7	18.3	16.7	13.3	20.0	60	12.0
Education	5.3	42.1	31.6	15.8	5.3	19	3.8
Real Estate	12.9	38.7	19.4	6.5	22.6	31	6.2
Loan Repay	41.7	12.5	16.7	16.7	12.5	24	4.8
Home Improve	35.5	16.1	16.1	12.9	19.4	31	6.2
Home Constr.	31.6	10.5	13.2	31.6	13.2	38	7.6
Transport	31.6	15.8	36.8	10.5	5.3	19	3.8
Other	41.4	24.1	17.2	20.7	10.3	29	5.8
Total	135	125	90	80	70	500 *	100
% of Total	27.0	25.0	18.0	16.0	14.0		

 Table 5.30: Credit Use, by Consumption Quintile (percent of loans)

* 1 missing

Again, the research hypotheses predict that the volume of commercial loans will increase as households use their land titles as collateral for bank credit. The research hypotheses explicitly indicate that two of the most visible areas are likely to be business and the housing and real estate sector. Recent discussions with banking and realty agents suggest that this is indeed likely to be the case. For example, although it appears people are continuing to use own resources for business start-ups, there has been an observed increase in the number of loans for business operations and expansion. This is attributed to the fact that people are less willing to borrow at the outset because of the perceived risks, but as their business shows signs of stability and profitability, they are more willing to obtain credit for expansion. There is also anecdotal evidence of an increase in borrowing for real estate, including speculation, and housing investments. This is likely to continue at least for several years as land values will continue increasing in the near to medium term. As a result, the share of loans for these two sectors is likely to increase, while the shares for other sectors will decrease. It is possible that the percentage of loans for farming will decrease along with shifts in land use patterns away from agriculture in the direction of residential and commercial or industrial uses.

Finally, it should also be noted that another indicator of the level of loan activity is the average amount of each loan. The research hypothesis predicts that as people increase the number of commercial bank loans they obtain using land titles as collateral, the use of the loan will shift along with the amount. If this is the case, then the average amount of loans secured with land titles should increase over time, bearing in mind that increasing land values will increase the amount of lending for which people are eligible. These trends may be fairly consistent across all quintiles, which suggests that those with more valuable land may benefit more from land-titling in this regard.

5B.6. Small and Medium Enterprises

Of the survey group, 439 households (28.4 percent) reported owning 470 small or medium businesses, 1.09 SMEs per owner. About 60.7 percent (286) of the SMEs were owned by female heads of households. Households in the uppermost quintile owned almost 23 percent (107) of the businesses, while 12.5 percent of the businesses were owned by the lowest quintile. The top two quintiles accounted for 47.6 percent (224) of the businesses, while the lower two quintiles accounted for 31.6 percent (149). Almost 58 percent of the SMEs have been started since 2000. Table 5.31 shows that the number of business start-ups has increased each year since 2001. Moreover, the increases have been steady in all quintiles, except for the top one.

Year	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Total	% of Total
2000	5	4	8	12	12	41	15.0
2001	2	5	4	7	4	22	8.1
2002	7	4	5	8	6	30	11.0
2003	9	9	14	9	10	51	18.7
2004	5	11	13	15	9	53	19.4
2005	17	14	17	20	8	76	27.8
Total	45	47	61	71	49	273	
% of Total	16.5	17.2	22.3	26.0	17.9		

 Table 5.31: SME Start-ups, by Year and Quintile

5B.6.1. SME Financing

Table 5.32 identifies 479 sources of financing for SMEs reported by the Phnom Penh survey group. Almost 92 percent of small business financing comes from own resources and savings, while family makes up 6.1 percent of the sources. So far both informal moneylenders and the formal commercial sector appear to play practically no role in financing SMEs. Some 95.5 percent of new businesses were started with own resources or savings, while 88.2 percent of

the businesses were sustained with own resources and 76.1 percent of expansions came from own resources.

Quintile	Own	Family	Moneylender	Private	Other	Total
	Resources	_	-	Bank		
1	60	0	0	0	0	60
2	84	5	0	1	0	90
3	92	7	1	1	1	102
4	106	8	0	1	1	116
5	98	9	0	1	3	111
Total	440	29	1	4	5	479
% of Total	91.9	6.1	.2	.8	1.0	

Table 5.32: Small Business Financing, by Sources and Consumption Quintile

These data closely parallel the information from the smaller survey in Sihanoukville's Sangkat 2 in 2004, which found that 93.0 percent of the financial sources for business startups involved own resources, which included family resources. This also closely parallels Kang's 2005 findings, in which 84.6 percent of business start-ups were financed with own resources. It is also interesting that in the rural land-titling baseline survey, more than 94.0 percent of agricultural input expenditures were purchased with own sources. The data collected in the Phnom Penh survey appear to be consistent with other surveys that suggest productive investments in both rural and urban sectors are still largely made with own resources and savings, and that there is as yet little commercial borrowing for such investments.

This observation raises the question of why so many small business owners continue to rely on own resources and family to finance business start-ups and maintain or expand businesses. Owners were asked to rank the degree to which they agreed with a particular reason for not seeking a loan. More than 81 percent of respondents indicated that they either "strongly agreed" or "somewhat agreed" with the reason that expected profit would not be higher than the interest. About 44.2 percent indicated that they strongly or somewhat agreed that collateral requirements inhibited them from seeking commercial loans, while 51.3 percent strongly or somewhat agreed that they have been refused a loan because of a lack of collateral. Other reasons for not seeking a loan included complicated bank procedures (33.7 percent), informal payments (30.8 percent) and being uninformed about banking requirements (29.4 percent).

While problems associated with collateral appear to play a role in some people's decisions not to seek a loan, the data clearly suggest that small business owners also perceive other important constraints on obtaining loans. Business profitability relative to interest rates is especially important. This suggests that while land titles may improve access to commercial loans by providing collateral, other constraints must also be addressed in order for the collateral effects of land titles to be fully realised.

There is a seeming discrepancy in the data concerning credit uses (Table 5.30) and the data concerning SME financing. The credit data above show a total of 179 loans during the period 2003–05 for business activities, while the data for SME financing show an almost negligible number of loans for any aspect of the SMEs covered in the survey. One might expect to find a higher percentage of loans from commercial sources in the SME sector. However, SMEs are a sub-sector of the business activities referred to in the above discussion. The number of business expansions referred to in the credit discussion, for example, may refer to businesses other than SMEs.²¹

²¹ For a more complete discussion, see Ballard and Phim (2007).

5B.6.2. SME Employment

The survey data suggest that SMEs are a potentially important source of employment in Phnom Penh. Table 5.33 shows that the number of employees, including permanent employees, daily wage labour and employed family members, steadily increases with the consumption quintile of the owner. It is also important that the total number of employees, including each labour category, tends to increase over time. The total number of employees at the time of start-up for 436 SMEs was 1037, an average of 2.38 employees per SME, while the number of employees at the time of the interview was 1364, an average of 3.13. The employment impact of land titles therefore may be significant if titles help stimulate increased SME investment and expansion by promoting a more secure tenure and providing access to more affordable credit.

Quintile	No. of	Permanent		Daily	Daily Labour		Members	Total	
	Businesses								
		Start*	Current	Start	Current	Start	Current	Start	Current
1	55	3.9	2.2	8.3	8.3	107.3	118.3	120	129
2	84	4.2	15.1	9.2	19.3	109.2	164.6	123	199
3	92	38.6	56.1	13.8	39.6	194.1	215.3	247	311
4	104	80.1	92.6	5.2	9.4	210.1	222.6	295	325
5	101	112.1	144.4	22.2	48.5	115.1	208.1	362	401
Total	436	239	310	59	125	739	929	1037	1364

Table 5.33: SME Employment, by Type and Consumption Quintile

* Fractional figures refer to average number of employees per SME.

According to recent discussions with banking officials, the number of loan applications and grants has increased dramatically over the past year, especially for business expansions, including SMEs. According to anecdotal evidence, investors are still reluctant to borrow from commercial banks for business start-ups because they are not willing to take loans when they are unsure of the return on the investment. As a result, business start-ups continue to rely primarily on own sources and savings as well as family and friends. However, once the business is up and running and the owner sees a steady and predictable flow of returns, he or she may be more willing to obtain a commercial loan to expand.

Such shifts in attitudes concerning commercial banks appear to be a relatively recent phenomenon. As the business climate in Cambodia improves and business investors develop more trust in the banking sector, it is likely that the trend toward increased demand for credit for a variety of purposes will continue. Because commercial banks require "hard" land titles to secure loans, it is quite likely that land titles will contribute to increased consumer and business borrowing. However, such an increase probably could not be solely attributable to land titles, but to the convergence of several factors to form a virtuous cycle of development and growth. These variables would include increased tenure security via land titles, business experience and skill (i.e., entrepreneurship), political stability and security, banks that want to lend and know how to do so and people's trust in banks. The development of land and credit markets and businesses all seem to rely on many of the same factors involving trust in institutions. Once these factors are reasonably in place, land and credit markets may function more efficiently, thus enabling entrepreneurs to predict returns with greater accuracy and reliability.

5B.7. Expected Benefits of Land Titling

Table 5.34 shows that more than 78 percent of respondents believed tenure security is the primary benefit from land titles. Assuming that land conflicts are also a function of insecure tenure, it then appears that 85 percent expected the primary benefit to concern improved

tenure security. Only 8.8 percent of respondents expected the primary benefit of land titles to be improved access to formal credit. If "facilitating transfers" refers to improved efficiency in registering transactions, and "facilitating sales" involves making the property more attractive to potential buyers, it appears that about 5.7 percent of respondents expected the primary benefit of titles to involve land transactions and transfers.

Quintile	Secure	End	Obtain	Facilitate	Facilitate	Nothing	Total
	Tenure	Conflict	Loan	Transfer	Sale	Different	
1	274	11	33	6	7	1	332
2	254	22	30	10	16	1	333
3	247	33	24	16	12	1	333
4	256	29	29	8	8	3	333
5	267	22	31	6	6	0	332
Total	1298	117	147	46	49	6	1663
% of Total	78.1	7.0	8.8	2.8	2.9	.4	

Table 5.34: Expected Benefits, by Consumption Quintile

The importance of land tenure security as an expected benefit is underscored by the responses concerning expected secondary benefits. About 40.7 percent of respondents identified ending conflict as the second most important benefit, while another 13.4 percent identified "stability in owning the land", for a total of 54.1 percent of responses that concerned land tenure security. About 20.3 percent identified "ease in borrowing money" as the second most important benefit, which suggests that access to credit is an important consideration for many people. Another 12.7 identified "ease in transferring inheritance" and 11.4 percent "ease of selling", for a total of 24.1 percent of responses that concerned improved facilities for land transactions.

Another survey conducted in 2006 found that 97 percent of respondents perceived "improved tenure, as well as a reduction of land disputes" as benefits of the land-titling programme, while 77 percent of respondents reported a reduction in the number of disputes since land titles were issued (Deutsch, 2006). Although the sample was rather small, 309 households in all, and spread over both rural and urban areas, including Phnom Penh, the data also suggested that people expect land titles to provide better security of tenure, a perception that is consistent with urban baseline survey data. Only about 20 percent of the respondents reported making improvements on or upgrading their property, "at least in the short term". This suggests that there are other intervening factors that influence a household's decision to improve or invest in land and housing. In other words, secure titles may be necessary in the long run, but they are not sufficient on their own to stimulate investments in real estate and housing.

5B.8. Summary: Expected Land-Titling Impacts in Phnom Penh

The expected land-titling benefits in Phnom Penh must be understood in the wider context of the municipality's rapid growth and longer-term development plans. The rapid growth of the real estate and housing markets, the expanding business sector, including SMEs and FDI, along with the expanding consumer loans, home mortgages and commercial investments, suggest that consumer and business confidence is high in the short to medium term. Therefore the introduction of the LMAP systematic land-titling programme appears to be quite timely in terms of promoting urban economic growth. Given the wide disparities in wealth, however, the impact of land titles on economic growth must also be understood within the context of expected social impacts, primarily with respect to poverty reduction, equity and gender.

The expected results of land titles and their associated tenure security include investments in land improvement and housing, investments in business start-ups and expansion, improved access to commercial credit, reduced conflicts over land and property, more efficient land markets that allocate land to more economically productive uses, increased use of the official registry to facilitate land transactions, increased revenues for government from land sales taxes and potential property taxes and, finally, poverty reduction. The recent trends in the key measurement indicators suggest that land titles are likely to have an important impact on many of these areas, and will therefore strengthen the institutional framework to sustain Cambodia's current growth.

Secure tenure is expected to stimulate investment in housing and home improvements. Lower quintile households may make modest improvements according to available resources, capacity to borrow and location of land relative to transport and services. Such activities can be measured by the quality of building materials and the extent of improvements. Upper quintile households, on the other hand, may have already made improvements in their housing, although they may add or improve other buildings on their plots. These households might instead purchase additional land for either productive or speculative purposes. They may also invest in housing complexes, which would increase the overall supply of housing. The distribution of such housing would depend on the price of individual units and rental markets. This is a matter of particular importance for the poor because access to good affordable housing is a key component of poverty reduction efforts in urban areas. Therefore it will also be important to monitor housing rental prices throughout the municipality.

Land titles may have an indirect impact on land use patterns to the extent they facilitate more efficient transactions that provide greater returns on investments. It is possible that in the three urban districts a shift will occur as residential property is converted to commercial or rental property, while in the peri-urban areas changes in land use patterns may involve a shift from residential and agricultural uses to commercial and industrial uses. The nature and direction of land use changes in and around Phnom Penh will also be highly dependent on the municipality's long-term development strategy. A key component of the strategy will be transparent land use planning and enforceable zoning.

Transactions will be the primary mode of land acquisition, and hence of land redistribution, throughout the municipality. As a result, the share of land acquired through state allocation is likely to decline, perhaps sharply, unless some social land concession mechanism is established. Such shifts will not be the direct result of land titles, but rather the result of expanding land markets for productive and speculative investment. Land-titling, however, will surely facilitate the process by reducing the transaction costs associated with contractual exchanges. As a result, land titles will also serve as the key institutional mechanism through which sufficient land assets can be mobilised for medium and large-scale development projects.

The degree to which such transactions will be recorded with the official registry is another matter. The survey data show people rely primarily on documents that are not legal in terms of validating ownership. There is anecdotal evidence that people continue to use such documents to validate land transactions at the village or commune level even after the issuance of the LMAP titles. This is of fundamental importance because a key reason for undertaking systematic land titling has been to facilitate land transfers through the official registry. Transactions that take place outside the official registry will also contribute to continued conflicts over land and represent a major source of lost revenue for the government.

The limited number of cases of conflict reported in the survey will not be useful for measuring the extent to which land titles will reduce land conflicts. In the follow-up survey, interviewees will need to be asked more specifically about their perceptions of the role land titles may or may not have played in reducing conflicts. It will be especially important to inquire about the role that land titles have played in resolving conflicts that arise after titles

were issued. Another important line of inquiry in this regard will concern the role the courts play in enforcing land titles in cases of conflict.

The research hypotheses predict that the volume of commercial loans will increase as households use land titles as collateral for bank credit. Two of the most visible areas of impact are likely to be housing and real estate, as well as business. Although it appears that people still use own resources for business start-ups, there is anecdotal evidence of a recent increase in the number of loans for business operations and expansion. The indirect employment impact of land titles may be significant if they help stimulate increased investment in business start-ups and expansions.

There is also evidence of an upsurge in borrowing for real estate, including speculation, and housing investments. Land titles are expected to sustain, if not accelerate, these trends in the near to medium term. As a result, the number of loans is expected to increase along with the average amount of loans. The use of loans is likely to shift as well as more people borrow from commercial banks. These trends may be fairly consistent across quintile groups, suggesting that upper quintile households may benefit more from titles than lower ones because the higher value of their land will make them eligible for larger loans.

The research hypotheses predict that land titles will have a wide range of impacts on genderrelated matters pertaining to land. Many indicators identified in the rural baseline survey, such as land holdings and access to affordable credit, are also applicable to assessing the impact of land-titling on gender in urban and peri-urban areas. Other indicators include investments in housing and business start-ups and expansion. It is expected that female-headed households will obtain an increased number of loans that are used for investing in housing and business improvements. However, the rate at which female-headed households obtain loans may depend on a variety of circumstances, including household assets.

The research hypotheses also predict that women's tenure security will be strengthened. One way to assess this will be to examine the outcome of cases involving the death of a husband or divorce and the extent to which the spouse's or widow's land rights have been upheld. Another way to assess this will be to examine cases involving households headed by single women. While household surveys may enable researchers to count the number of such cases, a qualitative approach will also be required to understand better the social and legal dynamics of such cases.

The data collected in the Phnom Penh survey largely conformed to expected trends and patterns and therefore appear to be reliable. Only in a few instances, such as the number of upper quintile households using zinc roofing and the surprisingly low number of reported land-related conflicts, did the data show questionable or puzzling results. More rigorous efforts concerning land and building valuation techniques and procedures should be employed in the follow-up survey. It is suggested that MLMUPC personnel familiar with land valuation techniques be more involved in both helping train field enumerators and helping oversee quality control of valuation in the field. Most importantly, the follow-up survey should incorporate qualitative research tools in order to provide more substance and texture to the household data, because many of the important nuances concerning the social and economic impacts of land titles cannot be captured well by a standard household survey.

Chapter 6. Siem Reap Baseline Survey

This section concerns the survey that was conducted in Siem Reap district from 5 to 17 December 2005. It provides an overview of some of the key aspects of Siem Reap district and city, including locational and demographic features, tourism development, business and SME development, land transactions and the role of the Apsara Authority²² in land use management. The second part of this section discusses the Siem Reap baseline survey data from Sala Kamraeuk and Siem Reap communes, following the same outline used above in presenting the Phnom Penh material. A summary of the key observations from the Siem Reap survey concludes this section.

6A.0. Background Overview²³

Siem Reap district is situated in the centre of Siem Reap province in Cambodia's north-west. The economy of Siem Reap town and the surrounding area is largely driven by rapidly expanding tourism centred on Angkor Wat and other ancient temple sites in the area. The residential population of the district in 2004 was estimated to be about 139,000, with an additional 16,000 who commuted from surrounding areas to work in tourism and other activities. Planners estimate that by 2020 the population may be as high as 211,000, with as many as 65,000 additional commuters. A great deal of this population growth will be fuelled by migrants seeking employment in the tourism industry. The scale of recent migration is reflected in the fact that the population of Siem Reap was about 50,000 in the early 1990s, while the estimated population of Angkor Park was 20,000 in 1994 and has been recently estimated at 100,000.

According to the JICA draft master plan (2004), Siem Reap district comprises three types of area: urban, rural/heritage and around the Tonle Sap Lake. The urban area consists of an urbanised area and an urbanising area. The urbanised area consists of a central urban and a peri-urban area. The urbanised area lies within a radius of about 2 km from the intersection of National Road 6 and the Siem Reap River, and is densely populated. The built environment includes hotels, guesthouses, shops and commercial complexes, as well as housing for local residents and administration buildings. The central urban area has already been built up with grid roads constructed earlier and features administrative and commercial facilities. The peri-urban area surrounds the central urban area and also accommodates commercial, tourism and residential facilities. The road grid in these areas has not yet been completed because of the "disordered expansion of the residential area".

The urbanising area is defined as the area surrounding the urban area and being developed with urban activities/facilities. It contains a populated area mainly for residential or mixed residential and commercial use. It is expanding. The JICA draft plan (JICA: III-1-13) observes that new development in the urbanising areas is expanding rapidly, although changes from agriculture to residential or hotel uses have been sporadic and "without much coordination with the provision of infrastructure". As a result, the urbanising area is characterised by urban sprawl involving "narrow, unpaved and winding ... disconnected roads and a shortage or lack of infrastructure on site".

²² Authority for the Protection and Management of Angkor and the Region of Siem Reap.

 $^{^{23}}$ This section borrows heavily from the JICA draft master plan (2004).

Unlike in Phnom Penh, the direction of Siem Reap's growth is somewhat constrained by zoning regulations enforced by the Apsara Authority. As a result, Siem Reap is expected to grow eastward along National Route 6 and south-south-eastward in response to commercial and residential development requirements. Growth toward the north and west will not be as great because these areas overlap Apsara-designated protection zones.²⁴

Table 6.1 summarises the population projections through 2020. Population density in Sala Kamraeuk is projected to grow rapidly, from 20.5 persons per ha in 2004 to 52.5 in 2020.²⁵ This growth will include increased density in the already urbanised area of the commune as well increased population in the urbanising areas. As in the rapidly urbanising areas around Phnom Penh, this growth will come from natural population growth and immigration and will increase demand for housing, public services and infrastructure. This growth is expected to restructure the composition of land use in Sala Kamraeuk. Table 6.2 shows that the total of land devoted to residential and commercial uses will increase somewhat, while the most significant increase will be in mixed use, from 94.2 ha to 424 ha.

Commune	Area (ha)	Total Population			Density (pop/ha)				
		2004	2012	2020	2004	2012	2020		
S. Kamraeuk	841	17,229	32,099	44,145	20.5	38.2	52.5		
Siem Reap	5515	16,128	17,742	19,401	2.9	3.2	3.5		
Tot. District	35,612	139,566	170,687	210,643	3.9	4.8	5.9		
Source: UCA draft Master Plan (2004)									

Table 6.1: Projected Population Density in the Survey Areas

Source: JICA draft Master Plan (2004)

In Siem Reap commune, population is expected to grow at a much slower pace, resulting in an increase in density from 3.9 to 5.9 persons per hectare. This is expected to result in a modest shift in land use, as 45.4 ha will be converted from other uses to mixed. Such changes will most likely occur along the road connecting Siem Reap town to the Tonle Sap Lake, as well as various Angkor Heritage preservation sites located elsewhere in the commune. This will generate a modest increase in demand for housing, public services and infrastructure, but certainly not on the scale of Sala Kamraeuk.

Land Use Classifications	Sala Kam	raeuk	Siem	Reap
	Current	2020	Current	2020
Administration Area	1.3	1.3	0	0
Residential Area	279.5	306.4	0	0
Commercial Area	23.5	38.9	0	0
Mixed Use Area	94.2	424.3	0	45.4
Small/Medium Industrial Area	0.4	0.4	0	0
Tourism Area	18.8	33,1	0	0
Culture Area	1.2	0	0	0
Public Space	0.0	12.8	0	0
Sport/Recreation Area	0.0	0	0	0
Technical Infrastructure	0.0	0	0	0
Reservation Area	0.0	0	0	0
Sub-Total	419	817.3	0	45.4
Others (agriculture, roads, etc)	421.6	23.3	5515.3	5469.8
Total	840.6	840.6	5515.3	5515.3

Table 6.2: Composition of Land Use (hectares)

Source: JICA draft Master Plan (2004)

²⁴ Apsara is mandated to maintain and preserve the Angkor Archaeological Park and other temple sites throughout the district and province. In addition to Apsara, the provincial department of land management will also play an important role in enforcing zoning regulations throughout the district.

Of the four urbanised communes in Siem Reap district, only Sala Kamraeuk is considered 100 percent urban.
6A.1. Tourism Development

The number of international visitors to Siem Reap increased from 44,833 in 1995 to 560,947 in 2004 and 653,639 in 2006 through November. It is targeted to reach more than 2.3 million in 2010. The number of domestic visitors has also been impressive, growing from 264,000 in 2001 to 372,382 through November 2006. (See Figure 6.1.) Siem Reap town itself is not a main tourist attraction, but rather serves as the base for the main attractions, which are Angkor Wat and other ancient temples. The Siem Reap/Angkor 2020 master plan envisions "a beautiful and unique tourist city based on a harmony of history, arts and nature of Khmer". The plan outlines three objectives: (1) a quality tourism city where tourists can fully feel a touch of Khmer culture, history and arts; (2) a human-scale city in the atmosphere of Angkor; and (3) an environmentally friendly city. The draft plan proposes six over-arching strategies for achieving these three objectives. These are (a) promoting tourism focused on up-market tourists; (b) maximising local benefits from tourism; (c) making the town more attractive to tourists; (d) making the town more environmentally sustainable; (e) strengthening infrastructure for tourists and people; and (f) strengthening local administration and finance.





^{*} In Hing and Tout (2007)

Source: NIS, Statistical Yearbook 2005 and MoT (2006).

The rapid increase in the number of international and domestic tourists, along with increased immigration, is placing enormous strain on the built environment and the natural environment. The development of the town's transport, power, water supply and waste and sewage facilities is barely keeping pace with the demand. There are also many concerns regarding the impact of tourism on the area, including fears that the water table that supports Angkor Wat and other temples is shrinking, thus posing a threat to the stability of the attractions that are drawing the tourists in the first place. The sustainable management of tourism, infrastructure development and the natural environment requires that the city's proposed master plan, including zoning regulations, be strictly enforced.

6A.2. Business and SME Development

Siem Reap also serves as an increasingly important transport hub connecting to the north at Anlong Veng, as well as the administrative centre for the province. This role is spurring growth in manufacturing and services. For example, the share of those employed in the primary (agriculture) sector decreased from 38.1 percent in 1998 to 24.5 percent in 2004, while employment in manufacturing grew from 10.2 to 18.7 percent. Employment in the tertiary sector (services) grew from 51.7 to 56.8 percent during the same period (Ballard 2005). It is quite likely that the shift away from the primary sector will continue as land is converted from agriculture to mixed uses.

The rapid development of the business sector, including SMEs, therefore cannot be solely attributed to tourism. Nevertheless, the dynamic development of businesses associated with tourism is indicative of the scope and scale of the overall growth in the business sector in Siem Reap. Table 6.3 highlights the impressive growth in several tourism-related business activities. To this one could add the growth in the transport industry as well as the number of restaurants of varying quality.

	1998	2001	2004	2006	% Growth 1998–06
Hotels	216	247	299	351	63
Guest-Houses	147	370	615	742	405
Tour Agencies	137	236	302	382	179
Tour Guides	369	727	1371	2712	635

Table 6.3: Growth in Tourism Services, 1998–06*

Units of measurement: rooms for hotels and guesthouses, offices for tour agencies, persons for tour guides.

Source: MoT, Annual Report of Tourism Statistics (2006) * In Hing and Tuot (2007)

6A.3. Land Markets

In a qualitative survey of the linkages between local communities and the tourism industry in and around Siem Reap town, Ballard (2005) observed that in certain areas near the town, some families were selling land in order to enter the labour market, establish a business and/or purchase land further from town. For some families, it appeared to make economic sense to sell land now rather than continuing with farming, which has carried high risks.

A more recent study (Hing and Tuot, 2007) provides more evidence of the scope and scale of the emerging real estate markets in and around Siem Reap. A survey of 509 households from eight villages around Siem Reap town found that 27 percent of households had reported selling at least some of their land. About 70 percent of land sales took place before 1998, while more non-poor households than poor had sold land since 1998. Meanwhile, about 14 percent of the households had purchased some land. Only 8 percent of the poor households had bought land, while 18 percent of the non-poor households bought land, most of it (62.3 percent) purchased since 1999. Although most land purchases (82 percent) were in the same village, non-poor households purchased about 24 percent of their land outside the village, while only three percent of the poor buyers bought land outside the village. This probably reflects the fact that land prices are much higher outside the survey villages, as most of the survey sample is located within Apsara Zones 1 and 2, where land sales are closely regulated. Poor households cannot afford to buy land outside the village, while non-poor households are better able to do so.

Although the areas regulated by Apsara are not representative of the real estate markets operating elsewhere, the reasons that households gave for buying land provide some clues about trends in the real estate markets in and around Siem Reap. For example, poor households buy land more often for housing than do the non-poor, while the non-poor households tend to buy more land for speculation and savings. This observation conforms with general expectations because more non-poor households will have already satisfied their housing needs, and with more disposable income are more inclined to look elsewhere for investment and savings opportunities.

6A.4. Role of the Apsara Authority

The rural/heritage area referred to above is under the jurisdiction of the Apsara Authority, which is mandated with oversight responsibilities for the management, preservation and development of Siem Reap-Angkor sites. The government has designated five different

categories of land for protection. Zones 1 and 2 have the strictest regulations concerning land use and real estate markets. People generally are not allowed to undertake any development or modification of land or buildings in which they are living without special permission. Residents may renovate or repair dilapidated houses, or construct a new house to replace an old one. The sale or purchase of land by traders for business activities or tourism facilities is strictly prohibited. However, land can be traded within the village. These regulations have kept land prices substantially lower than elsewhere in the area. Land in Zones 1 and 2 is essentially excluded from the real estate markets surrounding them. There is some Zone 1 and 2 land in Siem Reap commune, but not in Sala Kamraeuk.

6B.0. Siem Reap Baseline Survey

At the time of the baseline survey, land titling was just getting under way. Siem Reap district was the initial focal point of LMAP work, and Sala Kamraeuk and Siem Reap were chosen as the first communes to be surveyed and titled. As the location and order of subsequent titling areas were not known, it was decided to proceed with the baseline survey work in Sala Kamraeuk and Siem Reap. Although this presents some difficulties with regard to a control group, it does provide an opportunity to survey and compare both an urban and a peri-urban area in Siem Reap. A total of 10 villages in the two communes were surveyed. (See Appendix B.)

6B.1. Household Characteristics

A total of 536 households were interviewed during 5–17 December 2005. Of these, 20.9 percent were headed by women, slightly less than the figure of 22.7 in Phnom Penh. A total of 179 households were surveyed in Siem Reap commune, including 16.2 percent that were headed by women. A total of 357 households were interviewed in Sala Kamraeuk, including 23.2 percent headed by women.

As in Phnom Penh, respondents were asked to report on various household expenditures over different periods. This information was aggregated to estimate a per capita annual consumption figure. Table 6.4 shows the average household consumption expenditures for the year prior to the survey. Unlike in Phnom Penh, the overall per capita consumption for female-headed households was higher than male-headed households, by 4.3 percent. Although the averages for both male- and female-headed households were fairly similar for the first four quintiles, the average for female-headed households in the uppermost quintile was about 17.6 percent higher than that of male-headed households.

Quintile	Male	HH	Female HH		No. HH	Mean	Range
	No.	Mean	No.	Mean			
1	83	267	24	255	107	264	47-366
2	84	439	23	447	107	366	366–513
3	87	629	21	617	108	514	514 - 736
4	84	954	23	950	107	740	740-1203
5	86	2154	21	2534	107	1,204	1204 – 9397
Total	424	894	112	932	536	902	47-9397

 Table 6.4: Consumption Quintiles (USD/year/person)

The average annual per capita consumption expenditure in Sala Kamraeuk was USD1001, while in Siem Reap commune it was USD705. In comparison to Phnom Penh, the average per capita consumption figure for Siem Reap commune was almost exactly the same as that of the peri-urban district of Meanchey, while the average for Sala Kamraeuk was lower than all three urban districts in Phnom Penh, but higher than the three peri-urban districts.

Table 6.5 shows the distribution of the survey households in the two communes according to per capita consumption and gender of household head.

Commune	Quint	ile 1	Quin	tile 2	Quin	ntile 3	Quin	tile 4	Quir	tile 5		Total	[
	М	F	М	F	Μ	F	Μ	F	Μ	F	Μ	F	Tot
Siem Reap	37	5	36	12	33	7	26	4	18	1	150	29	179
S. Kamraeuk	46	19	48	11	54	14	58	19	68	20	274	83	357
Total M/F	83	24	84	23	87	21	84	23	86	21	424	112	
Total HH	10	7	1()7	1	08	10)7	1	07			536

Table 6.5: Household Surveys, by Gender and Consumption

6B.2. Plot/Unit Characteristics

This section provides some of the key data on plot/unit characteristics covered in the Siem Reap survey. These data concern the mode of acquisition, the type of documentation to claim or validate ownership, the access, the services available and plot values. As in Phnom Penh, two different data sets are used. The most important set concerns the plot or unit where the owner resides or works and where the interview took place. This is referred to as Plot/Unit 1 throughout the discussion and is the primary unit of analysis for land-titling impacts. In Siem Reap, the total number of Plots/Units 1 was 536, equal to the number of household interviews.

The second data set concerns all the plots that are owned by the survey respondents. These data are occasionally compared to the Plot/Unit 1 data as appropriate to highlight certain points and to serve as a check on the Plot/Unit 1 data. The total number of "all plots" in this data set is 1109, about 2.07 plots per household. This is higher than the average of 1.70 plots per household in Phnom Penh, and much higher than the 1.36 per household found in the earlier 2004 Sihanoukville survey of Sangkat 2. About 87.8 percent of "all plots" are located in the same village (79.5 percent) or elsewhere in Siem Reap district (8.3 percent), while the remaining 12.2 percent were located in another district of Siem Reap province (10.0 percent) or elsewhere (2.2 percent).

6B.2.1. Mode of Acquisition

Table 6.6 shows the mode of acquisition for Plot/Unit 1 according to commune. The mode of acquisition for both communes was almost equally divided among state allocation (33.8 percent), purchase (33.6 percent) and inheritance (32.6 percent). This differs from Phnom Penh, where the predominant mode of acquisition was purchase (50.5 percent), followed by state allocation (35.4 percent) and inheritance (12.9 percent). This suggests that land markets in and around Siem Reap are less developed than in Phnom Penh.

Table 6.6 also shows that the predominant mode of acquisition for Plot/Unit 1 in Sala Kamraeuk is purchase (41.2 percent), followed by state allocation and inheritance at 29 and 28.9 percent, respectively. The predominant mode of acquisition in Siem Reap commune is state allocation (42.5 percent) and inheritance (39.2 percent), while only 18.4 percent of the plots were acquired by purchase. This pattern is similar to that of Phnom Penh, where the predominant mode of acquisition in the urban areas was purchase, while in the peri-urban areas it was state allocation, followed by inheritance. The percentage of plots acquired through purchase tended to increase with consumption quintile, while the percentage acquired through inheritance tended to decrease as consumption increased. The share of plots acquired through state allocation varied across consumption quintiles.

	-	,				,			
Commune	Give	en by	Inherited		Purchased		Other		Total
	Sta	ate							
	Ν	%	Ν	%	Ν	%	Ν	%	
Siem Reap	76	42.5	70	39.2	33	18.4	0	0	179
Sala Kamraeuk	107	29.0	103	28.9	147	41.2	2		357
Total	181	33.8	173	32.3	180	33.6	2		536

Table 6.6: Mode of Acquisition, by Commune (Plot/Unit 1)

Of the 180 plots acquired through purchase, only 56, about 31.1 percent, were bought since 2000. Less than half of these plots, 42.9 percent, were purchased during 2003–05. This is surprising, because one would expect more recent purchases due to the impression of a booming land market. There also does not appear to be much variation across consumption quintiles. As expected, however, about 80.4 percent of the purchases (45) since 2000 have taken place in Sala Kamraeuk. The data concerning the mode of acquisition for Plot/Unit 1 suggest that increasing population density in Sala Kamraeuk is driving the higher number of purchases compared to Siem Reap commune, which in turn suggests that the land market in Sala Kamraeuk will continue to grow over the next few years.

The data concerning the mode of acquisition for all plots reaffirm this impression. Table 6.7 shows that the percentage of all plots acquired through purchase is close to that of Plot/Unit 1, and the distribution of purchases by commune is also very similar.

Commune	Given by Inheri		erited	Purchased		Other		Total	
	Sta	ate							
	Ν	%	Ν	%	Ν	%	Ν	%	
Siem Reap	249	56.8	102	23.3	81	18.5	6		438
Sala Kamraeuk	237	35.3	148	22.1	284	42.3	2		671
Total	486	43.8	250	22.5	365	32.9	8	.8	1109

Table 6.7: Mode of Acquisition, by Commune (All Plots)

Table 6.8 shows that there is a significant variation in purchases across the quintiles. The top two quintiles represent 26.3 and 29.0, respectively, of the plots acquired through purchase, while the bottom two quintiles represent 12.9 percent and 13.0 percent, respectively. The bottom two quintiles account for 24.0 and 21.2 percent of all plots acquired through inheritance, while the upper two quintiles account for 22.0 and 14.4 percent, respectively. There is a remarkable degree of consistency across all consumption quintiles concerning those plots acquired through state allocation, all being in the range of 20.2 to 20.6 percent.

 Table 6.8: Mode of Acquisition, by Consumption Quintile (All Plots)

Quintile	Sta	te	Inhe	erit	Purc	chase	Ot	her		Total	
	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Tot
1	76	23	48	12	37	10	0	3	161	48	209
2	70	28	48	5	38	10	0	0	156	43	199
3	84	16	42	9	51	17	0	1	177	43	220
4	73	16	38	12	82	14	0	0	193	42	235
5	79	21	36	0	92	14	3	1	210	36	246
Total M/F	382	104	212	38	300	65	3	5	897	212	
% of M/F	42.6	49.1	23.6	17.9	33.4	30.7	0.3	2.3	80.1	19.9	
Total	48	6	25	0	3	65	8	3		1,109	
% of Total	43	.8	22	.5	3	2.9).)8			

6B.2.2. Documentation

Table 6.9 shows that households in Siem Reap use a variety of means to document land ownership but, as elsewhere, tended to rely on unofficial documentation. For Plot/Unit 1, this included receipts of certificate applications (35.1 percent) and survey papers (17.9 percent). Compared to Phnom Penh, households seemed to rely more on village or commune transfer documents (13.3 percent). Households in the Siem Reap survey also used certificates more often to document landownership than the households in Phnom Penh, 20.3 percent compared to 5.4 percent. This may reflect differences in the transaction costs associated with obtaining certificates. It could also be a reflection of the possible sampling bias away from better off households, discussed above in Section 4.4, as the wealthiest households are probably underrepresented in the survey sample. A higher proportion of the better off households would presumably have more of their land documented with certificates than less well-off households.

Commune	App.	Survey	Certificate	Sales	ComVil	No	Total
	Receipt	Paper		Letter	Transfer	Paper	
Siem Reap	27	91	58	0	2	0	178
S. Kamraeuk	161	5	51	6	69	65	357
Total N	188	96	109	6	71	65	535 *
% of Total	35.1	17.9	20.3	1.1	13.3	12.1	

 Table 6.9: Land Documentation, by Location (Plot/Unit 1)

* 1 missing

App. Receipt = *Application Receipt; ComVil* = *Commune or Village*

Of the three most common means of documentation, property that is documented with a certificate has a significantly higher reported value, USD80,936, followed by application receipts (USD56,406) and survey papers (USD21,680).²⁶ As in Phnom Penh, where Plots/Units 1 with certificates also had a significantly higher reported value (USD100,704), this raises the question whether the certificate itself adds value to the property, or households tend to obtain certificates for property that has a higher value. While secure documentation may indeed add some value to a property, it probably has more to do with the fact that people want to secure more valuable property with "hard" paper. Village transfer documents have an average value of USD30,971, while those in the commune have an average value of USD45,113. Interestingly, the 22 households that lost their application receipts had an average reported value of USD68,603, while 41 with no paper available in the house averaged USD48,165. As in Phnom Penh, this suggests the potential for conflict prior to or during the LMAP titling.

Table 6.10 shows that the pattern of documentation for all plots closely approximated the pattern for Plot/Unit 1. The fact that there were more certificates for Plots/Units 1 (20.3 percent) than for all plots (16.9 percent) suggests that additional plots may be further from town, where land values are less. Although the percentage of application receipts and survey papers varied, these two types of not-legal documents together accounted for 54.5 percent of all plots, compared to 53.0 percent for Plot/Unit 1. The percentages of transfers documented at the village and commune level are quite similar, and the percentage of households with no paper or lost paper was identical. The similarities between the two data sets suggest that households have a consistent approach to documenting land ownership, regardless of its location.

²⁶ "Agreement papers"—or contracts—between buyer and seller had a higher average value, USD98,030, but the number of cases was small (six).

Commune	App.	Survey	Certifica	Sales	ComVil	No	Total
	Receipt	Paper	te	Letter	Transfer	Paper	
Siem Reap	74	218	118	1	13	14	438
S. Kamraeuk	270	43	69	13	156	120	671
Total N	344	261	187	14	169	134	1109
% of Total	31.0	23.5	16.9	1.2	15.2	12.1	

 Table 6.10: Land Documentation, by Location (All plots)

App. Receipt = Application Receipt; ComVil = Commune or Village

6B.2.3. Location and Services

Table 6.11 shows the type of access for Plot/Unit 1 in both communes. About 35.3 percent of the survey plots were located along main paved roads or 2-way access streets, paved or otherwise. This is a much lower percentage than in Phnom Penh (61.2 percent). As in Phnom Penh, special efforts were made to survey households with more difficult access, and this is reflected in the number of surveyed plots with one-way access (31.7 percent) or bicycle and foot access (23.3 percent), both of which are greater than in Phnom Penh. Although there is not a general survey of plot access in Siem Reap or Phnom Penh, the sample is probably more representative than that of Phnom Penh.

Table 6.11: Access, by	Commune	(Plot/Unit 1)
------------------------	----------------	---------------

Commune	Main Paved	2-way	1-way	Bicycle	Foot	Total
	Road	Access*	Access*	Access	Access	
Siem Reap	38	52	41	47	1	179
S. Kamraeuk	46	105	129	76	1	357
Total	84	157	170	123	2	536

* For a vehicle (Please see Survey Questionnaire, Question 2.1

As in Phnom Penh, households were asked about the utilities and services they received, including electricity, phone (land line), sewerage and water for cooking. Table 6.12 shows that 76.9 percent of the households had electricity service, which was somewhat higher than the survey households in Phnom Penh (61.7 percent). As expected, land located along main paved roads had a much higher rate of service (91.6 percent) than other locations. About 76.2 percent of households in Sala Kamraeuk had electricity service, while 78.2 of households in Siem Reap had it. The situation concerning sewerage was quite different. Only 37.1 of households reported having city sewage service, while no households in Sala Kamraeuk had it.

 Table 6.12: Electricity Services, by Type of Access (No. of households)

Commune	Main Paved	2-way	1-way	Bike Access	Foot	Total
	Road	Access*	Access*		Access	
Siem Reap	34	37	34	34	1	140
S. Kamraeuk	43	75	102	52	0	272
Total	77	112	136	86	1	412
% of Total	91.6	71.3	80.0	69.9	50.0	76.9

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

6B.2.4. Land Use and Value

This section presents some of the key data concerning land use and plot values in the two survey communes.

6B.2.4.1. Plot/Unit Types and Land Uses

Table 6.13 summarises Plot/Unit 1 according to consumption quintile and use. Almost all (97.1 percent) of the properties in the survey were used as residences only, or residences plus other uses. This was more than in Phnom Penh (88.6 percent), although the percentages were similar for residential only and residential commercial. In the Siem Reap group, there were no commercial only properties, while in Phnom Penh there were only 0.7 percent of properties used only for commercial purposes. As in Phnom Penh, the percentage of residential only property tended to decrease as consumption level increased, while residential/commercial use appeared to increase along with consumption. This pattern was more pronounced in Phnom Penh.

Quintile	Residential	Res/Com	Rental/Ag	Res/Rent	Res/Ag	Total
Quintile 1	70	11	3	4	19	107
Quintile 2	69	18	3	8	9	107
Quintile 3	54	24	2	13	15	108
Quintile 4	56	22	1	10	18	107
Quintile 5	49	28	2	13	15	107
Total	298	103	11	48	76	536
% of Total	55.6	19.2	2.1	9.0	14.2	

 Table 6.13: Plot/Unit 1 Use, by Consumption Quintile (No. of Households)

As discussed regarding Phnom Penh, land titles may have an indirect impact on land use to the extent that they facilitate more efficient transactions for greater returns on investments. About 51 percent of the properties in Sala Kamraeuk are residential only. Given the expected population growth and concomitant increase in the demand for housing, the percentage of residential only properties could increase. As rental prices increase, the residential/commercial and residential/agricultural land will probably decrease, as more property is converted to rental or mixed uses. The nature and direction of land use changes in and around Siem Reap may more closely approximate the pattern in the peri-urban areas of Phnom Penh, where changes may involve a shift from residential and agricultural to commercial uses. As in Phnom Penh, the direction and pace of land use change in Siem Reap will depend on the city's longer-term development strategy, a key component of which is clear and enforceable zoning.

6B.2.4.2. Land Values

Respondents were asked how much they would receive if they were to sell their Plot/Unit 1 at the time of the interview. Although the same caveats exist in Siem Reap as in Phnom Penh concerning the possibility of respondents either overestimating or underestimating the value of their property, the reported plot values in Siem Reap also closely conform to expected trends. Table 6.14 shows that average plot values in the urban commune of Sala Kamraeuk were significantly higher than the average plot values in the peri-urban commune of Siem Reap. In fact, the average reported value of property in Sala Kamraeuk was almost three times that in Siem Reap commune.

Second, average plot values steadily increased along with better access in both communes, in part because access to services is generally easier. Although main paved roads and 2-way access streets both represent two-way access, the reported average value of plots situated

along main paved roads was significantly more than the average reported value of plots situated along 2-way access streets.

Commune	Main Paved	2-way	1-way	Bicycle	Foot	Total	
	Road	Access*	Access*		Access		
						No	Ave \$
Siem Reap	38,979	28,839	15,455	12,581	2016	178	23,420
Sala Kamraeuk	149,386	81,195	52,858	28,900	504	354	68,422
Average Value	100,168	63,631	43,837	22,613	1260		
Total HH	83	155	170	122	2		532**

Table 6.14: Land Values, by Access and Commune (USD/plot)

* For a vehicle (Please see Survey Questionnaire, Question 2.1. ** 4 missing

Again as in Phnom Penh, the data conform closely to expectations when plot values are analysed according to consumption quintiles. As Table 6.15 shows, the average value of the plot steadily increased along with consumption quintile in both communes. The average plot value ranged from USD27,984 in the first quintile to USD94,404 in the fifth, while the range among the middle three quintiles was of course less, USD36,404 to USD57,254).

Commune	Quint 1	Quint 2	Quint 3	Quint 4	Quint 5	۲.	Fotal
						HH	Ave \$
Siem Reap	14,313	19,790	25,883	31,879	34,781	178	23,420
Sala Kamraeuk	36,955	50,153	66,049	67,254	106,740	354	68,422
Average Value	27,984	36,404	51,173	57,254	94,404	53,365	
Total HH	106	106	108	107	105	532 *	

 Table 6.15: Reported Plot/Unit 1 Values, by Commune and Consumption Quintile (USD)

* 4 missing

6B.2.5. Buildings and Housing

Data concerning the number, use and quality of buildings on the survey plot were also collected. This information will be important for assessing potential land-titling impacts according to the hypothesis that people will take advantage of more secure tenure to invest in housing and building improvements (which may also include more buildings), and may use titles as collateral for credit for making such improvements. Table 6.16 shows that most plots in the Siem Reap survey group, about 64.6 percent, had one building. In this section, only a small sample of the available data for the primary building is presented.

 Table 6.16: No. of Buildings per Plot/Unit 1, by Consumption Quintile

Quintile	1	2	3	4	5	Total
1	72	22	12	1	0	107
2	70	21	14	2	0	107
3	72	26	9	1	0	108
4	72	26	6	2	1	107
5	60	31	13	3	0	107
Total	346	126	54	9	1	536
% of Total	64.6	23.5	10.1	1.7	0.1	

Information was collected on the building materials used for roofing, walls, floors and fencing. As roofing is generally considered a good indicator of household consumption level and well-being, one would expect to find a higher proportion of low-cost roofing among the

lower quintile households than among the higher. Over time, the overall quality of roofing would be expected to change, with houses at each consumption level upgrading the quality of their roofing material, although upper quintile households might remain static because they were already using better materials. The data concerning roofing material are presented in Table 6.17 and suggest that such an assumption may be misleading. As in Phnom Penh, although the quality of roofing generally did improve with quintile level, there was a somewhat higher than expected amount of zinc roofing in the upper two quintiles. As a result, roofing may not be a useful indicator of household consumption, at least in urban areas, and therefore not a useful indicator of land-titling impacts.

Quintile	Thatch/Tent	Zinc sheets	Tile	Wood/plywood	Concrete	
1	14	65	28	0	0	107
2	4	70	31	1	0	106
3	4	48	52	2	2	108
4	0	49	50	2	6	107
5	1	49	47	3	7	107
Total	23	281	208	8	15	535*

Table 6.17:	Roofing	Type,	bv	Consumption	Ouintile
Lable of L	1.cooming	- J P - ,	\sim_J	company	Zumme

* 1 missing

6B.3. Land Transactions

This section presents some of the key data concerning land purchases and sales reported by the survey households, as well as the documentation used to facilitate and validate sales. As with Phnom Penh, this discussion is particularly relevant for assessing the impact of landtitling in urban and peri-urban areas around Siem Reap in the context of emerging land markets and the role that land titles are expected to play in facilitating more efficient land transfers through the official registry.

6B.3.1. Purchases

The 536 survey households reported purchasing a total 365 plots since 1980. Of this number, 162, or 44.4 percent, were purchased since 2000. Table 6.18 shows the average reported purchase price of the plots/units acquired through purchase since 2000. More than half, 56.8 percent, of these purchases took place in the most recent three-year period, 2003–05. As expected, the number of purchases tended to increase along with consumption level, especially in Sala Kamraeuk. About 78.4 percent of the purchases since 2000 took place in Sala Kamraeuk.

Commune	Qui	ntile 1	Qui	ntile 2	Qui	ntile 3	Qui	ntile 4	Qui	ntile 5	То	otal
	Ν	Amt	Ν	Amt	N	Amt	Ν	Amt	Ν	Amt	N	Amt
Siem Reap	6	1946	5	545	9	10,272	6	4183	9	6994	35	5569
S.	21		14		21		32		37	16,950	125	
Kamraeuk		2593		3040		4971		6053				8279
Total	27	2449	19	3047	30	6,651	38	5758	46	15,002	160*	7686

Table 6.18: Average Purchase Price, 2000–05 (All Plots) (USD/plot)

* 2 missing

Table 6.18 also shows that the average reported value of all plot purchases between 2000 and 2005 was USD7686. As expected, the reported values were higher in the urban district of Sala Kamraeuk (USD8279) than in the peri-urban district of Siem Reap (USD5569), but the difference between the two was not as great as in Phnom Penh. Generally speaking, the average reported purchase price of all plots steadily increased according to consumption quintile, with the exception of the third quintile in Siem Reap commune. This was very

similar to the pattern found in Phnom Penh, although on a smaller scale in terms of average values. As in Phnom Penh, this trend suggests that those who are better off tend to pay more per plot when buying property.

As in Phnom Penh, it is interesting to compare the information in Table 6.18 with that in Table 6.19, which shows the estimated current value of each plot reported at the time of interview. Again as in Phnom Penh, the difference in the reported purchase price and the current value was remarkable. Across each consumption quintile and both communes, the current estimated plot value was significantly higher than the reported purchase price. While this could reflect understated purchase prices and/or deliberately inflated estimates of current value, the data provide a very good sense of people's perceptions and expectations of how the Siem Reap land market is developing. Generally speaking, the average reported purchase price during 2000–05 and the estimated current value differ by a factor of three in both communes and across most consumption quintiles.

Qui	ntile 1	Qui	ntile 2	Qu	intile 3	Qu	intile 4	Qu	intile 5	T	otal
Ν	Amt	Ν	Amt	Ν	Amt	Ν	Amt	Ν	Amt	N	Amt
6	4471	3	5067	9	39,138	6	8317	9	14,324	33	17,366
22	7909	14	29,048	21	15,604	32	21,032	37	45,892	126	26,027
28	7172	17	24,816	30	22,664	38	19,024	46	39,716	159*	24,229
	Quii N 6 22 28	Quintile 1 N Amt 6 4471 22 7909 28 7172	Quintile 1 Qui N Amt N 6 4471 3 22 7909 14 28 7172 17	Quintile 1 Quintile 2 N Amt N Amt 6 4471 3 5067 22 7909 14 29,048 28 7172 17 24,816	Quintile 1 Quintile 2 Qu N Amt N Amt N 6 4471 3 5067 9 22 7909 14 29,048 21 28 7172 17 24,816 30	Quintile 1 Quintile 2 Quintile 3 N Amt N Amt N Amt 6 4471 3 5067 9 39,138 22 7909 14 29,048 21 15,604	Quintile 1 Quintile 2 Quintile 3 Qu N Amt N Amt N Amt N 6 4471 3 5067 9 39,138 6 22 7909 14 29,048 21 15,604 32 - - - - - - 28 7172 17 24,816 30 22,664 38	Quintile 1 Quintile 2 Quintile 3 Quintile 4 N Amt N Amt N Amt N 6 4471 3 5067 9 39,138 6 8317 22 7909 14 29,048 21 15,604 32 21,032 - - - - - - - 28 7172 17 24,816 30 22,664 38 19,024	Quintile 1 Quintile 2 Quintile 3 Quintile 4 Qu N Amt N Amt N Amt N Amt N 6 4471 3 5067 9 39,138 6 8317 9 22 7909 14 29,048 21 15,604 32 21,032 37 28 7172 17 24,816 30 22,664 38 19,024 46	Quintile 1 Quintile 2 Quintile 3 Quintile 4 Quintile 5 N Amt M Amt N Amt M Amt N Amt N Amt M Amt </td <td>Quintile 1 Quintile 2 Quintile 3 Quintile 4 Quintile 5 T N Amt Amt N Amt Amt N Amt Amt Amt Amt Amt Amt Amt <</td>	Quintile 1 Quintile 2 Quintile 3 Quintile 4 Quintile 5 T N Amt Amt N Amt Amt N Amt Amt Amt Amt Amt Amt Amt <

 Table 6.19: Current (Dec. 2005) Estimated Value of All Plots (USD/Plot)

* 3 missing

Of the total plots purchased between 1980 and 2005, about 19 percent were purchased during 1980–89. As in Phnom Penh, it appears there was a nascent land market in Siem Reap even before 1989, when the government reintroduced private property rights (see Section 5B.3.1).

6B.3.2. Land Sales

Of the 536 survey households in Siem Reap, 181 (33.8 percent) reported selling a total of 209 plots since 1980, an average of 1.15 per selling household. About 77.5 percent (162) of plot sales took place in Sala Kamraeuk. The uppermost quintile accounted for about 29 percent of the plot sales, while the remaining sales were fairly evenly distributed across the other four quintiles.

Of these sales, 120 occurred during 2000–05. Table 6.20 shows that 78.3 percent of the plot sales took place in Sala Kamraeuk. Generally speaking, the land sales by year do not show a consistent pattern either in the number of sales or their average value. This is different from Phnom Penh, where 72 percent of the sales took place in peri-urban areas. As would be expected, however, the average reported value of sales was significantly higher in Sala Kamraeuk than in Siem Reap commune, in fact more than double. This is similar to Phnom Penh, where the average sale values were consistently higher in urban than in peri-urban areas.

Commune	2000	2001	2002	2003	2004	2005		Total
							Ν	Average
Siem Reap	2225	11,000	6500	14,500	12,910	650	26	8540
S. Kamraek	10,904	9556	11,565	16,805	13,976	49,302	94	19,347
Average	8975	9917	10,374	16,252	13,786	39,571]	17,005
Total N	18	12	17	25	28	20	120	
% of Total	15.0	10.0	14.2	20.8	23.3	16.6		

Table 6.20: Average Land Sale Values (USD), by Location (2000–05)

The average value of the reported land sales during 2000–05 was higher than the reported value of all plot purchases during the same period. In the case of Sala Kamraeuk, it was more than double the reported value of the purchased plots. However, it was not nearly as great as the difference between the average purchase price and the estimated current value of plots. This is also similar to Phnom Penh, although the range of values was much larger there.

6B.3.3. Land Sale Documentation

As in Phnom Penh, almost 90 percent of land sales were facilitated by changing names on documents below the provincial level. Table 6.21 shows that a majority of sales were documented in the commune level, about half that number in the village and fewer than a tenth at district level. While the percentage of sales documented in the commune was similar in the two communes, there was more documentation at village level in Siem Reap commune.

Quintile	Village	Commune	District	Province	Title/MLUPC	Contract	Total
1	16	18	2	0	0	1	37
2	8	25	3	0	0	2	39
3	9	18	3	0	0	2	32
4	9	26	3	1	0	1	40
5	15	27	5	2	3	9	61
Total No	57	114	16	3	4	15	209
% of total	27.3	54.5	7.7	1.4	1.9	7.1	

 Table 6.21: Land Sales, by Mode of Documentation

As in Phnom Penh, the high percentage of land sales documented in the commune across all consumption quintiles suggests that people view the commune as the locus of local governmental authority. While it has been suggested that a preference for documenting land sales at the village and commune level is also related to high transaction costs, which includes transportation costs, associated with provincial documentation, neither Sala Kamraeuk nor Siem Reap commune is far from the provincial cadastral office. This suggests there are other costs involved that explain why people avoid registering land transactions officially and legally, including the land sales tax and high unofficial fees. As in Phnom Penh and elsewhere, the fact that people avoid using the official registry is a matter of serious concern because transactions taking place outside the official registry are likely to contribute to continued land conflicts and represent lost revenue for the government.

6B.4. Conflicts

Households were asked, "Have you ever had conflicts over land or other property since the national election of 2003"? As in Phnom Penh, surprisingly few households (20) reported being involved in a conflict over land during the period from July 2003 to December 2005. This represented about 2.0 percent of all the plots covered in the survey, only slightly higher than the 1.3 percent of conflicted plots in the rural survey sample. As in Phnom Penh, one would expect more land conflicts in such a dynamic emerging land market. One reason there are so few reported conflicts may be that the LMAP is targeting initial titling in Siem Reap where there is less conflict over land. Also, in the areas surveyed, most land conflicts may have been resolved prior to 2003, and any fencing put in place since then would tend to secure boundaries. As in Phnom Penh, it is also possible that the extent of land-related conflicts has been under-reported.

The 20 cases were fairly evenly divided across consumption quintiles, while 12 cases occurred in Sala Kamraeuk. Ten of the cases occurred in 2005. Although this might suggest that conflicts may be increasing, the number is far too small to allow for a firm conclusion. Seventeen of the cases involved residential land, while the other three were mixed.

More than 60 percent of the households reported seeking resolution from either the village chief or the commune leader, while another 10 percent of the cases were resolved with the help of neighbours or through mutual agreement. As with documentation, this suggests a preference for managing conflicts locally. Fewer than half of the cases (seven), however, were reported to have been successfully resolved, and only one-third of the households reported satisfaction with the result.

6B.5. Credit and Finance

Among the 536 survey households, 135 (25.2 percent) reported obtaining a total of 195 loans between the national election of 2003 and the survey, making about 1.44 loans per borrowing household. The percentage of households reporting credit activity was only slightly higher than in Phnom Penh (21.6 percent). This suggests that urban and peri-urban landowners tend to borrow less than their rural counterparts, perhaps because they are generally better off and have less need for borrowing, as suggested in the discussion concerning credit activity in Phnom Penh. Also, the number of loans per borrowing household in Siem Reap was very similar to the rate of 1.40 in Phnom Penh.

In Siem Reap, there was a sharp increase in the number of borrowings reported over the three years: in 2003, there were 37 loans, in 2004 there were 58 and in 2005 there were 100. This pattern is quite similar to that in Phnom Penh. Of the loans, 28.7 percent came from relatives (25.1 percent) and friends (3.6 percent), while almost half (49.7 percent) were from formal institutions, including MFIs and commercial banks. Nearly 19 percent were from moneylenders. Again, this pattern was similar to that of Phnom Penh, although there was a higher percentage of loans from formal sources in Siem Reap than in Phnom Penh (33.9 percent).

Table 6.22 shows loan uses across consumption quintiles. Some 38.5 percent of all loans were used for business purposes. This closely approximated the percentage used for business purposes in Phnom Penh, 35.8 percent. Real estate, home improvements and housing construction together accounted for another 25.1 percent of loan uses, slightly more than the 20.0 percent in Phnom Penh. Food, health and education accounted for 13.8 percent, compared to 22.8 percent in Phnom Penh. Farming and animal raising was 6.8 percent, while loan repayments and transport accounted for 12.3 percent. These figures are almost equal to those of Phnom Penh, 7.0 percent and 12.4 percent, respectively.

Credit Use	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	То	tal
			Numb	er			%
Farming/animal	3	7	1	2	0	13	6.8
Bus Start-up	4	2	6	5	3	20	10.3
Bus Expansion	0	13	9	6	11	39	20.0
Bus Input	3	7	2	1	3	16	8.2
Food	7	1	1	3	0	12	6.1
Health	2	3	2	6	1	14	7.1
Education	0	0	0	0	1	1	.5
Real Estate	0	1	1	3	2	7	3.6
Loan Repay	2	1	2	0	0	5	2.6
Home Improve	3	1	2	1	0	7	3.6
Home Constr.	10	10	3	4	8	35	17.9
Transport	1	5	5	4	4	19	9.7
Other	1	0	0	4	2	7	3.6
Total	36	51	34	39	35	195	
% of Total	18.5	26.2	17.4	20.0	17.9		100

 Table 6.22: Credit Use, by Consumption Category

As in Phnom Penh, the volume of borrowing from commercial sources is expected to increase in part because land titles will enable more credit seekers to meet collateral requirements. It is expected that the shares of loans for business, real estate and housing will increase as a result, while other uses will decrease, and land use patterns will shift in the direction of more residential and commercial uses. This will be the case especially in Sala Kamraeuk, where the draft master plan for city development envisions increasing population density increasing the demand for housing up to 2020. Also as in Phnom Penh, it is expected that the average size of loans obtained using land titles as collateral will increase, as rising land values raise the amount for which borrowers will be eligible. Again, although these trends are likely to be fairly consistent across all consumption quintiles, upper quintile households will likely benefit more because their property against which loans can be secured tends to be of higher value than the property of lower quintile households.

6B.6. Small and Medium Enterprises

Of the survey group, 164 household respondents reported operating a total of 192 SMEs, which means that about 30.6 percent of the survey households operated an average of 1.17 SMEs. Almost half (49.2 percent) of business owners were female, less than the percentage of female owners in the Phnom Penh survey (60.7 percent). About 60.4 percent of the businesses (116) had been started since 2000, and of this number more than half (58.6 percent) were started during 2003–05. Households in the upper two quintiles owned 47.4 percent of the businesses, while the lower two quintiles owned 26.7 percent. Table 6.23 shows that the number of business start-ups increased each year from 2002.

Year	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	To	otal
						Tot	%
2000	2	2	7	1	5	17	14.7
2001	1	4	2	2	7	16	13.8
2002	6	1	2	3	3	15	12.9
2003	1	2	5	7	4	19	16.4
2004	1	2	6	6	6	21	18.1
2005	5	4	8	7	4	28	24.1
Total	16	15	30	26	29	116	
% of Total	13.8	12.9	25.9	22.4	25.0		100

Table 6.23: SME Start-ups, by Year and Quintile

6B.6.1. SME Financing

Table 6.24 identifies 194 sources of financing for SMEs reported by the Siem Reap survey group. Some 86.1 percent of SME financing came from own resources and savings, while family were 8.2 percent of the sources. The financing from own sources was slightly less than in Phnom Penh (almost 92 percent), while family financing was slightly higher. As in Phnom Penh, both informal moneylenders and the formal commercial sector appeared to play almost no role in financing SMEs.

Table 6.24: Sources of Small Business Financing, by Consumption Quintile

Quintile	Own Resources	Family	Private Bank/MFI	Other	Total
1	18		1		19
2	27		3	1	31
3	38	7	1	1	47
4	40	2	3		45
5	44	7	1		52
Total	167	16	9	2	194
% of Total	86.1	8.2	4.6	1.0	

Of the 100 sources of finance used to start a new business, 91 percent were from own sources and savings; of the 64 sources used to operate the business, 78.1 percent were own sources and savings. Of the 33 sources of credit used for expanding business, 84.8 percent were own sources and savings.

As in Phnom Penh, it appears that small business investors in Siem Reap are reluctant to borrow from commercial banks for business start-ups, because they are not willing to take loans when they are unsure of the returns on the investment. As a result, business start-ups continue to rely primarily on own sources and saving as well as family and friends. Once the business is up and running, however, and an owner sees a steady and predictable flow of returns, at least some entrepreneurs appear to be more willing to obtain a loan from commercial banks to expand their business.

The 135 owners who said they were not seeking a loan were asked to rank the degree to which they agreed with particular reasons for not doing so. About 75.5 percent of the respondents indicated that they either "strongly agreed" (40.7 percent) or "somewhat agreed" (34.8 percent) that profit from the business would not be higher than the interest. About 45.1 percent indicated that they strongly (28.1 percent) or somewhat agreed (17.0 percent) that collateral requirements inhibited them, while 43.7 percent strongly disagreed. Other reasons for not seeking a loan included complex bank procedures (33.7 percent), informal payments (30.8 percent) and being uninformed about banking requirements. Informal payments and time-consuming procedures were considered less important constraints.

As in Phnom Penh, the data clearly demonstrate that small business owners perceived several important constraints on obtaining loans. This suggests that while land titles may improve access to commercial loans by providing collateral, other constraints must also be addressed in order for this effect to be fully realised.

The same apparent discrepancy between data concerning credit and data concerning SME financing, and its explanation, exist in Siem Reap as in the case of Phnom Penh, discussed in Section 5B.6.1.

6B.6.2. SME Employment

As in Phnom Penh, the survey data suggest that SMEs are a potentially important source of employment. Table 6.25 shows that the number of employees generally increased with the consumption quintile of the business owner, with a significant increase in the number of employed in the top quintile. As in Phnom Penh, the total number of employees, in each category, increased over time. The total number of employees at the time of start-up for the 192 SMEs was 554, an average of 2.89 per SME, 20 percent higher than the average of 2.38 reported in Phnom Penh. The current average was 3.19 per SME, almost the same as in Phnom Penh (3.13). Again, this suggests that the employment impact of land titles may be significant if titles stimulate increased SME investment and business expansions.

Quintile	No. of Employers	Pern	Permanent		Daily Labour		Members	Total	
		Start	Current	Start	Current	Start	Current	Start	Current
1	19	0	0	8	10	30	33	38	43
2	30	1	1	5	5	56	55	62	61
3	46	1	3	7	5	90	93	98	101
4	45	11	11	0	7	79	80	90	98
5	52	40	72	128	133	98	104	266	309
Total	192	53	87	148	160	353	365	554	612
Employees / SME		0.28	0.45	0.77	0.83	0.84	1.90	2.89	3.19

Table 6.25: SME Employment, by Consumption Quintile

6B.7. Expected Benefits of Land Titling

Table 6.26 shows that 75.4 percent of survey respondents believed that tenure security is the primary benefit from land titles. If we assume that land conflicts are a function of insecure tenure, then almost 85 percent of survey households expect the primary impact to be improved tenure security. This is almost exactly the percentage found in the Phnom Penh survey. Also, 10.8 percent of the survey households in Siem Reap felt that the primary benefit of titles would be to improve access to formal credit; in Phnom Penh, 8.8 percent mentioned improved access to formal credit as the primary benefit. The remaining 4.3 percent mentioned "facilitating transfers" (improved efficiency of registering sales and inheritances), and "facilitating sales" (making the property more attractive to potential buyers).

Quintile	Secure	End	Obtain	Facilitate	Facilitate	Total
	Tenure	Conflict	Loan	Transfer	Sale	
1	86	9	8	1	3	107
2	80	9	11	4	3	107
3	85	8	13	2	0	108
4	77	15	10	5	0	107
5	76	10	16	3	2	107
Total	404	51	58	15	8	536
% of Total	75.4	9.5	10.8	2.8	1.5	

 Table 6.26: Expected Benefits of Land Titling, by Quintile

As in Phnom Penh, the importance of land tenure security was also underscored by the responses concerning expected secondary benefits. About 41.8 percent of Siem Reap households identified ending conflicts as the second most important expected benefit, while another 16.6 percent identified "stability in owning land", making a total of 58.4 percent of responses that concerned land tenure security. This closely approximated the total of 54 percent found in Phnom Penh. About 22.4 percent identified "ease in borrowing money" as the second most important benefit. In Phnom Penh, the percentage was 20.3, 12.7 percent identified "ease in transferring inheritance" and 11.4 percent "ease in selling", a total of 24.1 percent that concerned facilitating land transactions.

6B.8. Summary of Siem Reap Land Titling Impacts

The potential impacts of land titling in Siem Reap district must be considered in the context of the city's booming tourism industry and rapidly increasing population, both of which are now placing considerable demands and strains on land resources. Until now, much urban and tourism infrastructure has been unplanned and haphazard in the absence of enforceable zoning regulations, resulting in anarchic development in some areas. How the city responds to such challenges in the near future will be guided by a medium-term (to 2020) city master development plan prepared with assistance from JICA. In terms of land use planning and infrastructure development, land titles may play an important role in strengthening the institutional fabric of the city's development, as envisioned in the draft master plan, provided that zoning regulations are rigorously and objectively enforced.

It is important to clarify and designate land use areas and boundaries of private land, state private land and state public land. Although the systematic titling of private land by LMAP represents an important step, it can make land management more problematic if the titling is confused with the other two processes. For this reason, it will be important to demarcate state private land and state public land prior to land titling. It will also be important to ensure that any conversion of state land to private use be done in an orderly and transparent fashion according to the overall urban development plan. In areas where land use is transferred in

CDRI

anarchic ways, land titles will eventually have to be adjusted to make space for public services. Such a process would be costly and burdened with conflict.

As in Phnom Penh, land titles could also have a significant, albeit indirect or complementary, impact on land use patterns to the extent that they facilitate more efficient transactions and provide greater and more predictable returns on investments. The nature and direction of land use impacts will be highly dependent on the district's medium-term development strategy. As with the Apsara Authority's management of rural heritage land, land management in the urban and urbanising areas must be strictly enforced according to transparent procedures and zoning regulations within a context of clearly defined boundaries throughout the district. The impact of land titles will be undermined and eroded to the extent that boundaries are ambiguous or contested.

The expected economic impacts of land titles in Siem Reap are similar to the expected impacts in Phnom Penh. Given the nature of the local economy, land titles are likely to have the most significant impact in sectors most closely associated with tourism, housing and transportation. The primary conduit for impacts will be formal credit for housing, services and small business development, which in turn may generate a potentially significant impact on employment. Housing development will include residential property to accommodate the rapid population growth as well as guesthouses and hotels for international and domestic tourists. Services will include both those specifically targeted at tourists and those for local residents, such as schools, water and sanitation. Small business development includes a wide range of commercial activities for tourists, residents and merchants.

As in Phnom Penh, there had been a sharp increase in the number of reported loans during the two and a half years prior to the survey. Almost half of the reported loans were obtained from the formal sector, somewhat more than in Phnom Penh. It is expected that the number of loans obtained from commercial banks and MFIs for productive investments will increase as property owners use their land titles for collateral. Although it is also expected that the average amount of loans will increase, those with more property may be able to obtain larger loans and therefore benefit more than those with less property.

Also as in Phnom Penh, the employment impact of land titles may be significant if titles stimulate increased SME investment and business expansions by promoting a more secure tenure and providing access to more affordable credit. Shifts in attitudes concerning borrowing from commercial banks appear to be relatively recent. As the business climate in Cambodia improves and investors develop more trust in the banking sector, it is likely that the increased demand for credit for a variety of purposes will continue. By creating reliable collateral, it is quite likely that land titling will support an increase in consumer and business borrowing. This may unfold at a somewhat slower pace among borrowers in Siem Reap because traditional attitudes toward financing small business with own and family sources may be stronger in a less diversified economy. It may take more time for business entrepreneurs in Siem Reap to embrace alternative modes of financing, and so the overall employment impact through SMEs may be somewhat less in Siem Reap than in Phnom Penh, at least initially.

Despite the fact that the volume of land transactions has been increasing rapidly, Siem Reap can still be characterised as an emerging land market when compared to the more active and diverse market in and around Phnom Penh. The urbanised commune of Sala Kamraeuk, however, exhibits characteristics suggesting it is a rapidly maturing land market. As in Phnom Penh, land transactions will be the primary mode of land acquisition, and hence of redistribution. The survey data show that people in Siem Reap also rely on a wide range of documentation modes, although the use of certificates was somewhat higher than in Phnom Penh. This may reflect differences in the transaction costs associated with obtaining certificates in the two locations, as the extent to which people will use the official registry is sensitive to such costs, particularly with respect to unofficial fees. The concern expressed in the Phnom Penh section about the use of the official registry to facilitate land transactions is equally valid for Siem Reap. Although land ownership was more frequently documented with certificates by survey households in Siem Reap than in Phnom Penh, there was still widespread use of unofficial papers that are not legally valid. Unless land transfers are facilitated through the official registry, the prospect of land conflicts remains serious. That being said, the number of land conflicts reported by survey households was just as small as in Phnom Penh. It may be that the instrument used in this survey was not the most appropriate tool for understanding how land titles affect land conflicts. Again as in Phnom Penh, another important question concerns the role that the courts and local authorities will play in enforcing land titles in cases of conflict.

As in Phnom Penh, it is expected that women's tenure security will be strengthened. One way to assess the impact of land titles on women's tenure security would be to examine the data concerning female-headed households. Another would be to examine the outcome of cases involving the death of a husband or divorce and evaluate the extent to which the woman's land tenure rights are upheld. Qualitative methods will be required to understand the social and legal dynamics of land tenure security for women.

The data collected in the Siem Reap survey largely conform to expected trends and patterns and reasonably approximate the Phnom Penh data in areas such as land values according to location and consumption quintile, modes of documentation and credit. In other areas there are interesting variations that reflect the different characteristics of the two cities and their respective development trajectories. As in Phnom Penh, the follow-up study to the Siem Reap baseline survey should involve MLMUPC personnel who are familiar with land valuation techniques. The follow-up survey should also employ qualitative research approaches.

Chapter 7. Serei Saophoan Baseline Survey

The Serei Saophoan baseline survey was conducted from 18 to 27 December 2005. The landtitling programme, however, has yet to begin in Banteay Meanchey, except for a pilot effort conducted in 2005 in Svay Chek. The gap between the time of the baseline survey and when titling is likely to get under way (2008) will be too long for the current survey data to be used as a baseline. One possible solution to this problem would be to repeat the survey just prior to land titles being issued. In the meantime, the baseline survey data for Serei Saophoan are still useful because they offer an interesting contrast to the Phnom Penh and Siem Reap data. A second baseline survey would also help to create a set of panel data with which to analyse the development of emerging land markets in and around Serei Saophoan, which appears to be an increasingly dynamic area affected by the transport industry.

7.1. Household Characteristics

A total of 507 households were interviewed in Serei Saophoan. Of these, 18.7 percent were headed by women, less than the 22.7 percent in Phnom Penh and 20.7 percent in Siem Reap. Two hundred and ninety households were surveyed in Kompong Svay commune, including 17.9 percent headed by women. One hundred and eight households were interviewed in Preah Ponlea commune, only 11.1 percent headed by women. One hundred and nine households were surveyed in Ou Ambel commune, 28.4 percent headed by women.

As in the other surveys, respondents were asked to report on various household expenditures over different periods, and this information was aggregated to estimate a per capita annual consumption figure. Table 7.1 shows the distribution of the survey households in the three communes according to per capita consumption and sex of household head.

Communo	Quin	tile 1	Quin	tile 2	Quin	tile 3	Quin	tile 4	Quin	tile 5		Total	
Commune	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Tot
Kg. Svay	53	5	47	17	51	9	43	10	44	11	238	52	290
Preah Ponlea	13	0	15	3	19	4	26	3	23	2	96	12	108
Ou Ambel	23	7	13	7	7	11	15	5	16	5	78	31	109
Total M/F	89	12	75	27	81	20	84	18	83	18	412	95	
Total Quintile	1()1	10)2	1()1	10)2	1(01			507
% of Quintile	88.1	11.9	73.5	26.5	80.2	19.8	82.4	17.6	82.2	17.8	81.3	18.7	

Table 7.1: Households Surveyed, by Consumption Quintile and Sex of Household Head

Table 7.2 shows the average per capita consumption expenditures for the year prior to the survey. The aggregate average as well as the averages for both male- and female-headed household are much lower than in Siem Reap. The aggregate average per capita consumption for female-headed households is lower than male-headed households, by about 5.5 percent.

Quintile	Male	e HH	Fema	le HH	No. HH	Mean	Range
	No.	Mean	No.	Mean			
1	89	237	12	225	101	236	120-311
2	75	392	27	374	102	387	312–463
3	81	551	20	564	101	554	464 - 636
4	84	788	18	808	102	792	639–984
5	83	1857	18	1685	101	1,826	985–18,450
Total	412	766	95	726	507	758	120-18,450

Table 7.2: Average Consumption, by Quintiles and Sex of Head (USD/year/person)

The average annual per capita consumption expenditure in Preah Ponlea was USD870, in Kompong Svay USD737 and in Ou Ambel USD704. Ou Ambel was almost exactly the same as Siem Reap commune in Siem Reap district. Kompong Svay also closely approximated Siem Reap commune, while Preah Ponlea was closer to Sala Kamraeuk. All three communes in Serei Saophoan more closely resembled the peri-urban district averages in Phnom Penh than the averages in the urban districts.

7.2. Plot/Unit Characteristics

This section provides some of the key data concerning plot/unit characteristics covered in the survey sample. As in the same sections concerning Phnom Penh and Siem Reap, the data presented are the mode of acquisition, the type of documentation to claim or validate ownership, the location of and services available to the plot and plot values. Two different data sets are used. The most relevant set concerns Plot/Unit 1, the plot or unit where the owner resides or works and where the interview took place. The total number of Plot/Unit 1 in Serei Saophoan was 507, equal to the number of survey households.

The second data set concerns all the plots owned by the 507 respondents. The total number of "all plots" represented in the data from Serei Saophoan is 905, about 1.79 plots per survey household. This closely approximates the 1.70 plots per household found in Phnom Penh, but is somewhat lower than the 2.07 per household in Siem Reap. About 73.3 percent of "all plots" were located in the same village, while another 6.0 percent were located elsewhere in the same district. About 11.3 percent were within Banteay Meanchey province but outside the district. The remaining 9.4 percent were located outside the province.

7.2.1. Mode of Acquisition

Table 7.3 shows that a majority (53.1 percent) of Plots/Units 1 were acquired through purchase, followed by state allocation and inheritance. This pattern more closely resembled that of Phnom Penh than Siem Reap. In Phnom Penh, 50.5 percent of Plot/Unit 1 acquisitions were through purchase, followed by state allocation (35.4 percent) and inheritance (12.9 percent), while in Siem Reap the modes of acquisition were fairly evenly divided.

Commune	Given	by State	Inhe	rited	Purcl	nased	Ot	her	Total
	No.	%	No.	%	No.	%	No.	%	
Kg. Svay	87	30.0	29	10.0	173	60.0	1	0.3	290
Preah Ponlea	40	38.5	12	37.0	54	50.0	2	0.3	108
Ou Ambel	45	41.3	22	20.2	42	38.5	0	0	109
Total	172	33.9	63	12.4	269	53.1	3	0.6	507

 Table 7.3: Mode of Acquisition, by Commune (Plot/Unit No. 1)

Table 7.4 shows the mode of acquisition by sex of household head. Of the 95 Plots/Units 1 owned by female heads of household, just over 50 percent were acquired through purchase, followed by state allocation (40 percent) and inheritance (8.4 percent). This differed from

Siem Reap, where a smaller percentage (30.7) of the female-headed households acquired their plot through purchase, while a greater percentage (49.1) acquired it through state allocation. A greater percentage (17.9) also acquired their plot through inheritance. As for male-headed households, a greater percentage (53.6) also purchased their land in Serei Saophoan than in Siem Reap (33.4 percent). However, a smaller percentage (35.5) received plots through state allocation than in Siem Reap (42.6 percent). A smaller percentage (13.3) also received plots from inheritance than in Siem Reap (23.6 percent).

Commune	Given b	by State	Inhe	erited	Purc	hased	Otl	ner		Total	
	Μ	F	Μ	F	Μ	F	Μ	F	Μ	F	Tot
Kg. Svay	72	15	26	3	139	34	1	0	238	52	290
Preah Ponlea	34	6	12	0	49	5	1	1	96	12	108
Ou Ambel	28	17	17	5	33	9	0	0	78	31	109
Total	134	38	55	8	221	48	2	1	412	95	507
% of Gender	35.5	40.0	13.3	8.4	53.6	50.5					
Total	1'	72	6	53	2	69	3	3	50)7	
% of Total	33	3.9	11	2.4	53	3.1	0.	6			

 Table 7.4: Mode of Acquisition, by Gender of HH Head (Plot/Unit 1)

Table 7.5 shows that the majority (55.2 percent) of all plots in the Serei Saophoan survey were acquired through purchase. This was similar to Phnom Penh, where 52.3 percent were purchased, and much higher than in Siem Reap, where the figure was 32.9 percent. The pattern of acquisition for all plots closely approximated the pattern of Plots/Units 1. These patterns suggest that despite many rural characteristics, the emerging land market in Serei Saophoan is taking on at least some urban characteristics.

Commune	State	Inherit	Purchase	Clearing	Other	Total
Kg. Svay	138	68	289	10	2	507
Pr. Ponlea	52	22	109	3	0	186
Ou Ambel	64	34	102	11	1	212
Total	254	124	500	24	3	905
% of Total	28.1	13.7	55.2	2.7	0.3	100

Table 7.5: Mode of Acquisition, by Commune (All Plots/Units)

7.2.2. Documentation

Table 7.6 shows that households in Serei Saophoan, like those in Phnom Penh and Siem Reap, used a wide range of means to document landownership. The households in Serei Saophoan tended to rely on documents that might be officially endorsed, but are not legal in terms of securing tenure rights. For Plot/Unit 1, the percentage of survey properties documented with official certificates was highest in Serei Saophoan at 22.5 percent, (Phnom Penh 5.4 percent and Siem Reap 20.3 percent). The use of certificate application receipts in Serei Saophoan (38.1 percent) was similar to that of Siem Reap (35.1 percent). Survey papers were hardly used at all by the Serei Saophoan survey households, while 17.9 of properties in Siem Reap wee documented with such paper. A higher percentage of properties (23.3 percent) were documented with transfer documents at either the village or commune levels in Serei Saophoan than in Siem Reap (13.3 percent). As in Phnom Penh and Siem Reap, a surprising percentage of households in Serei Saophoan, about 10.5 percent, had either lost the documentation or never had any.

Commune	App.	Survey	Certificate	Sales	ComVil	No	Total
	Receipt	Paper		Letter	Transfer	Paper	
Kg. Svay	106	3	55	10	81	35	290
Preah Ponlea	56	1	13	5	17	16	108
Ou Ambel	31	0	46	0	20	12	109
Total N	193	4	114	15	118	63	507
% of Total	38.1	0.7	22.5	3.0	23.3	10.5	

 Table 7.6: Type of Land Documentation, by Location (Plot/Unit 1)

App. Receipt = Application Receipt; ComVil = Commune or Village

It is interesting to observe the differences in documentation between Plot/Unit 1 and all plots, the latter being shown in Table 7.7. In the case of Plot/Unit 1, the use of certificates and village or commune transfer papers was practically the same. In the case of all plots, village or commune transfer papers are used almost twice as often as certificates. This suggests that households in Serei Saophoan were more inclined to obtain certificates for their primary plot or unit, while relying on less secure but locally accepted documentation for other plots/units. These other plots/units could be less valued land, including agricultural or *chamkar* land outside the town. Approximately the same percentage of properties were documented with application receipts in both data sets, about 38 percent of Plot/Unit 1 and 34 percent of all plots. There is a higher percentage of all plots than Plot/Unit 1 with no paper.

Commune	App.	Survey	Certificate	Sales	ComVil	No	Total
	Receipt	Paper		Letter	Transfer	Paper	
Kg. Svay	160	3	71	21	155	97	507
Preah Ponlea	87	1	16	6	47	29	186
Ou Ambel	60	0	53	4	60	35	212
Total N	307	4	140	31	262	161	905
% of Total	33.9	0.4	15.5	3.4	29.0	17.8	

Table 7.7: Type of Land Documentation, by Location (All Plots)

App. Receipt = *Application Receipt; ComVil* = *Commune or Village*

7.2.3. Location and Services

Table 7.8 shows the location of Plot/Unit 1 in all three communes according to access. Eighty-nine percent of the survey plots were located on main paved roads or 2-way access streets, while the remaining 10.9 percent were in less accessible locations. This pattern contrasts with Siem Reap, where 54.9 of Plots/Units 1 were in less accessible locations. While this may reflect to some degree a selection bias on the part of field enumerators, it is probably more reflective of the general character of the urban design in Serei Saophoan, where main paved roads (residential and commercial property) and 2-way access streets (mostly residential) are arranged on a rectangular grid. As mentioned above, in Siem Reap the road grid has yet to be completed.

rusie not rype			100 0 1110 1)			
Commune	Main Paved	2-way	1-way	Bike Access	Foot	Total
	Road	Access*	Access*		Access	
Kg. Svay	78	182	20	8	2	290
Preah Ponlea	54	34	17	3		108
Ou Ambel	22	81	1	5		109
Total	154	297	38	16	2	507
% of Total	30.4	58.6	75	3.2	04	

 Table 7.8: Type of Access, by Commune (Plot/Unit 1)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

Commune	Main Paved	2-way	1-way	Bike Access	Foot	Total
	Road	Access*	Access*		Access	
Kg. Svay	25	33	0	0	0	58
Preah Ponlea	1	1	0	0	0	2
Ou Ambel	2	5	0	0	0	7
Total	28	39	0	0	0	67
% of Total	41.8	58.2				

Table 7.9: Households with Electricity Service, by Type of Access (Plot/Unit 1)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

The situation regarding sewage service was similar to that of electricity. Table 7.10 shows that 83 properties, about 16.4 percent, had sewage service. As with electricity, most properties with sewage service (75.9 percent) were located in Kompong Svay. Also, all properties except two with sewage services were located along main paved roads (69) or 2-way access streets (12). This pattern closely resembled Siem Reap, where none of the survey households in the peri-urban commune of Siem Reap reported having sewage service. Except for two properties, those few households (52, or 9.7 percent of the total) in Sala Kamraeuk that had sewage service were all located on main paved roads (31) or 2-way access streets (19).

Commune	Main Paved	2-way	1-way	Bike Access	Foot	Total
	Road	Access*	Access		Access	
Kg. Svay	53	10	0	0	0	63
Preah Ponlea	14	0	2	0	0	16
Ou Ambel	2	2	0	0	0	4
Total	69	12	2	0	0	83
% of Total	83.1	14.5	2.4			

 Table 7.10: Households with Sewage Service, by Type of Access, (Plot/Unit 1)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

7.2.4. Land Use and Values

7.2.4.1. Land Values

Respondents were asked how much they would receive if they were to sell Plot/Unit 1 at the time of interview. Although some respondents may have over- or underestimated their plot values, the data concerning reported plot values conform to expected trends. Table 7.11 shows that the average reported value of a Plot/Unit 1 in Serei Saophoan was USD15,469, considerably lower than the averages of USD40,412 in Phnom Penh and USD53,365 in Siem Reap. It was also lower than in the peri-urban commune of Siem Reap, where the average was USD23,420 per plot, and lower than the reported average value of USD23,120 for the three peri-urban districts of Phnom Penh. Land values in and around Serei Saophoan seem to resemble rural areas more closely than urban or peri-urban areas.

The reported plot values, however, shared similarities with both Phnom Penh and Siem Reap. First, the average steadily increased with improved access in all three communes. As in Phnom Penh and Siem Reap, access appears to have a significant impact on plot values.

CDRI

Although main paved roads and 2-way access streets both represent 2-way access, the reported average value of plots located along main paved roads was significantly more than the average reported value of plots along 2-way access streets in all three communes. In fact, in Kompong Svay and Ou Ambel, the reported value of houses on main paved roads was more than twice that on 2-way access streets.

Commune	Main Paved	2-way	1-way	Bicycle	Foot]	Fotal	
	Road	Access*	Access*	-	Access			
						No.	Ave \$	
Kg. Svay	23,938	9426	3413	2500	8750	290	12,719	
Preah Ponlea	31,972	18,162	9000	7333		108	23,324	
Ou Ambel	30,682	11,275	4000	8610		109	15,003	
Average	27,719	10,931	5928	5316	8750	15,469		
Total HH	154	297	38	16	2		507	

 Table 7.11: Average Land Values, by Type of Access (USD/plot)

* For a vehicle (Please see Survey Questionnaire, Question 2.1.

Second, the data also conform closely to expectations when average plot values are considered according to consumption quintiles. As Table 7.12 shows, the average value of the plot steadily increased with the consumption quintile in all three communes. The average plot values ranged from USD9,008 to USD27,146 between the first and fifth consumption quintiles, while the range between the middle three quintiles was of course less, USD10,382 to USD17,860.

Commune	Quint 1	Quint 2	Quint 3	Quint 4	Quint 5	Total	
						HH	Ave \$
Kg. Svay	8156	8785	12,078	14,647	20,949	290	12,719
Preah Ponlea	10,212	14,792	16,478	25,448	40,120	108	23,324
Ou Ambel	9,008	11,522	13,372	15,370	27,931	109	15,003
Average	9008	10,382	13,311	17,860	27,146	15,469	
Total HH	101	102	101	102	101	507	

Table 7.12: Reported Plot/Unit 1 Values (USD), by Consumption Quintile

The gap between quintiles one and five was USD18,138 in Serei Saophoan, while the gap in Siem Reap was USD66,420 and in Phnom Penh USD50,481. The narrower gap suggests again that Serei Saophoan still retains certain rural characteristics.

7.2.5. Buildings and Housing

As in Phnom Penh and Siem Reap, data concerning the number, use and quality of buildings on Plot/Unit 1 were also collected. This information will be important for assessing potential land-titling impacts according to the hypothesis that people will take advantage of more secure tenure to invest in housing and building improvements, and may use titles as collateral for credit for making such improvements. Table 7.13 shows that most plots, about 65.9 percent, had one building.

Quintile	1	2	3	4	Total
1	53	39	0	1	93
2	66	31	3	2	102
3	69	24	8	1	102
4	75	20	6	2	103
5	70	24	5	3	102
Total	333	138	22	9	505*
% of Total	65.9	27.3	6.1	1.7	

Table 7.13: No. of Buildings Per Plot/Unit 1, by Consumption Quintile

* 2 missing

As in Phnom Penh and Siem Reap, information was collected on the building materials used for roofing, walls, floors and fencing. Since roofing is generally considered a good indicator of household consumption level and well-being, one would expect to find a higher proportion of low-cost roofing among the lower quintile groups. Over time, the overall quality of roofing would be expected to change, with houses in each quintile upgrading, although upper quintile households might remain static if they were already using better materials. The data concerning roofing material are presented in Table 7.14 and seem to conform to expectations, because the percentage of houses using zinc sheets for roofing does decrease as consumption increases. This is unlike Phnom Penh and Siem Reap, where the patterns were puzzling. This may represent another instance in which Serei Saophoan more closely resembles a rural area and, as a result, here roofing may be a useful indicator of household consumption levels and of possible land-titling impacts.

Quintile	Thatch/Tent	Zinc sheets	Tile	Wood/plywood	Concrete	Total
1	1	99	1	0	0	101
2	0	92	7	1	1	101
3	0	91	6	2	2	101
4	3	84	8	3	4	102
5	2	79	8	1	9	99
Total	6	445	30	7	16	504*

Table 7.14: Roofing Type, by Consumption Quintile

* 3 missing

7.3. Land Transactions

7.3.1. Purchases

The 507 survey households reported purchasing a total of 501 plots since 1980, almost one plot per household. Of this number, 152, about 30.3 percent, were purchased between 2000 and 2005. Over half of these purchases, 58.6 percent, took place in the latter half of this period, 2003–05. Table 7.15 shows the average reported purchase price of all plots/units acquired through purchase since 2000. There does not appear to be any pattern in the number of purchases or the average value of purchase by year. The average reported purchase value was considerably less than the overall average for either Phnom Penh or Siem Reap, as would be expected, and reinforces the impression of Serei Saophoan as an emerging land market.

Commune	2000	2001	2002	2003	2004	2005	Total
Kg. Svay	2725	1179	7925	1948	5024	3335	3360
Pr. Ponlea	2634	2585	9275	781	2273	1943	2918
Ou Ambel	479	9520	2868	1013	2637	1233	2830
Average	2158	3887	6366	1649	3734	2676	3142
Number	29	18	16	25	29	35	152

Table 7.15: Average Plot/Unit Purchase Prices (USD), 2000–05, by Quintile (All Plots)

Of the 501 reported land purchases, 103 took place during 1980–89, including 20 and 21 purchases in 1985 and 1989, respectively. As with Phnom Penh and Siem Reap, the data concerning the year of purchase may contain some reporting errors, but it appears there was a nascent land market in this area even before 1989.

The 507 survey households reported acquiring 269 Plots/Units 1 through purchase, 53.1 percent of the total. Of this number, only 49 (18.2 percent) were purchased during d 2000–05. Fewer than half of these purchases were made during the second half of this period,

suggesting that the land market in Serei Saophoan is rather lethargic compared to Phnom Penh and Siem Reap. One possible explanation for most plots being purchased prior to 2000 could be a large influx of domestic migrants who purchased land in and around Serei Saophoan, perhaps including refugees returning from the camps along the Thai-Cambodian border in the early 1990s.

Table 7.16: Average Price of Plot Purchases (2000–05), by Commune and Year (USD per Plot/Unit 1)

Commune	2000	2001	2002	2003	2004	2005	Total
Kg. Svay	6738	11,875	11,375	9730	12,750	10,192	9791
Pr. Ponlea	10,750	31,000	8000	6333	9000	2500	11,536
Ou Ambel	4500	60,000	7250		4000		14,583
Average	7564	29,150	9714	8456	10,250	9093	10,877
Total	14	5	7	8	8	7	49
% of Total	28.6	10.2	14.3	16.3	16.3	14.3	

Table 7.17 also does not show any discernible pattern in terms of the number of plot purchases across consumption quintiles during 2000–05. However, the average purchase price of Plot/Unit 1 increased along with consumption quintile, which conforms to expectations and is similar to the patterns found in Phnom Penh and Siem Reap, albeit on a significantly smaller scale.

Table 7.17: Average Price of Plot Purchases (2000–05), by Commune and Quintile (USD per Plot/Unit 1)

Commune	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Total
Kg. Svay	3850	3460	8607	16,000	14,111	9791
Pr. Ponlea	5250	8000	2500	30,000	21,750	11,536
Ou Ambel	3500	4500		8000	60,000	14,583
Total Ave.	4340	4725	6775	15,714	19,571	10,877
Number	10	8	10	7	14	49
% of Total	20.4	16.3	20.4	14.3	28.6	

7.3.2. Sales

Of the 507 survey households, 99 (19.5 percent) reported selling a total of 118 plots since 1980, an average of 1.19 plots per selling household. Well over half (64.6 percent) of the sales occurred in Kompong Svay. Between 2000 and 2005, 51 households reported selling one plot/unit each. Table 7.18 shows that of the 51 sales, just under half (49.0 percent) took place in Kompong Svay, while 31.4 percent took place in Preah Ponlea. As in Siem Reap, the percentage of land sales by year does not show a consistent pattern in the number of sales or the average reported value of the sales.

Commune	2000	2001	2002	2003	2004	2005	Т	Total	
							N	Ave.	
Kg. Svay	4223	6000	2800	2817	2917	10,1227	25	4529	
Pr. Ponlea	4315	1200	375	11750	455	9250	16	2878	
Ou Ambel	746	20,250	4000	1250	3375		10	5924	
Total Ave.	3759	11,925	2494	2117	2040	9835		4285	
Total No.	14	4	8	6	13	6	51		
% of Total	27.5	7.8	15.6	11.8	25.5	11.8	100		

Table 7.18: Average Price of Land Sales, by Commune and Year (2000–05) (USD/Plot)

Table 7.19 shows the average reported values for all plot sales between 2000 and 2005 in each commune. As with the year of land sales, the data for consumption quintiles do not reveal a clear pattern. One possible explanation may have to do with where the plots are located. The data above concerning the commune refer to the location of the selling household, not the location of the sold plot. The sample size of 51 is also quite small.

· · · ·						
Commune	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Total
Kg. Svay	3596	3907	1588	7958	4250	4529
Pr. Ponlea	629	6625	581	6675		2878
Ou Ambel	1000	2746	1750	40,000	2313	5924
Total Ave.	2236	4219	1177	10,405	2958	4285
Total No.	15	10	9	11	6	51

Table 7.19: Average Price of Land Sales, by Commune and Quintile (2000-05)(USD/Plot)

7.3.3. Documentation

The data showed a strong preference for documenting land sales by changing the name of the owner at the village (48) or commune (44) level, 78 percent of all sales. Only 5.9 percent changed the name at the district level, while even fewer (3.4 percent) registered the transfer with the official provincial registry. Another 12.7 percent of plots were transferred using a contract or letter between the buyer and seller. As in Phnom Penh and Siem Reap, the preference for transferring ownership locally may be due to the high transaction costs of unofficial fees and sales taxes associated with legal transfers. As in Siem Reap, location cannot be a factor, because Serei Saophoan is both the provincial headquarters and the district headquarters.

7.4. Conflicts

As in the other surveys, households were asked, "Have you ever had conflicts over land or other property since the national election of 2003?" The number of affirmative answers (eight) was even fewer than in Phnom Penh (36) and Siem Reap (20). This represented only a very small fraction of all 905 plots in the survey. In fact, the number of cases was so small that it is not possible to identify any trends one way or another. This could bode well for LMAP's future titling efforts in the area because the adjudication phase may be less impeded by conflicts than in some other areas. The relevant question is why so few conflicts were reported. One possible explanation might be that real estate values have yet to reach the point where competing owners see it as worthwhile to contest boundaries or ownership. If this is the case, more conflicts may emerge as land values increase.

7.5. Credit and Finance

A total of 113 households, 22.3 percent of the survey, reported taking 145 loans, 1.28 loans per borrowing household, since the national election in 2003. Although the percentage of households taking loans was similar to Siem Reap (25.2 percent), the average number of loans per borrowing household was somewhat less, since Siem Reap reported an average of 1.44. The number of loans decreased steadily as the consumption quintile rose. As in Siem Reap, there was a sharp increase in the number of loans during 2003–05. In 2003, there were 19 loans reported, while in 2004 there were 37; this number more than doubled to 87 in 2005.

The sources of loans in the Serei Saophoan survey were considerably different from those in Phnom Penh and Siem Reap. Some 19.3 percent of all loans were obtained from relatives (15.9 percent) and friends (3.4 percent), fewer than the 28.7 percent in Siem Reap. Another 38.6 percent were obtained from formal sector institutions, including MFIs (10.3 percent) and

Acleda (28.3 percent), fewer than the 49.7 percent in Siem Reap. Moneylenders provided 37.2 percent of the loans. This figure is much higher than the 19 percent found in Siem Reap, and suggests that in terms of credit, Serei Saophoan again more closely resembles a rural area than it does other urban areas.

Table 7.20 shows loan uses across consumption quintiles. More than 34 percent of all loans were used for business purposes. This closely approximated the findings in both Phnom Penh (35.8 percent) and Siem Reap (38.3 percent). Real estate, home improvements and housing construction accounted for another 21.0 percent of loans, slightly more than in Phnom Penh (20.0 percent) but less than in Siem Reap (25.1 percent). Food, health and education accounted for 17.5 percent of all loans, compared to 22.8 percent in Phnom Penh and 13.8 percent in Siem Reap. Farming and animal raising were 11.9 percent, more than in both Siem Reap and Phnom Penh, which suggests again that Serei Saophoan still retains attributes of a rural environment. Transportation accounted for 10.5 percent, which closely approximated Siem Reap.

Credit Use	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	To	tal
						No	%
Farming/animals	9	4	2	1	1	17	11.9
Bus. Start-up	2	2	8	3	0	15	10.5
Bus. Expansion	3	4	3	1	5	16	11.2
Bus. Input	6	5	1	3	3	18	12.6
Food	5	1	2	1	1	10	7.0
Health	5	3	1	2	0	11	7.7
Education	0	2	0	2	0	4	2.8
Real Estate	4	0	3	2	1	10	7.0
Loan Repay	0	0	1	2	0	3	2.1
Home Improve	1	1	1	2	0	5	3.5
Home Constr.	4	2	3	3	3	15	10.5
Transport	4	4	5	2	0	15	10.5
Other	1	1	2	0	0	4	2.8
Total	44	29	32	24	14	143*	
% of Total	30.8	20.3	22.4	16.8	9.8		

 Table 7.20: Credit Use, by Consumption Category

* 2 missing

As in Phnom Penh and Siem Reap, the volume of borrowing from commercial sources is expected to increase because land titles will enable more credit seekers to meet collateral requirements. The shares of loans for business-related activities and real estate and housing will increase, while other uses will decrease; land use patterns will shift in the direction of commerce and residences. The average amount of loans using land titles as collateral will probably increase, albeit at a somewhat slower pace than in Phnom Penh and Siem Reap, as increasing land values increase the size of loans for which potential borrowers can qualify. Upper quintile households may benefit more than lower ones because their property used as collateral tends to be valued more highly. The gap in benefits between upper and lower quintile households may be smaller than in either Phnom Penh and Siem Reap because the gap in property values is smaller.

7.6. Small and Medium Enterprises

In the survey, 241 households reported operating a total of 285 SMEs, which means that 47.5 percent of the households operate an average of 1.18 SMEs each. The percentage of households operating SMEs was far higher than in Siem Reap, where the figure was 30.6 percent. The average number of SMEs per household, however, was almost identical to the 1.17 in Siem Reap.

More than half (55.8 percent) of SME operators were female heads, between Siem Reap (49.2 percent) and Phnom Penh (60.7). Some 52.6 of the SMEs were operated by households in Kompong Svay commune; households in Preah Ponlea operated 29.1 percent; households in Ou Ambel operated 18.2 percent.

Table 7.21 shows that 56.1 percent (160) of the SMEs had been started since 2000, and of this number more than half (59.4 percent) were started during 2003–05, very similar to the trend found in Siem Reap. Households in the upper two quintiles operated 43.7 percent of the businesses, while the lower two quintiles operated 31.9 percent. Table 7.21 also shows that the number of start-ups in 2001 was sharply down, while the number doubled from 2004 to 2005.

Year	Quintile 1	Quintile 2	Quintile 3	Quintile 4	Quintile 5	Тс	otal
						Tot	%
2000	2	6	13	8	5	34	21.3
2001	1	0	3	1	4	9	5.6
2002	5	5	4	6	2	22	13.8
2003	4	4	7	9	5	29	18.1
2004	2	4	2	9	5	22	13.8
2005	8	10	10	9	7	44	27.5
Total	22	29	39	42	28	160	
% of Total	13.8	18.1	24.4	26.2	17.5		100

 Table 7.21: SME Start-ups, by Year and Quintile (2000–05)

7.6.1. SME Financing

Table 7.22 identifies 301 sources of SME financing reported by the Serei Saophoan group. Eighty-five percent of the SME financing came from own sources and savings, while family accounted for 8.6 percent. These figures are almost identical to Siem Reap, where own resources and savings and family accounted for 86.1 percent and 8.2 percent, respectively, of the sources. As in the other centres, non-formal moneylenders and formal commercial and MFI lenders appeared to play almost no role in financing SMEs in Serei Saophoan.

As for the use of SME financing sources, more than half (54.0 percent) were used to start a new business, 27.3 percent to operate the business and the remaining 18.7 percent to expand the business. Of the financing from own sources and savings, in 58.3 percent of cases the funds were used to start a new business, in 29.1 percent to operate the business and in 12.2 percent to expand.²⁷

		e,	V	•	
Quintile	Own	Family	Private Bank	Other	Total
	Resources		MFI		
1	27	1	2	1	31
2	43	3	1	3	50
3	64	5	1	5	75
4	66	8	1	4	79
5	56	9	1	0	66
Total	256	26	7	12	301
% of Total	85.0	8.6	2.3	4.1	

 Table 7.22: Small Business Financing, Sources by Consumption Quintile

²⁷ The remaining 0.4 percent of usage is not mentioned.

7.6.2. SME Employment

As in Phnom Penh and Siem Reap, the survey data suggest that SMEs are a potentially important source of employment in Serei Saophoan. Table 7.23 shows that the number of employees was higher in the upper quintile businesses than in the lower. Again, the total number of employees in each labour category had increased over time. The total number of employees at the time of start-up for the 285 SMEs was 566, an average of 1.99 per SME, fewer than in Siem Reap and Phnom Penh. The number of employees at the time of the interview was 649, an average of about 2.28 per SME, also lower than in the other two cities.

Quintile	No.	Permanent		Daily Labour		Family Members		Total	
		Start	Current	Start	Start Current St		Current	Start	Current
1	28	0	0	0	0	47	50	47	50
2	43	1	5	12	1	80	88	93	94
3	60	8	16	16	17	127	145	151	178
4	58	1	9	4	21	117	125	122	155
5	52	43	62	4	7	106	103	153	172
Total	241	53	92	36	46	477	511	566	649

Table 7.23: Type of SME Employment, by Consumption Quintile

7.7. Expected Benefits of Land Titles

Table 7.24 shows that 80.9 percent of survey households in Serei Saophoan believed tenure security to be the primary benefit from land titles. Adding in the 5.7 percent who expected fewer land conflicts gives a total of 86.6 percent of households that expected the primary benefit to be improved tenure security. This was very similar to both Phnom Penh and Siem Reap. Another 10.8 percent identified improved access to formal credit as the primary benefit, exactly the same as in Siem Reap and slightly more than in Phnom Penh. Another 2.2 percent of households mentioned facilitating transfers, fewer than the 4.3 percent in Siem Reap.

Quintile	Secure	End	Obtain	Facilitate	Facilitate	Total
	Tenure	Conflict	Loan	Transfer	Sale	
1	81	6	12	1	1	101
2	90	5	4	2	1	102
3	76	8	13	1	1	99
4	81	5	13	1	2	102
5	82	5	13	1	0	101
Total	410	29	55	6	5	505*
% of Total	80.9	5.7	10.8	1.2	1.0	

Table 7.24: Expected Primary Benefits of Land-Titling, by Quintile

* 2 = "Nothing different"

The importance of land tenure security as an expected benefit was again highlighted by the responses concerning expected secondary benefits. Some 39.3 percent of the Serei Saophoan survey households identified ending conflict as the second most important expected benefit, while another 12.6 percent identified "stability in owning the land", making a total of 51.9 percent of responses that concerned land tenure security. This was less than the 58.4 percent found in Siem Reap, but closer to the 54.0 percent found in Phnom Penh. Another 27.2 percent identified "ease in borrowing money" as the second most important benefit. In Siem Reap, the percentage was 22.4 percent and in Phnom Penh 20.3 percent. Another 11 percent cited "ease in transferring inheritance" and 7.7 percent "ease in selling", a total 18.7 percent of responses concerning improved facilities for transfers. The remaining 1.2 percent identified "no idle land" as the second most important benefit.

CDRI

The potential impact of land-titling in Serei Saophoan must be considered in the context of its strategic location on key transport networks in Cambodia's north-west. This suggests that Serei Saophoan will likely develop into an important transportation and, perhaps, commercial hub in the medium to long term. The surrounding area is fertile farmland, and Serei Saophoan could also eventually emerge as an important ago-processing zone. Its commercial and residential areas are structured on a rectangular grid with wide, largely unpaved, streets that should facilitate the development of urban infrastructure and services. It is likely that the city's commercial sector will initially expand in three directions along the major national routes, while residential areas will develop off the major highways.

As in Phnom Penh and Siem Reap, land titles could have a significant if indirect impact on land use patterns to the extent that they facilitate more efficient transactions and provide larger and more predictable returns on investments. Land in and around Serei Saophoan must be managed according to transparent procedures and enforceable zoning regulations in the context of clearly defined boundaries between state public, state private and private land. The impact of land titles will be undermined and eroded to the extent that such boundaries are ambiguous or contested. Serei Saophoan does not have a city development or land use master plan.

The expected economic impacts of land titles in Serei Saophoan are similar to those in Phnom Penh and Siem Reap, although on a smaller scale. As the economy of Serei Saophoan is likely to be increasingly based on transport, trade and light manufacturing, land titles are likely to have the most significant impact in those sectors most closely associated with services and SMEs. The primary conduit for land-titling impacts will be formal credit for services and business development, which in turn may generate a potentially significant impact on employment. Services will include activities that support a growing transport sector, including vehicle maintenance and repair facilities, petrol service stations and road and rail construction and maintenance. Small businesses will include restaurants, trade, light manufacturing and perhaps limited food processing. The housing market could also begin to pick up if an expanding and diversifying economy in the town increases household income and attracts domestic migrants.

As in Phnom Penh and Siem Reap, the employment impact may depend on the degree to which land titles actually stimulate increased SME investment and business expansions. Borrowing from the commercial sector, however, has not yet reached similar proportions to that in Siem Reap, suggesting that people will turn to commercial banks for financing businesses and home improvements more slowly, because traditional attitudes toward finance may be even stronger in Serei Saophoan than in Siem Reap. Borrowing from moneylenders continues to be very high.

The number of recent land transactions is somewhat lower than in Siem Reap, suggesting that land markets are only now emerging as an important feature of the local economy. It is likely, however, that as road and rail transport improves, stimulating increased trade between Thailand and the rest of Cambodia, real estate markets will expand more robustly. Land transactions therefore will eventually be the primary mode of land acquisition and redistribution also in Serei Saophoan. As elsewhere, households in Serei Saophoan rely on a wide range of documentation to claim or validate land ownership, most of which are of an "official but not legal" variety. This raises similar concerns as in Phnom Penh and Siem Reap about the use of the official registry to facilitate transactions. Unless land transfers are facilitated through the official registry, the prospect of land conflicts remains high, especially in an emerging land market like this.

Again, land titles are expected to strengthen women's land tenure security. One way to assess the impact on women's land tenure security will be to examine the data concerning female-

91

headed households. Another way will be to examine the outcome of cases involving the death of a husband or divorce and the extent to which the woman's land tenure rights are upheld. Qualitative methods will be required to understand land tenure security for women.

As in the other cities, the data from Serei Saophoan largely conform to expected trends and patterns and reasonably approximate the Siem Reap data in areas such as land values varying with urban or peri-urban location and consumption quintile, modes of documentation and credit. In other areas, there are differences that reflect the different characteristics of the three cities and their development.

Unfortunately, the 2005 survey data cannot be used as a baseline with which to assess the impact of the land titles because land titling in Serei Saophoan has not yet begun. The time lag between the 2005 survey and the issuance of titles will be too great; too many other intervening variables and factors will cloud or distort the impact measurements. It is recommended that a new baseline survey be undertaken in 2008, prior to the issuing of titles in the three survey communes. The 2005 data remain extremely useful, providing an initial set of panel data that can be used to analyse the emerging land markets in the area. As in the two other urban areas, the follow-up study should involve MLMUPC personnel who are familiar with land valuation techniques. The follow-up should also employ qualitative research approaches.

References

- Analyzing Development Issues (2007), *Land Titling and Poverty Reduction: A Study of Two* Sangkat *in Prey Nob District, Sihanoukville Municipality* (Phnom Penh; Cambodia Cooperation Committee in collaboration with the Land Information Centre)
- Ballard, Brett (2004); "Short-term Trend or Long-term Shift: Institutional Credit in Rural Cambodia", *Cambodia Development Review*, Vol. 8, No.4 October-December (Phnom Penh: Cambodia Development Resource Institute)
- Ballard, Brett (2005), "Linking Tourism to Poverty Reduction: A Siem Reap Case Study" in *Annual Development Review 2004-05*, eds. K.A.S. Murshid and Brett M. Ballard, Phnom Penh: CDRI.
- Ballard, Brett & So Sovannarith (2004), "Cambodia Land Titling Program: Baseline Survey Project: Rural Phase 1 Report" (Phnom Penh: Cambodia Development Resource Institute)
- Ballard, Brett & Phim Runsinarith (2007); "Urban Land Titles, Formal Credit, and SME Investment", *Cambodia Development Review*, Vol. 11, Issue 3, July-September (Phnom Penh: CDRI)
- Barzel, Yoram (1997), *Economic Analysis of Property Rights* (New York: Cambridge University Press)
- Brandao, A.S.P. & Gershon Feder (1996), "Regulatory Policies and Reform: The Case of Land Markets", PSD Occasional Paper No. 15 (Washington, DC: World Bank)
- Chan Sophal & Sarthi Acharya (2002), Land Transactions in Cambodia: An Analysis of Transfers and Transaction Records, Working Paper No. 22 (Phnom Penh: CDRI)
- Deutsch, Robert (2006), Beneficiary Assessment of Land Title Recipients under the Land Management and Administration Project (LMAP) (Phnom Penh: MLMUPC and GTZ)
- Deninger, Klaus (2003), *Land Policies for Growth and Poverty Reduction*, New York: The World Bank and Oxford University press.
- Hing Vutha and Tuot Sokphally (2007), "Pro-Poor Tourism: Siem Reap Case Study", in Pro-Poor Tourism in the Greater Mekong Sub-Region, Phnom Penh: CDRI.
- JICA (2004), The Study on Integrated Master Plan for Sustainable Development of Siem Reap/Angkor Town in the Kingdom of Cambodia.
- Kang, Chandararot (2005); The Enabling Environment for Micro, Small, Medium Enterprises in Cambodia: An Assessment Based on Surveys in Urban Areas in Kompong Cham, Kompong Chhnang and Kampot (Phnom Penh: CDRI)
- Kim, Annette (2004), "A Market Without the 'Right' Property Rights: Ho Chi Minh City, Vietnam's Newly-emerged Private Real Estate Market", in *Economics of Transition*, Vol. 12(2), pp. 275–305
- Markussen, Thomas (2007), "Land Titling in Cambodia—findings from a field trip", mimeo. World Bank. Second draft. March 2007
- MoT (2006), Tourism Statistics: Annual Report 2006 (Phnom Penh: Ministry of Tourism)
- Mulamir, Murat & Geoffrey Payne (2001, "Legality and Legitimacy in Urban Tenure Issues", paper/abstract submitted to the ESF/N-AERUS Conference at Katholiek Universiteit, Leuven, Belgium, 23–26 May 2001
- NIS (National Institute of Statistics) (1999), *Cambodia Socio-economic Survey 1999* (Phnom Penh: NIS)
- NIS (2005), Statistical Yearbook 2005 (Phnom Penh: National Institute of Statistics, Ministry of Planning).

- Onchan, Tongroj and Saroj Aungsumalin (2002), *Socio-Economic and Environmental Impact* of the Land Titling Program (Centre for Applied Economics Research, Kasetsart University, Thailand)
- Payne, Geoffrey (2000), "Urban Land Tenure Policy Options: Titles or Rights?" Paper presented at the World Bank Urban Forum, Westfields Marriott, Virginia, 3–5 April
- Payne, Geoffrey, Din Somethearith & Beng Hong Socheat Khemro (2004) *Increasing tenuire* security and options for new housing in Phnom Penh, Cambodia, Final Report (Phnom Penh: MLMUPC, Phnom Penh Municipality, UN-Habitat and GTZ Cambodia)
- Payne, Geoffrey, Alain Durand-Lasserve & Carole Rakodi (2007), "Social and economic impacts of land titling programmes in urban and peri-urban areas: A review of the literature", mimeo
- So Sovannarith, Real Sopheap, Uch Utey, Sy Rathmony, Brett Ballard & Sarthi Acharya (2002), *Social Assessment of Land in Cambodia: A Field Study*, Working Paper No. 20 (Phnom Penh: CDRI)
- de Soto, Hernando (2000), *The Mystery of Capital: Why capitalism triumphs in the West and fails everywhere else* (New York: Basic Books)
- Urban Resource Center (2002), *Study on the Relocation of Urban Poor Communities in Phnom Penh* (Phnom Penh: Asian Coalition for Housing Rights)
- World Bank (2006), Halving Poverty by 2015: Poverty Assessment (Phnom Penh: World Bank)
- World Bank (2007), *Sharing Growth: Equity and Development Report 2007* (Phnom Penh: World Bank)

Annex A:

Baseline Survey: Land Titling/Registration (Urban—Phase 2)

Interviewer code:... Date of Interview...../...2005 Ordinal Number of Questionnaire:.....

Interview time: Started at...... Ended at...... Total Time:...... minutes

Family book N.:....

Address:

House No: Street:	Distance of house to:
Village:	Municipality/Provincial office:
Sangkat/commune:	Khan/District office: (km)
Khan/District:	Commune office:(km)
City/provincial town:	Market: (km)

Interview Record

Interviewee's name:	Age:	Sex:	Sex: 1	= male $2 = f$	emale

Household head (plot owner)? 1 = Yes 2 = No

Or what relation to hh head (plot owner)? (Use code 1.2b)

Interviewer's Signature after reviewing the completed questionnaire:

Interviewer's Remarks (according to your feeling, how do you rate your interview process): 1= very good 2= fairly good 3= moderate 4= low

Quality Control Record

Survey Team Leader's Name (code based on the enumerator list):							
Signature after checking all the questions:	Checked on:	. /	2005				
Remarks by Survey Team Leader:							

Supervision by CDRI Researcher

Name of the CDRI Researcher who checked the questionnaire:	Date: .	/	2005
Questions clarified:			

Questions called back:

Records on Data Cleaning and Entry

Name of Data Cleaning Person:	Signature .	Date:/2005
Remarks, questions with problems:		
Data entry by	Signature	Date

I. HOUSEHOLD DEMOGRAPHY

1.1 How many members are in your household (both children and adults)?..... persons

1.2 Sex of household head: 1 = Male 2 = Female

▲ Detailed information about household members: Please list also children who attend kindergarten, if they are not in kindergarten/school, please list from 6 years old and above. (Write number 0 where there are no answers).

	Name (first name only)	Relationship	Sex 1=M	Age Marital	Education	Econo mically	Occupation of each family member.			
No	HH Head/owner (full name)	with hh head	2=F	rige	status	Education	active?	Occ 1	Occ 2	Occ 3
	1.2 a	code 1.2 b	1.2 c	1.2 d	code 1.2 e	code 1.2 f	code 1.2 g	code 1.2 h	code 1.2 i	code 1.2 j
01										
02										
03										
04										
05										
06										
07										
08										
09										
10										
11										
12										
13										
14										
15										
16										
17										
18										
19										
20										
21										

For question **1.2 b**:

1 = Household head 2 = husband or wife 4 = son or daughter 5 = in-law son/daughter 7 = step/foster children 8 = parents 3 =in-law siblings

6 = grandchildren

9 = Grandparents

10 =niece/nephew 11 =Other relatives

12 = home servant
For question 1.2 e:

1 =married 2 =single 3 =Divorced 4 =widow/widower 5 =deserted

For question 1.2 f:

1= No	2 = little	3 = some
4 = primary level	5 = Lower secondary	6 = Upper secondary
7 = tertiary or above	8 = kindergarten	

For question 1.2 g:

1 = Can work for income	2 = Work for no income
3 = study and work (with no income and income)	
4 = Only study	5 = Disabled cannot work
6 = Too old to work. $7 = Too young to work$	

For question 1.2 h - j:

0 = None	1= Government official	2 = Skilled worker $3 =$ Unskilled worker							
4 = Private staff	5= NGO staff	6 = Vendor							
7 = Services (car	7 = Services (car wash, tel/internet/ CP service provider,)								
8 = Motodop/Cy	clo/Taxi driver	9 = hotel/guesthouse owner							
10 = migration ov	verseas	11 = Garment factory worker							
12 = Sales agent/	credit provider,	13 = Agriculture work							
14 = selling labo	ur in agriculture	15 = Gambling							
16 = Workshop (rice mill, distillery,)								

II. PLOT/UNIT CHARACTERISTICS

2.1 What is the location of your plot /unit? (circle one suitable answer only)

- 1 =On the side of main paved road. 2 = Two-way road accessible by vehicle.
- 3 = One-way road accessible by vehicle. 4 = Accessible only by bicycle or motorbike.
- 5 = Accessible only by foot.

2.2 What is the size of this plot in square metres?

Please provide dimensions: (.....m x.....m)

2.3 How is this plot/unit used?

- 1 = Owner residential only (skip sections IV & VIII)
- 2 = Owner commercial only (skip section III)
- 3 = Owner residential and commercial (Fill in sections IV&VIII)
- 4 = Owner residential and agricultural
- 5 = Mixed (owner residential/commercial/agricultural/rent...)
- 6 =Owner residential and rent (for housing or business)

2.4 How many buildings/units in this plot?Buildings/Units

2.5 Does this plot/unit have (state) electricity service? 1 = Yes 2 = No

- **2.6** Does this plot/unit have landline telephone service? 1 = Yes 2 = No
- **2.7** Does this plot/unit have municipal sewage service? 1 = Yes 2 = No

2.8 What is the main source of water for cooking at this plot/unit?

- 1 = piped in dwelling 2 = hand pump/bore hole 3 = dug well
- 4 = pond/stream 5 = (big) river 6 = Purchase

III. RESIDENTIAL PLOT/UNIT

3.1 How many buildings/units are on this plot of land? buildings/units How is each one used?

Building/	Used for	Rental/month (in	No. of Families in	Relationship with owner			
umit	(code) 3.2 a	3.2 b	(code) 3.2 c	(code) 3.2d1	(code) 3.2d2	(code) 3.2d3	
No. 1							
No. 2							
No. 3							
No. 4							
No. 5							
No. 6							
No. 7							

Code **3.2a**: 1 =Owner of the plot only 2 =Owner residential and business

3 =Rent to other for housing 4 =Guesthouse/hotel

5 =Owner resident, commercial and rent to other for staying and business

- 6 = Kitchen 7 = garage 8 = Owner resident, and rent to other for staying
- 9 = relative resident 10 = bathroom 11 = *Kchos*, stock (rice barn,....)

Code **3.2b**: Exchange rate (\$1=.....R):

1993: 2797	1994: 2569	1995: 2467	1996: 2640	1997: 2991	1998: 3774
1999: 3814	2000: 3854	2001: 3924	2002: 3922	2003: 3970	2004: 4023

Code **3.2d1-3**: 1 = Owner of the plot 2 = other family residence/non-rental

3 = other family residence/rental 4 = other non-family residence/non-rental

5 = other non-family residence/rental

6 = other (please specify).....

Building/unit No. 1

Condition of the building/unit where the plot owner currently lives:

(3.3 a)	(3.3 b)	(3.3 c)	(3.3 d)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 - Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

3.4 How long have you been living here? years.

Construction: extension and repairs

apply.

N	Establishment	3.5a Total area (sq. m)	3.5b Year of building	3.5c N. of rooms	3.5d (code) Building permit	3.5e (code) If no BP	3.5f (code) Repair/ex tend	3.5g construction /purchase/ repair cost (US\$)	3.5h (code) Source of funds	3.5i Current price (US\$)	3.5j (code) use it for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Code **3.5d**: 1 =None, 2 =Permission from khan/district 3 = Permission from municipality/provincial,

4 = Permission from LMUPC, 5 = Informed village/commune chief

Code 3.5e :	1 = Complicated and slow administrative system.
	2 = Too expensive $3 =$ Not necessary $4 =$ Don't know where to

- 5 = Old building 6 = Not allowed by law
- Code **3.5f**: 0 = None 1 = Roofing 2 = Wall 3 = Floor 4 = Electricity 5 = Pump/water
 - 6 = Expansion (e.g. new room, new floor,...)
 - 7 = Security (i.e inside bar at window,....)
 - 8 =Repairing bathroom/toilet. 9 =whole
- Code **3.5h**: 1 = relative 2 = friend 3 = moneylender 4 = Trader (name....) 5 = NGO (Name......) 6 = MFI (Name......) 7 = ACLEDA 8 = Own money 9 = Mixed (own money & borrow...) 10 = Other (specify)......
- Code **3.5j**: 1 = living 2 = living and own business place 3 = own business place only 4 = others for living 5 = rent to other for living and business 6 = rent to others for business place 7 = Guest house 8 = owner resident, business and rent to others for living and business 9 = garage 10 = Bathroom 11 = *Kchos*/Stock (rice barn, ...) 12 = kitchen

Building/Unit No. 2

Condition of building/unit:

(3.3 aa) (3.3 ba)		(3.3 ca)	(3.3 da)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 - Tree branches
3 – Tile	3 - Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

N	Establishment	3.5aa Total	3.5ba Year of	3.5ca N. of	3.5da (code)	3.5ea (code)	3.5fa (code)	3.5ga construction	3.5ha (code)	3.5ia Current	3.5ja (code)
		area	building	rooms	Building	If no	Repair	/purchase/	Source	price	used for
		(sq. m)			permit	BP	/extend	repair cost	of funds	(US\$)	
								(US\$)			
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Construction: extension and repairs

Building/unit No. 3

Condition of building/unit:

(3.3 ab)	(3.3 bb)	(3.3 cb)	(3.3 db)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension and repairs

N	Establishment	3.5ab Total area (sq. m)	3.5bb Year of building	3.5cb N. of rooms	3.5db (code) Building permit	3.5eb (code) If no BP	3.5fb (code) Repair/ extend	3.5gb construction /purchase/ repair cost	3.5hb (code) Source of funds	3.5ib Current price (US\$)	3.5jb (code) used for
1	Ground floor							(03\$)			
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Building/unit No. 4

Condition of building/unit:

(3.3 ac)	(3.3 bc)	(3.3 cc)	(3.3 dc)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension and repairs

N	Establishment	3.5ac Total area (sq. m)	3.5bc Year of building	3.5cc N. of rooms	3.5dc (code) Building permit	3.5ec (code) If no BP	3.5fc (code) Repair/ extend	3.5gc construction /purchase/ repair cost (US\$)	3.5hc (code) Source of funds	3.5ic Current price (US\$)	3.5jc (code) used for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Building/unit No. 5

Condition of building/unit:

(3.3 ad)	(3.3 bd)	(3.3 cd)	(3.3 dd)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 - Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 - Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension and repairing

N	Establishment	3.5ad Total	3.5bd Year of	3.5cd N. of	3.5dd (code)	3.5ed (code)	3.5fd (code)	3.5gd construction/pu	3.5hd (code)	3.5id Current	3.5jd (code)
		area	building	rooms	Building	lf no	Repair/	rchase/ repair	Source	price	used for
		(sq. m)			permit	BP	extend	$\cos (0.53)$	of funds	(022)	
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
1	establishment										

Building/unit No. 6

Condition of building/unit:

(3.3 ae)	(3.3 be)	(3.3 ce)	(3.3 de)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

N	Establishment	3.5ae Total area (sq. m)	3.5be Year of building	3.5ce N. of rooms	3.5de (code) Building permit	3.5ee (code) If no BP	3.5fe (code) Repair/ extend	3.5ge construction /purchase/ repair cost	3.5he (code) Source of funds	3.5ie Current price (US\$)	3.5je (code) used for
1	Ground floor							(05\$)			
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire establishment										

Construction: extension and repairs

Building/unit No. 7

Condition of building/unit:

(3.3 af)	(3.3 bf)	(3.3 cf)	(3.3 df)
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 - Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 - Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension and repairs

N	Establishment	3.5af Total area (sq. m)	3.5bf Year of building	3.5cf N. of rooms	3.5df (code) Building permit	3.5ef (code) If no BP	3.5ff (code) Repair/ extend	3.5gf construction /purchase/ repair cost (US\$)	3.5hf (code) Source of funds	3.5if Current price (US\$)	3.5jf (code) used for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire establishment										

IV. COMMERCIAL PLOT/UNIT

4.1 How many commercial buildings/units are on this plot of land?

How is each one used?

Building/	Used for (code)	1 = Rent 2 = Not rent	Rental/month (in USD)	No. of business	Type of business (see SME code)			
unit	4.2 a	4.2 b	4.2 c	4.2 d	(code) 4.2 e1	(code) 4.2 e2	(code) 4.2 e3	
No. 1								
No. 2								
No. 3								
No. 4								
No. 5								
No. 6								
No. 7								

Code **4.2a**: 1 = Business of the owner of the plot 2 = other family business/non-rental 3 = other family business/rental 4 = other non-family business/non-rental

5 = other non-family business/rental 6 = Owner and rent to other for business.

Building/Unit No. 1

The condition	of the P	Plot Owner	business	building/	'unit:

4.3 a	4.3 b	4.3 c	4.3 d
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 – On the ground	1–None
2 – Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 – Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension/repairs

N	Establishment	4.5a Total area (sq. m)	4.5b Year of building	4.5c N. of rooms	4.5d (code) Building permit	4.5e (code) If no BP	4.5f (code) Repair/ extend	4.5g construction /purchase cost (US\$)	4.5h (code) Source	4.5i Current price (US\$)	4.5j (code) used for
1	Ground floor	(54.11)			permit	21	ententa	0000 (0000)	orrunus	(0.54)	
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Code **3.5d**:

1 = None, 2 = Permission from Khan/district

3 = Permission from municipality/provincial

4 =Permission from LMUPC

5 = Informed to village/commune chief

Code **3.5e**:

1 = Complicated and slow administrative system

2 = Too expensive 3 = Not necessary 4 = Don't know where to apply

5 = Old building 6 = Not allowed by law

Code **3.5f**:

```
0 = None 1 = Roofing 2 = Wall 3 = Floor 4 = Electricity 5 = Pump/water
```

6 = Expansion (i.e. new room, new floor,...)

7 = Security (i.e inside bar at window,)

8 = Repairing bathroom/toilet.

9 = whole

Code 3.5h:

1 = relative 2 = friend 3 = moneylender 4 = Trader (name.....)

5 = NGO (Name.....) 6 = MFI (Name.....) 7 = Acleda = Own money

9 = Mixed (own money & borrow ...)

10 =Other (specify).....

Code 4.5j:

1 = Living 2 = Living and own business place 3 = own business place only

4 = rent to other for living 5 = rent to other for living and business

6 = rent to other for business place 7 = guesthouse

8 =own living, business and rent to other for living, business

9 = other (specify)...

Building/unit No. 2

4.3 aa	4.3 ba	4.3 ca	4.3 da
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

The condition of building/unit:

Construction: extension/repairing

		4.5aa	4.5ba	4.5ca	4.5da	4.5ea	4.5fa	4.5ga	4.5ha	4.5ia	4.5ja
Ν	Establishment	Total	Year of	N. of	(code)	(code)	(code)	construction	(code)	Current	(code)
		area	building	rooms	Building	If no	Repair/	/Purchase	Source	price	use it
		(sq. m)			permit	BP	extend	cost (US\$)	of fund	(US\$)	for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Building/unit No. 3

The condition of building/unit:

4.3 ab	4.3 bb	4.3 cb	4.3 db		
Roof	Wall	Floor	Fence		
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None		
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches		
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo		
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood		
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire		
6 – None	6 – None		6 – Concrete/Bar		

Construction: extension/repairing

101
-

Building/unit No. 4

The condition of building/unit:

4.3 ac	4.3 bc	4.3 cc	4.3 dc
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

0	Johsti dettoli. Extension/repairing										
		4.5ac	4.5bc	4.5cc	4.5dc	4.5ec	4.5fc	4.5gc	4.5hc	4.5ic	4.5jc
Ν	Establishment	Total	Year of	N. of	(code)	(code)	(code)	construction	(code)	Current	(code)
		area	building	rooms	Building	If no	Repair/	/Purchase	Source	price	use it
		(sq. m)			permit	BP	extend	cost (US\$)	of fund	(US\$)	for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Construction: extension/repairing

Building/unit No. 5

The condition of building/unit:

4.3 ad	4.3 bd	4.3 cd	4.3 dd		
Roof	Wall	Floor	Fence		
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None		
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches		
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo		
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood		
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire		
6 – None	6 – None		6 – Concrete/Bar		

Construction: extension/repairing

	sousdated on the story repairing										
		4.5ad	4.5bd	4.5cd	4.5dd	4.5ed	4.5fd	4.5gd	4.5hd	4.5id	4.5jd
Ν	Establishment	Total	Year of	N. of	(code)	(code)	(code)	construction	(code)	Current	(code)
		area	building	rooms	Building	If no	Repair/	/Purchase	Source	price	use it
		(sq. m)			permit	BP	extend	cost (US\$)	of fund	(US\$)	for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										1

Building/unit No. 6

4.3 ae	4.3 be	4.3 ce	4.3 de		
Roof	Wall	Floor	Fence		
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None		
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches		
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo		
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood		
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire		
6 – None	6 – None		6 – Concrete/Bar		

The condition of building/unit:

Construction: extension/repairing

		1 590	4 5he	4 5 60	4 5de	4 500	4 5fo	4 500	4 5he	4 5ie	4 5io
N	Establishment	Total	Vaca of	T.SCC	(and a)	4.500 (and a)	4.51C	4.5gc	(and a)	Cumant	
IN	Establishment	Total	rear of	IN. 01	(code)	(code)	(code)	construction	(code)	Current	(code)
		area	building	rooms	Building	If no	Repair/	/Purchase	Source	price	use it
		(sq. m)			permit	BP	extend	cost (US\$)	of fund	(US\$)	for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

Building/unit No. 7

The condition of building/unit:

4.3 af	4.3 bf	4.3 cf	4.3 df
Roof	Wall	Floor	Fence
1 – Thatch/tent	1 – Thatch/tent	1 - On the ground	1–None
2 - Zinc sheets	2 - Zinc sheets	2 – Wood/bamboo strips	2 – Tree branches
3 – Tile	3 – Wood	3 – Concrete/brick	3 – Small pole/bamboo
4 – Wood/plywood	4 – Concrete/Brick	4 – Concrete and wood	4 - Wood
5 – Concrete	5 – Concrete and wood	5 – Other:	5 – Wire
6 – None	6 – None		6 – Concrete/Bar

Construction: extension/repairing

		4.5af	4.5bf	4.5cf	4.5df	4.5ef	4.5ff	4.5gf	4.5hf	4.5if	4.5jf
Ν	Establishment	Total	Year of	N. of	(code)	(code)	(code)	construction	(code)	Current	(code)
		area	building	rooms	Building	If no	Repair/	/Purchase	Source	price	use it
		(sq. m)			permit	BP	extend	cost (US\$)	of fund	(US\$)	for
1	Ground floor										
2	Mezzanine										
3	1 st floor										
4	Mezzanine										
5	2 rd floor										
6	3 rd floor										
7	4 th .floor										
8	5 th .floor										
9	entire										
	establishment										

V. LAND OWNERSHIP, LAND TRANSACTIONS, AND LAND CONFLICTS

This section concerns all the plots of land or units on which you are currently living and/or working and/or farming. In the table below, your residential plot is referred to as Plot 1.

5.1 Do you have any other residential/unit or commercial or agricultural lands? 1 =Yes (skip 5.2) 2 =No (ask plot 1)

5.2 If "No", why?

CDRI

1 = sold

2 = never had any other land

3 = 1 lost other land due to grabbing by other/state took the land back

4 = lost land due to displacement (just settled in this village/ divorce)

5 = gave all away to offspring

5.3 If "Yes", How many plots do you own?

These additional plots are referred to as Plots 2-4.

Q.	a	b	с	d	e	f	g	h	i	j	k	1	m	n	0	р	q
	P. 1	P. 2	P. 3	P. 4	P. 5	P. 6	P. 7	P. 8	P. 9	P. 10	P. 11	P. 12	P. 13	P. 14	P. 15	P. 16	P. 17
5.4																	
5.5																	
5.6																	
5.7																	
5.8																	
5.9																	
5.10																	
5.11																	
5.12																	
5.13																	
5.14																	
5.15																	

Note:

P. = Plot. Please fill in plot by plot.

Please see the questionnaires in the table below. Please feel in the number for each question as in the column.

Question N	Question for the above questionnaires	Use the following
5.4	 Where is it located? (commune/district/province) 1. In the village of interview 2. In the district/khan of interview 3. In other district/khan 4. In peri – urban of Phnom Penh 5. In province nearby the city 6. In district/provincial town 7. In rural area 8. In other place (specify) 	1 2 3 4 5 6 7 8
5.5	What is the size? (in square metres)	
5.6	 What is plot used for? (Use codes) 1 = own living only/giving other to live/garage 2 = living and own business place 3 = own business place only/store/other doing business 4 = rent to other for living 5 = rent to other for living and business/agriculture 6 = rent to other for business place/agriculture 7 = guesthouse 8 = own living/business and rent to other for living and business/agriculture 9 = rice field (own/other) 10 = chamkar land/plantation/fruit trees 11 = living and agriculture (planting, raising animal,) 12 = leave it idle 	$ \begin{array}{c} 1\\ 2\\ 3\\ 4\\ 5\\ 6\\ 7\\ 8\\ 9\\ 10\\ 11\\ 12\\ \end{array} $

5.7	Irrigation system $1 = $ Yes, $2 = $ No	
5.8	If you sell this land now, how much would you get in US\$?	
5.9	When did you acquire this plot? 1 = Before 1979 2 = Between 1979 and the 1993 (UNTAC Election) After 1993 (UNTAC Election)	1 2 3
5.10	How did you acquire it? 1 = given by the state (or local authority) 2 = inherited it or donated by relatives (parents/brother/sister) 3 = donated by friends 4 = own purchase 5 = cleared land/occupied for free 6 = mixed (given by the state and inheritance/purchase	1 2 3 4 5 6
5.11	If you bought this plot, how much did you pay for it? (US\$)	
5.12	When did you buy this plot? (in which year)	
5.13	Do you have any paper to certify this land? 1 = application receipt 2 = land investigation paper 3 = certificate (or title) 4 = agreement paper between buyer and seller 5 = letter transferring ownership at village level 6 = letter transferring ownership at commune level 7 = none (or no document available in the house) (ask 5.14) 8 = lost application receipt	1 2 3 4 5 6 7 8
5.14	If you haven't registered, what was the main reason? 1 = thought it was not necessary 2 = to avoid paying tax 3 = too much under table (unofficial) fee 4 = not knowing how to register 5 = land conflict 6 = no confidence in land title 7 = too complicated administrative processing 8 = no justify 9 = no measuring plan by the state/prohibited zone by the state	1 2 3 4 5 6 7 8 9
5.15	If you leave it idle, why? 1 = not enough labourers 2 = not profitable 3 = rotation 4 = lack of investment capital 5 = for transaction 6 = for rental/giving to children	1 2 3 4 5 6

Land Transactions

5.16 Have you sold any land/house/unit since 1989? (Circle an appropriate code) 1 = Yes 2 = No (If "No",go to Question 5.27)

	1							1
(Ask by plot)	a Plot 1	b Plot 2	c Plot 3	d Plot 4	e Plot 5	f Plot 6	g Plot 7	h Plot 8
5 17 Where was the land/house you sold?	11001	11002	1100.5	1101 4	1100.5	1101.0	11007	11010
(circle one)								
1 in the village of interview	1	1	1	1	1	1	1	1
2 in the district/khan of interview	2	2	2	2	2	2	2	2
2 in other district/khan	3	23	2	2	2	3	2	2
A in peri urban area of Dhnom Danh	4		- 3	- 3	- 3	- 3		4
5 in province peerby the sity	5	5	-+ -5	4 5	- -	5	5	
6 in district/provincial town	5	5	5	5	5	5	5	5
7 in much area	7	7	7	7	7	7	07	0
7. In rural area	/	/	/	/	/	/	/	/
8. in other place (specify)	8	8	8	8	8	8	8	8
5.18 what is the location of this plot/unit?								
(Circle an appropriate code)								
I = On the side of main paved road	1	1	1	1	1	1	1	1
2 = T wo directions accessible for vehicle	2	2	2	2	2	2	2	2
3 = One direction accessible for vehicle	3	3	3	3	3	3	3	3
4 = Accessible only by bicycle or motorbike	4	4	4	4	4	4	4	4
5 = Accessible only by foot	5	5	5	5	5	5	5	5
5. 19 What was the size of the plot you sold?								
(in sq. m.)								
5.20 What kind of land was the plot?								
I-Residential	1	1	1	1	1	1	1	1
2- House/unit	2	2	2	2	2	2	2	2
3- Rice land	3	3	3	3	3	3	3	3
4- Chamkar land	4	4	4	4	4	4	4	4
5- Lake/forest /Apsara zone	5	5	5	5	5	5	5	5
5.21 How much did you sell it for? (in US\$)								
5 22 When did you sall it? (year)								
3.22 when did you sen it? (year)								
5.23 How did you document your								
transaction?	1	1	1	1	1	1	1	1
1 change name with receipt at the village	2	2	2	2	2	2	2	2
level	3	3	3	3	3	3	3	3
2 change name with receipt at the commune	1	4	1	4	4	1	1	1
2. Change name with receipt at the commune	5	5	5	5	5	5	5	5
3. change name with receipt at the district	5	5	5	5	5	5	5	5
level	6	6	6	6	6	6	6	6
4. change name on title at provincial level								
5. change name of ownership by making title								
at the MLUPC								
6. Agreement between buyer and seller								
5.24 Who paid for this document?								
1=Seller 2=Buyer 3=both								
5.25 How much was that? (in USD)								

	l I	ĺ				l I		
5.26 Why did you sell it? (select the one								
main reason)	1	1	1	1	1	1	1	1
1 = need cash for business/for profit	2	2	2	2	2	2	2	2
2 = pay for health treatment	3	3	3	3	3	3	3	3
3 = need cash for food	4	4	4	4	4	4	4	4
4 = pay off debts/for children's education	5	5	5	5	5	5	5	5
5 = funeral for family member	6	6	6	6	6	6	6	6
6 = too small for profitable farming	7	7	7	7	7	7	7	7
7 = emigrate	8	8	8	8	8	8	8	8
8 = find job/change occupation	9	9	9	9	9	9	9	9
9 = poor soil/none productive.	10	10	10	10	10	10	10	10
10 = pressure from rich/powerful men/no path	11	11	11	11	11	11	11	11
11 = flooded	12	12	12	12	12	12	12	12
12 = too distant	13	13	13	13	13	13	13	13
13 = changing the place of residence	14	14	14	14	14	14	14	14
14 = built house/purchased means of								
transport								

Land Conflicts

5.27 Have you ever had conflicts over land or other property since the national election of 2003?

1 =Yes 2 =No (go to Section VI Credit Market)

5.28 When did it happen? In what year.....

5.29 The conflict occurred over which land ?

- 1 = Residential land 2 = Agricultural land 3 = Commercial land
- 4 = Mixed use land 5 = Condominium property (flat/shop)

5.30 How much land/property was in conflict?sq metres.

5.31 What was the type of the land conflict? (can be multiple answers)

- 1 = the land was grabbed by private individuals
- 2 = the land was grabbed by private company
- 3 = the land was grabbed by soldiers/armed officials
- 4 = the land was grabbed by officials
- 5 = the land was taken/reclaimed by state authorities
- 6 = boundary conflict with neighbours, others
- 7 =ownership conflict with non-relatives
- 8 =ownership conflict with relatives
- 9 = other (specify)

If you had a land conflict, where and/or who did you first go to for solving the conflict?

Conflict at	5.32a Code Who first helped solving the conflict?	5.32b Solved? 1 = Yes 2 = No	5.32c Cost	5.32d Result satisfied? 1 = Yes 2 = No	5.32e Comment, why "yes" or why "no"	5.32f Code Conflict was solved.
Plot 1						
Plot 2						
Plot 3						
Plot 4						
Plot 5						

Code 5.32a :	1 = Mutual agreement	2 = neighbour, family	$\frac{1}{\text{friends}}$ 3 = Monk/ Achar
	4 = Village chief	5 = Commune leader	
	6 = Dispute Settlement	Committee (district)	
	7 = Dispute Settlement	Committee (province))
	8 = Dispute Settlement	Committee (Phnom P	enh)
	9 = Court at the province	ce	
	10 = Appeal court in Pl	hnom Penh	
	11 = At the national cou	ırt	
Code 5.32f	1 = Lost some of it	2 = Lost all of it	3 = Didn't lose any land.
	4 = No answer or no id	ea	

VI. CREDIT MARKET

6.1 Have you obtained any loans in cash or gold since the national election of 2003? 1 = Yes 2 = None (Skip to VII)

6.1a How many loans in total have you had since then? loans

6.2 Of that number, how many outstanding loans do you currently have? (number of loans)

Ν	6.2a	6.2b	6.2c	6.2d	6.2e	6.2f	6.2g	6.2h	6.2i	6.2j
	Code	When	Code	Amount	[code below if less than	Code	Code	Interest	Both	Debt
	Туре	month	From	of loan	a month]	For	Collateral	rate per	principle	1=paid off
	of	/ year	whom?	in US\$	For how many months	what	to obtain	month	and interest	2=in debt
	loan				(from start to when you	purpose	loan	(%)	to be repaid	
					promise to pay back)					
1										
					month					
2										
					month					
3										
					month					
4										
					month					
5										
					month					
6										
					month					

Complete the below bable for all loans announced in 6.1 a:

Code 6.2a	1 = Loan in cash or gold (please convert to USS)	5)
------------------	--	----

- 2 = Loan in paddy/rice need to be repaid in kind
- 3 = Loan in rice or fertiliser or cash to be repaid in labour

```
Code 6.2c 1 = \text{relative } 2 = \text{friend } 3 = \text{moneylender } 4 = \text{trader (what.....)}
```

$$5 = NGO$$
 (name......) $6 = MFI$ (name......) $7 = Acleda \ 8 = Canadia Bank$

Code **6.2e**
$$0.7 = 20$$
 days $0.5 = 15$ days $0.2 = 6$ days $999 =$ No fixed date

```
Code 6.2f
```

1 = Farming/raising animals	9 = Buying land/house
2 = Business start-up/looking for job	10 = Pay off debt
3 = Business expansion	11 = Home improvement
4 = Buying input for business	12 = Ceremony
5 = Food shortage	13 = Transportation
6 = Health	14 = Build house
7 = Education	15 = Other
8 = Solving hh conflict	
-	

8 = group loan/guarantee $9 = other:$	Code 6.2g $1 = no$ $2 = gold$ 5 = letter of ownership transferred 8 = group loan/guarantee	3 = animal 6 = land/house title 9 = other:	4 = land receip 7 = other asset
---------------------------------------	--	--	------------------------------------

VII. HOUSEHOLD EXPENDITURES

Since Khmer New Year (April) this year (2005), estimated expenditure on non-food items by all members of household:

Item/Activity	Amount	a Code From which source
7.1 Medical care/health treatment	USD	
 7.2 Social Ceremony (wedding/funeral/etc) 7.2 a: In the house 7.2 b: Outside the house 	USD USD	
7.3 Travel/tourism	USD	
 7.4 Education 7.4 a: Husband/Wife 7.4 b: Children (material/school fee/private class) 7.4 c: Other members 	USD USD USD	
7.5 Contributions/fees for public construction	USD	
7.6 Purchasing furniture/household appliance	USD	
7.7 Modes of transportation (car, motorbike, bike)	USD	
7.8 Other (telephone,)	USD	

Code **a**: 1 = loan/credit 2 = own sources 3 = gift 4 = mixed

7.9 Total (sum of Q7.1- Q7.8) :USD

Average monthly expenses on (last month's expenses):

Item/Activity	Amount	a Code (From which source)
7.10 Telephone card/service	USD	
7.11 Electricity/garbage	USD	
7.12 Water	USD	
7.13 Cooking (gas, charcoal, chopwood,)	USD	
7.14 Clothes/shoes/make up/soap	USD	
7.15 Pay for servant	USD	

Code **a**: 1 = 10an/credit 2 = 0 own sources 3 = gift 4 = mixed

7.16 Total (sum of Q7.10- Q7.15):USD

Average weekly expenses on (last week's expenses):

Item/Activity	Amount	a Code From which source
7.17 Enterainment	USD	
7.18 Fuel/transportation	USD	
7.19 Food/household consumption	USD	
7.20 Eat out	USD	
7.21 Other (specify)	USD	

Code **a**: 1 = loan/credit 2 = own sources 3 = gift 4 = mixed

7.28 Total (sum of Q7.17 - Q7.21):USD

VIII. SMALL/MEDIUM ENTERPRISES (SMEs) (for commercial and mixed plots)

Background Information about Enterprise

	а	b	с	d
	Business 1	Business 2	Business 3	Business 4
8.1 Relationship of respondent with owner:				
1 = Owner $2 =$ Family member $3 =$ Partner				
8.2 Is this establishment managed directly by the owner? 1 = Yes 2 = No				
8.3 What is the gender of the owner/principal owner? 1 = Male 2 = Female				
8.4 Please describe the type of industry/business and the product/services: (Code SMEs)				
8.5 In which year did your production/operation start?				
8.6 Is this business registered? $1 = \text{Yes } 2 = \text{No}$				
8.7 If not, why? (Code):				
1 = Not necessary 2 = Too expensive				
3 = Too complicated $4 = Will$ raise attention of				
authorities $5 = $ Other/don't know				

Finance and Credit

Γ

	8.8 a	8.8 b	8.8 c
Sources	Sources of finance from	Amount	For what?
	(coding below)		
Number 1		USD	
Number 2		USD	
Number 3		USD	
Number 4		USD	

Code 8.8b	1 = Own resources/saving 2 = Family money/supplier 3 = Selling land/property						
	4 = Selling other assets	Selling other assets $5 =$ Personal borrowing $6 =$ M					
	7 = Private commercial bank (Give name of institution)						
	8 = NGOs /Micro credit organisations (Give name of institution)						
	9 = State bank $10 =$ Partnership/ shares		11 = Pawned goods				
	12 = Other (specify)						
Code 8.8c	1 = start a new business	2 = operate the business	3 = expand business				

Access to Formal Credit (Commercial Bank, Acleda, MFI)

8.9 Are you looking for credit at present? 1 = Yes (ask q. 8.11) 2 = No (ask q. 8.10)

8.10 I have not sought a loan because : (Code 9.10 a - f): 1 = strongly disagree; 2 = what disagree; 3 = no comment; 4 = somewhat agree; 5 = strongly agree	some
8.10a I am uninformed regarding banking/loan procedures (above coding)	
8.10b The profit from expansion would not be higher than the interest rate (above coding)	
8.10c Getting a loan requires informal payment (above coding)	
8.10d Collateral requirements of bank make it impossible for me to take out a loan (above coding)	
8.10e Bank procedures are too time consuming (above coding)	
8.10f Other (specify)	
I have sought a loan, but was refused because: (<i>Code 9.11 a - f</i>): 1 = strongly disagree; 2= some what disagree; 3 = no comment; 4 = somewhat agree; 5 = strongly agree	
8.11a I do not have sufficient collateral (above coding)	
8.11b I have an outstanding loan (above coding)	
8.11c I am a firs-ttime borrower (above coding)	
8.11d I improperly or incompletely filled out the application forms (above coding)	
8.11e I do not known why it was rejected (above coding)	
8.11f Other (specify)	

Employment Structure

	Α	В	С
How many employees do you have now?	Male	Female	Total
8.12 Permanent employees			
8.13 Daily wage workers			
8.14 Family members			
How many employees did you have when you started?	A Male	B Female	C Total
8.15 Permanent employees			
8.16 Daily wage workers			
8.17 Family members			

IX: FEELINGS OF PEOPLE ON LAND TITLING AND REGISTRATION

9.1 In your opinion, what are the benefits from	Α	В	С	D
land/property titling and registration? Fill in from most to				

Coding: (multiple answers permitted)

- 1 Having stability in owning the land.
- 2 End conflict.
- 3 Ease in borrowing money.
- 4 Ease in transferring inheritance.
- 5 Nothing different.
- 6 No idle land.
- 7 Ease in selling.

Annex B: Villages and Commune	s Included in the Survey
--------------------------------------	--------------------------

District	Commune	Village	N. of HHs		Type of land use			se		
		C	Experimental	Control	1	2	3	4	5	6
Daun Penh	Boeng Reang	Village 1	9		6	0	2	0	1	0
		Village 2	10		6	0	4	0	0	0
		Village 5	7		3	0	4	0	0	0
		Village 6	3		1	0	2	0	0	0
		Village 7	31		20	0	8	0	0	3
		Village 8	9		7	0	1	0	0	1
		Village 9	17		4	0	9	0	2	2
Tuol Kork	Boeng Kak 1	Village 1	36		27	0	7	0	1	1
		Village 2	53		45	0	7	0	0	1
		Village 7	47		36	0	6	3	0	2
		Village 14	42		27	0	12	0	3	0
	Tuek L'ak 1	Village 1	12		3	0	4	0	3	2
		Village 2	48		30	0	10	0	1	7
		Village 3	33		13	0	9	0	5	6
		Village 4	3		2	0	1	0	0	0
		Village 6	24		15	0	4	0	1	4
		Village 10	45		35	0	6	0	0	4
		Village 13	11		5	1	4	1	0	0
		Village 14	13		13	0	0	0	0	0
Chamkar										
Mon	Olympic	Village 3	47		27	0	6	0	3	11
		Village 4	48		31	0	9	0	1	7
		Village 5	48		22	0	16	0	1	9
	Tonle Basak	Village 7		48	30	0	7	0	2	9
		Village 9		49	19	0	24	0	4	2
		Village 16		24	7	1	14	0	1	1
	Boeng									
	Trabaek	Village 2		24	13	0	8	0	0	3
		Village 3		24	8	0	9	0	3	4
		Village 4		24	12	0	5	0	0	7
		Village 5		24	14	0	6	0	0	4
		Village 6		24	11	0	5	0	2	6
		Village 8		24	6	0	7	0	5	6
	Experimental		596							_
	Control			265						
Total		861	l							

B1. Urban Areas of Phnom Penh

District	Commune	Village	No. of HHs	Type of land use					use	
	· · ·		Experimental	Control	1	2	3	4	5	6
Meanchey	Prek Pra	Preah Ponlea	•	60	43	0	11	4	1	1
•		Prek Ta Pov		60	50	0	8	2	0	0
	Chaom									
Dangkao	Chau	Chaom Chau	48		13	0	4	0	12	19
		Trapeang Thloeng	48		5	0	4	0	10	29
		Prey Sandaet	48		40	0	7	1	0	0
		Prey Lvea	54		32	0	4	15	2	1
	Cheung Aek	Cheung Aek		60	44	0	14	0	2	0
		Roluos		55	44	0	6	3	2	0
Russey Keo	Khmuonh	Khmuonh	60		46	0	14	0	0	0
		Banla S'et	60		30	0	13	7	5	5
		Anlong Kngan	72		52	0	17	2	1	0
		Trapeang Reang	60		40	0	10	1	0	0
		Thinei	00		49	0	10	1	0	0
	Prek Leap	Prek Leap		31	19	0	8	4	0	0
	•	Khtor		30	9	0	6	3	5	7
		Bak Khaeng		60	13	0	19	18	4	6
	experimental		450							
	control			356						
Total in Peri	-Urban Areas	806								

B3. Siem Reap Province										
District	Commune	Village	N. of HHs			Ty	pe of l	and u	use	
			Experimental	Control	1	2	3	4	5	6
Siem Reap	Siem Reap	Spean Chreav	60		38	0	10	10	2	0
		Pou	60		32	0	19	8	1	0
		Kakranh	60		40	0	6	10	2	2
		Krasang Roleung	30		14	0	4	10	1	1
	Sala									
Siem Reap	Kamreuk	Chonlong		30	20	0	4	4	2	0
		Wat Svay		71	36	0	13	12	8	2
		Ta Vien		42	21	0	9	3	6	3
		Trapeang Traeng		30	9	0	2	16	3	0
		Wat Bour		65	38	0	11	9	4	3
		Wat Damnak		90	36	0	29	6	13	6
			210	328	285		110	92	47	17
Total number	er in Siem									
Reap		538								
B4. Serei Sa	ophoan									
District	Commune	Village	N. of HHs		·	Ty	pe of l	and u	ise	
		~~~~	Exp	Con	1	2	3	4	5	6
Serei	Kompong									
Saophoan	Svay	Phum 2		108	46	2	34	17	7	2
-		Souphi		110	50	0	32	23	3	2
		Kg. Svay		72	31	0	27	8	5	1
Serei										
Saophoan	Preah Ponlea	Phum 1		72	23	0	31	14	3	1
•		Phum 4		36	14	0	21	0	1	0
Serei										
Saophoan	Ou Ambel	Roung Masin		37	17	0	7	9	2	2
		Ou Ambel		36	14	0	3	15	4	0
		Kourothan		36	16	0	14	3	3	0
				507	211	2	169	89	28	8
Total numb	er in Banteay									
Meanchev	v	507								
<b>v</b>										
Note: Type	of land uses									
1 = Owner re	esidential only	1								
2 = Owner co	ommercial only	V								
3 = Owner re	esidential and c	ommercial								
4 = Owner re	esidential and a	gricultural								
5 = Mixed (or	wner residenti	al/commercial/agri	cultural/rent)							
6 - 0	aidantial and r	ont (for stay or bus	inoss)							
o = Owner re	sidential and r	em (for stay or bus	mess)							

# **CDRI** Working Papers

- Kannan, K.P. (November 1995), Construction of a Consumer Price Index for Cambodia: A Review of Current Practices and Suggestions for Improvement (Working Paper No. 1) \$5.00
- McAndrew, John P. (January 1996), Aid Infusions, Aid Illusions: Bilateral and Multilateral Emergency and Development Assistance in Cambodia, 1992-1995 (Working Paper No. 2) \$5.00
- 3) Kannan, K.P. (January 1997), Economic Reform, Structural Adjustment and Development in Cambodia (Working Paper No. 3) \$5.00
- 4) Chim Charya, Srun Pithou, So Sovannarith, John McAndrew, Nguon Sokunthea, Pon Dorina & Robin Biddulph (June 1998), *Learning from Rural Development Programmes in Cambodia* (Working Paper No. 4) \$7.50
- 5) Kato, Toshiyasu, Chan Sophal & Long Vou Piseth (September 1998), *Regional Economic Integration for Sustainable Development in Cambodia* (Working Paper No. 5) \$6.00
- 6) Murshid, K.A.S. (December 1998), *Food Security in an Asian Transitional Economy: The Cambodian Experience* (Working Paper No. 6) \$9.00
- 7) McAndrew, John P. (December 1998), Interdependence in Household Livelihood Strategies in Two Cambodian Villages (Working Paper No. 7) \$9.00
- Chan Sophal, Martin Godfrey, Toshiyasu Kato, Long Vou Piseth, Nina Orlova, Per Ronnås & Tia Savora (January 1999), *Cambodia: The Challenge of Productive Employment Creation* (Working Paper No. 8) \$9.00
- 9) Teng You Ky, Pon Dorina, So Sovannarith & John McAndrew (April 1999), *The UNICEF/Community Action for Social Development Experience—Learning from Rural Development Programmes in Cambodia* (Working Paper No. 9) \$4.50
- 10) Gorman, Siobhan, with Pon Dorina & Sok Kheng (June 1999), *Gender and Development in Cambodia: An Overview* (Working Paper No. 10) \$6.00
- 11) Chan Sophal & So Sovannarith (June 1999), *Cambodian Labour Migration to Thailand: A Preliminary Assessment* (Working Paper No. 11) \$3.00
- 12) Chan Sophal, Toshiyasu Kato, Long Vou Piseth, So Sovannarith, Tia Savora, Hang Chuon Naron, Kao Kim Hourn & Chea Vuthna (September 1999), *Impact of the Asian Financial Crisis on the SEATEs: The Cambodian Perspective* (Working Paper No. 12) \$4.50
- 13) Ung Bunleng, (January 2000), Seasonality in the Cambodian Consumer Price Index (Working Paper No. 13) \$3.00
- 14) Toshiyasu Kato, Jeffrey A. Kaplan, Chan Sophal and Real Sopheap (May 2000), Enhancing Governance for Sustainable Development (Working Paper No. 14) \$6.00
- 15) Martin Godfrey, Chan Sophal, Toshiyasu Kato, Long Vou Piseth, Pon Dorina, Tep Saravy, Tia Savara and So Sovannarith (August 2000), *Technical Assistance and Capacity Development in an Aid-dependent Economy: the Experience of Cambodia* (Working Paper No. 15) \$10.00
- 16) Sik Boreak, (September 2000), Land Ownership, Sales and Concentration in Cambodia (Working Paper No. 16) \$7.00
- 17) Chan Sophal, and So Sovannarith, with Pon Dorina (December 2000), *Technical Assistance and Capacity Development at the School of Agriculture Prek Leap* (Working Paper No. 17) \$8.00

- 18) Martin Godfrey, So Sovannarith, Tep Saravy, Pon Dorina, Claude Katz, Sarthi Acharya, Sisowath D. Chanto and Hing Thoraxy (August 2001), A Study of the Cambodian Labour Market: Reference to Poverty Reduction, Growth and Adjustment to Crisis (Working Paper No. 18) \$7.00
- 19) Chan Sophal, Tep Saravy and Sarthi Acharya (October 2001), *Land Tenure in Cambodia: a Data Update* (Working Paper No. 19) \$10.00
- 20) So Sovannarith, Real Sopheap, Uch Utey, Sy Rathmony, Brett Ballard and Sarthi Acharya (November 2001), Social Assessment of Land in Cambodia: A Field Study (Working Paper No. 20) \$10.00
- 21) Bhargavi Ramamurthy, Sik Boreak, Per Ronnås and Sok Hach (December 2001), Cambodia 1999-2000: Land, Labour and Rural Livelihood in Focus (Working Paper No. 21) \$10.00
- 22) Chan Sophal and Sarthi Acharya (July 2002), Land Transactions in Cambodia: An Analysis of Transfers and Transaction Records (Working Paper No. 22) \$8.00
- 23) Bruce McKenney and Prom Tola. (July 2002), *Natural Resources and Rural Livelihoods in Cambodia* (Working Paper No. 23) \$10.00
- 24) Kim Sedara, Chan Sophal and Sarthi Acharya (July 2002), Land, Rural Livelihoods and Food Security in Cambodia (Working Paper No. 24) \$10.00
- 25) Chan Sophal and Sarthi Acharya (December 2002), Facing the Challenge of Rural Livelihoods: A Perspective from Nine Villages in Cambodia (Working Paper No. 25) \$10.00
- 26) Sarthi Acharya, Kim Sedara, Chap Sotharith and Meach Yady (February 2003), *Off-farm and Non-farm Employment: A Perspective on Job Creation in Cambodia* (Working Paper No. 26) \$5.00
- 27) Yim Chea and Bruce McKenney (October 2003), Fish Exports from the Great Lake to Thailand: An Analysis of Trade Constraints, Governance, and the Climate for Growth (Working Paper No. 27) \$10.00
- 28) Prom Tola and Bruce McKenney (November 2003), *Trading Forest Products in Cambodia: Challenges, Threats, and Opportunities for Resin* (Working Paper No. 28) \$7.00
- 29) Yim Chea and Bruce McKenney (November 2003), *Domestic Fish Trade: A Case Study of Fish Marketing from the Great Lake to Phnom Penh* (Working Paper No. 30) \$5.00
- 30) Caroline Hughes and Kim Sedara with the assistance of Ann Sovatha (February 2004), *The Evolution of Democratic Process and Conflict Management in Cambodia: A Comparative Study of Three Cambodian Elections* (Working Paper No. 30) \$13.50
- 31) Robert B. Oberndorf, J.D. (May 2004), *Law Harmonisation in Relation to the Decentralisation Process in Cambodia* (Working Paper No. 31) \$6.50
- 32) K.A.S. Murshid and Tuot Sokphally (April 2005), *The Cross Border Economy of Cambodia: An Exploratory Study* (Working Paper No. 32) \$5.00
- 33) Kasper K. Hansen and Neth Top (December 2006), *Natural Forest Benefits and Economic Analysis of Natural Forest Conversion in Cambodia* (Working Paper No. 33) \$8.00
- 34) Pak Kimchoeun, Horng Vuthy, Eng Netra, Ann Sovatha, Kim Sedara, Jenny Knowles and David Craig (forthcoming), *Accountability and Neo-patrimonialism in Cambodia: A Critical Literature Review* (Working Paper No. 34)
- 35) Kim Sedara and Joakim Öjendal With the assistance of Ann Sovatha, Where Decentralisation Meets Democracy: Civil Society, Local Government, and Accountability in Cambodia (Working Paper No. 35)
- 36) Lim Sovannara, Youth Migration and Urbanisation in Cambodia (Working Paper No. 36)



Phnom Penh: Land use in the area around CDRI

# CDRI - Cambodia's LeadingIndependent Development Policy Research Institute

56 Street 315, Tuol Kork, Phnom Penh, Cambodia

- PO Box 622, Phnom Penh, Cambodia
- (855-23) 881-384/881-701/881-916/883-603/012 867-278
- ₿ (855-23) 880-734

E-mail: cdrimail@online.com.kh / pubs@cdri.forum.org.kh Website: http://www.cdri.org.kh

